Penal Law

CHAPTER 40. OF THE CONSOLIDATED LAWS

PART ONE. GENERAL PROVISIONS

TITLE A. GENERAL PURPOSES, RULES OF CONSTRUCTION, AND DEFINITIONS

Article 1. General Purposes

§ 1.00 Short title

This chapter shall be known as the "Penal Law."

§ 1.05 General purposes

The general purposes of the provisions of this chapter are:

- 1. To proscribe conduct which unjustifiably and inexcusably causes or threatens substantial harm to individual or public interests;
- 2. To give fair warning of the nature of the conduct proscribed and of the sentences authorized upon conviction;
- 3. To define the act or omission and the accompanying mental state which constitute each offense;
- 4. To differentiate on reasonable grounds between serious and minor offenses and to prescribe proportionate penalties therefor;
- 5. To provide for an appropriate public response to particular offenses, including consideration of the consequences of the offense for the victim, including the victim's family, and the community; and
- 6. To insure the public safety by preventing the commission of offenses through the deterrent influence of the sentences authorized, the rehabilitation of those convicted, the promotion of their successful and productive reentry and reintegration into society, and their confinement when required in the interests of public protection.

Article 5. General Rules of Construction and Application

§ 5.00 Penal law not strictly construed

The general rule that a penal statute is to be

strictly construed does not apply to this chapter, but the provisions herein must be construed according to the fair import of their terms to promote justice and effect the objects of the law.

Article 10. Definitions

§ 10.00 Definitions of terms of general use in this chapter

Except where different meanings are expressly specified in subsequent provisions of this chapter, the following terms have the following meanings:

- 1. "Offense" means conduct for which a sentence to a term of imprisonment or to a fine is provided by any law of this state or by any law, local law or ordinance of a political subdivision of this state, or by any order, rule or regulation of any governmental instrumentality authorized by law to adopt the same.
- 2. "Traffic infraction" means any offense defined as "traffic infraction" by section one hundred fifty-five of the vehicle and traffic law.
- 3. "Violation" means an offense, other than a "traffic infraction," for which a sentence to a term of imprisonment in excess of fifteen days cannot be imposed.
- 4. "Misdemeanor" means an offense, other than a "traffic infraction," for which a sentence to a term of imprisonment in excess of fifteen days may be imposed, but for which a sentence to a term of imprisonment in excess of one year cannot be imposed.
- 5. "Felony" means an offense for which a sentence to a term of imprisonment in excess of one year may be imposed.
 - 6. "Crime" means a misdemeanor or a felony.
- 7. "Person" means a human being, and where appropriate, a public or private corporation, an unincorporated association, a partnership, a government or a governmental instrumentality.
- 8. "Possess" means to have physical possession or otherwise to exercise dominion or control over tangible property.
- 9. "Physical injury" means impairment of physical condition or substantial pain.
- 10. "Serious physical injury" means physical injury which creates a substantial risk of death, or which causes death or serious and protracted disfigurement, protracted impairment of health or

protracted loss or impairment of the function of any bodily organ.

- 11. "Deadly physical force" means physical force which, under the circumstances in which it is used, is readily capable of causing death or other serious physical injury.
- 12. "Deadly weapon" means any loaded weapon from which a shot, readily capable of producing death or other serious physical injury, may be discharged, or a switchblade knife, gravity knife, pilum ballistic knife, metal knuckle knife, dagger, billy, blackjack, plastic knuckles, or metal knuckles.
- 13. "Dangerous instrument" means any instrument, article or substance, including a "vehicle" as that term is defined in this section, which, under the circumstances in which it is used, attempted to be used or threatened to be used, is readily capable of causing death or other serious physical injury....
- 18. "Juvenile offender" means (1) a person thirteen years old who is criminally responsible for acts constituting murder in the second degree as defined in subdivisions one and two of section 125.25 of this chapter or such conduct as a sexually motivated felony, where authorized pursuant to section 130.91 of the penal law; and
- (2) a person fourteen or fifteen years old who is criminally responsible for acts constituting the crimes defined in subdivisions one and two of section 125.25 (murder in the second degree) and in subdivision three of such section provided that the underlying crime for the murder charge is one for which such person is criminally responsible; section 135.25 (kidnapping in the first degree); 150.20 (arson in the first degree); subdivisions one and two of section 120.10 (assault in the first degree); 125.20 (manslaughter in the first degree); subdivisions one and two of section 130. 35 (rape in the first degree); subdivisions one and two of section 130.50 (criminal sexual act in the first degree); 130.70 (aggravated sexual abuse in the first degree); 140.30 (burglary in the first degree); subdivision one of section 140.25 (burglary in the second degree); 150.15 (arson in the second degree); 160.15 (robbery in the first degree); subdivision two of section 160.10 (robbery in the second degree) of this chapter; or section 265.03 of this chapter, where such machine gun or such firearm is possessed on school grounds, as that phrase is defined in subdivision fourteen of section 220.00 of this chapter; or defined in this chapter as

an attempt to commit murder in the second degree or kidnapping in the first degree, or such conduct as a sexually motivated felony, where authorized pursuant to section 130.91 of the penal law.

TITLE B. PRINCIPLES OF CRIMINAL LIABILITY

Article 15. Culpability

§ 15.00 Culpability; definitions of terms

The following definitions are applicable to this chapter:

- 1. "Act" means a bodily movement.
- 2. "Voluntary act" means a bodily movement performed consciously as a result of effort or determination, and includes the possession of property if the actor was aware of his physical possession or control thereof for a sufficient period to have been able to terminate it.
- 3. "Omission" means a failure to perform an act as to which a duty of performance is imposed by law
- 4. "Conduct" means an act or omission and its accompanying mental state.
- 5. "To act" means either to perform an act or to omit to perform an act.
- 6. "Culpable mental state" means "intentionally" or "knowingly" or "recklessly" or with "criminal negligence," as these terms are defined in section 15.05.

§ 15.05 Culpability; definitions of culpable mental states

The following definitions are applicable to this chapter:

- 1. "Intentionally." A person acts intentionally with respect to a result or to conduct described by a statute defining an offense when his conscious objective is to cause such result or to engage in such conduct.
- 2. "Knowingly." A person acts knowingly with respect to conduct or to a circumstance described by a statute defining an offense when he is aware that his conduct is of such nature or that such circumstance exists.
- 3. "Recklessly." A person acts recklessly with respect to a result or to a circumstance described

by a statute defining an offense when he is aware of and consciously disregards a substantial and unjustifiable risk that such result will occur or that such circumstance exists. The risk must be of such nature and degree that disregard thereof constitutes a gross deviation from the standard of conduct that a reasonable person would observe in the situation. A person who creates such a risk but is unaware thereof solely by reason of voluntary intoxication also acts recklessly with respect thereto.

4. "Criminal negligence." A person acts with criminal negligence with respect to a result or to a circumstance described by a statute defining an offense when he fails to perceive a substantial and unjustifiable risk that such result will occur or that such circumstance exists. The risk must be of such nature and degree that the failure to perceive it constitutes a gross deviation from the standard of care that a reasonable person would observe in the situation.

§ 15.10 Requirements for criminal liability in general and for offenses of strict liability and mental culpability

The minimal requirement for criminal liability is the performance by a person of conduct which includes a voluntary act or the omission to perform an act which he is physically capable of performing. If such conduct is all that is required for commission of a particular offense, or if an offense or some material element thereof does not require a culpable mental state on the part of the actor, such offense is one of "strict liability." If a culpable mental state on the part of the actor is required with respect to every material element of an offense, such offense is one of "mental culpability."

§ 15.15 Construction of statutes with respect to culpability requirements

1. When the commission of an offense defined in this chapter, or some element of an offense, requires a particular culpable mental state, such mental state is ordinarily designated in the statute defining the offense by use of the terms "intentionally," "knowingly," "recklessly" or "criminal negligence," or by use of terms, such as "with intent to defraud" and "knowing it to be false," describing a specific kind of intent or knowledge. When one and only one of such terms appears in a statute defining an offense, it is presumed to apply to every element of the offense unless an intent to limit its application clearly appears.

2. Although no culpable mental state is expressly designated in a statute defining an offense, a culpable mental state may nevertheless be required for the commission of such offense, or with respect to some or all of the material elements thereof, if the proscribed conduct necessarily involves such culpable mental state. A statute defining a crime, unless clearly indicating a legislative intent to impose strict liability, should be construed as defining a crime of mental culpability. This subdivision applies to offenses defined both in and outside this chapter.

§ 15.20 Effect of ignorance or mistake upon liability

- 1. A person is not relieved of criminal liability for conduct because he engages in such conduct under a mistaken belief of fact, unless:
- (a) Such factual mistake negatives the culpable mental state required for the commission of an offense; or
- (b) The statute defining the offense or a statute related thereto expressly provides that such factual mistake constitutes a defense or exemption; or
- (c) Such factual mistake is of a kind that supports a defense of justification as defined in article thirty-five of this chapter.
- 2. A person is not relieved of criminal liability for conduct because he engages in such conduct under a mistaken belief that it does not, as a matter of law, constitute an offense, unless such mistaken belief is founded upon an official statement of the law contained in (a) a statute or other enactment, or (b) an administrative order or grant of permission, or (c) a judicial decision of a state or federal court, or (d) an interpretation of the statute or law relating to the offense, officially made or issued by a public servant, agency or body legally charged or empowered with the responsibility or privilege of administering, enforcing or interpreting such statute or law.
- 3. Notwithstanding the use of the term "knowingly" in any provision of this chapter defining an offense in which the age of a child is an element thereof, knowledge by the defendant of the age of such child is not an element of any such offense and it is not, unless expressly so provided, a defense to a prosecution therefor that the defendant did not know the age of the child or believed such age to be the same as or greater than that specified in the statute.

4. Notwithstanding the use of the term "knowingly" in any provision of this chapter defining an offense in which the aggregate weight of a controlled substance or marihuana is an element, knowledge by the defendant of the aggregate weight of such controlled substance or marihuana is not an element of any such offense and it is not, unless expressly so provided, a defense to a prosecution therefor that the defendant did not know the aggregate weight of the controlled substance or marihuana.

§ 15.25 Effect of intoxication upon liability

Intoxication is not, as such, a defense to a criminal charge; but in any prosecution for an offense, evidence of intoxication of the defendant may be offered by the defendant whenever it is relevant to negative an element of the crime charged.

Article 20. Parties to Offenses and Liability through Accessorial Conduct

§ 20.00 Criminal liability for conduct of another

When one person engages in conduct which constitutes an offense, another person is criminally liable for such conduct when, acting with the mental culpability required for the commission thereof, he solicits, requests, commands, importunes, or intentionally aids such person to engage in such conduct.

§ 20.05 Criminal liability for conduct of another; no defense

In any prosecution for an offense in which the criminal liability of the defendant is based upon the conduct of another person pursuant to section 20.00, it is no defense that:

- 1. Such other person is not guilty of the offense in question owing to criminal irresponsibility or other legal incapacity or exemption, or to unawareness of the criminal nature of the conduct in question or of the defendant's criminal purpose or to other factors precluding the mental state required for the commission of the offense in question; or
- 2. Such other person has not been prosecuted for or convicted of any offense based upon the conduct in question, or has previously been acquitted thereof, or has legal immunity from prosecution therefor; or
- 3. The offense in question, as defined, can be committed only by a particular class or classes of

persons, and the defendant, not belonging to such class or classes, is for that reason legally incapable of committing the offense in an individual capacity.

§ 20.10 Criminal liability for conduct of another; exemption

Notwithstanding the provisions of sections 20.00 and 20.05, a person is not criminally liable for conduct of another person constituting an offense when his own conduct, though causing or aiding the commission of such offense, is of a kind that is necessarily incidental thereto. If such conduct constitutes a related but separate offense upon the part of the actor, he is liable for that offense only and not for the conduct or offense committed by the other person.

§ 20.15 Convictions for different degrees of offense

Except as otherwise expressly provided in this chapter, when, pursuant to section 20.00, two or more persons are criminally liable for an offense which is divided into degrees, each person is guilty of such degree as is compatible with his own culpable mental state and with his own accountability for an aggravating fact or circumstance.

TITLE C. DEFENSES

Article 25. Defenses in General

§ 25.00 Defenses; burden of proof

- 1. When a "defense," other than an "affirmative defense," defined by statute is raised at a trial, the people have the burden of disproving such defense beyond a reasonable doubt.
- 2. When a defense declared by statute to be an "affirmative defense" is raised at a trial, the defendant has the burden of establishing such defense by a preponderance of the evidence.

Article 30. Defense of Infancy

§ 30.00 Infancy

- 1. Except as provided in subdivision two of this section, a person less than sixteen years old is not criminally responsible for conduct.
- 2. A person thirteen, fourteen or fifteen years of age is criminally responsible for acts constituting murder in the second degree as defined in subdivisions one and two of section 125.25 and in subdivision three of such section provided that the

underlying crime for the murder charge is one for which such person is criminally responsible or for such conduct as a sexually motivated felony, where authorized pursuant to section 130.91 of the penal law; and a person fourteen or fifteen years of age is criminally responsible for acts constituting the crimes defined in section 135.25 (kidnapping in the first degree); 150.20 (arson in the first degree); subdivisions one and two of section 120.10 (assault in the first degree); 125.20 (manslaughter in the first degree); subdivisions one and two of section 130.35 (rape in the first degree); subdivisions one and two of section 130.50 (criminal sexual act in the first degree); 130.70 (aggravated sexual abuse in the first degree); 140.30 (burglary in the first degree); subdivision one of section 140.25 (burglary in the second degree); 150.15 (arson in the second degree); 160.15 (robbery in the first degree); subdivision two of section 160.10 (robbery in the second degree) of this chapter; or section 265.03 of this chapter, where such machine gun or such firearm is possessed on school grounds, as that phrase is defined in subdivision fourteen of section 220.00 of this chapter; or defined in this chapter as an attempt to commit murder in the second degree or kidnapping in the first degree, or for such conduct as a sexually motivated felony, where authorized pursuant to section 130.91 of the penal law.

3. In any prosecution for an offense, lack of criminal responsibility by reason of infancy, as defined in this section, is a defense.

Article 35. Defense of Justification

§ 35.00 Justification; a defense

In any prosecution for an offense, justification, as defined in sections 35.05 through 35.30, is a defense.

§ 35.05 Justification; generally

Unless otherwise limited by the ensuing provisions of this article defining justifiable use of physical force, conduct which would otherwise constitute an offense is justifiable and not criminal when:

- 1. Such conduct is required or authorized by law or by a judicial decree, or is performed by a public servant in the reasonable exercise of his official powers, duties or functions; or
- 2. Such conduct is necessary as an emergency measure to avoid an imminent public or private

injury which is about to occur by reason of a situation occasioned or developed through no fault of the actor, and which is of such gravity that, according to ordinary standards of intelligence and morality, the desirability and urgency of avoiding such injury clearly outweigh the desirability of avoiding the injury sought to be prevented by the statute defining the offense in issue. The necessity and justifiability of such conduct may not rest upon considerations pertaining only to the morality and advisability of the statute, either in its general application or with respect to its application to a particular class of cases arising thereunder. Whenever evidence relating to the defense of justification under this subdivision is offered by the defendant, the court shall rule as a matter of law whether the claimed facts and circumstances would, if established, constitute a defense.

§ 35.10 Justification; use of physical force generally

The use of physical force upon another person which would otherwise constitute an offense is justifiable and not criminal under any of the following circumstances:

- 1. A parent, guardian or other person entrusted with the care and supervision of a person under the age of twenty-one or an incompetent person, and a teacher or other person entrusted with the care and supervision of a person under the age of twenty-one for a special purpose, may use physical force, but not deadly physical force, upon such person when and to the extent that he reasonably believes it necessary to maintain discipline or to promote the welfare of such person.
- 2. A warden or other authorized official of a jail, prison or correctional institution may, in order to maintain order and discipline, use such physical force as is authorized by the correction law.
- 3. A person responsible for the maintenance of order in a common carrier of passengers, or a person acting under his direction, may use physical force when and to the extent that he reasonably believes it necessary to maintain order, but he may use deadly physical force only when he reasonably believes it necessary to prevent death or serious physical injury.
- 4. A person acting under a reasonable belief that another person is about to commit suicide or to inflict serious physical injury upon himself may use physical force upon such person to the extent that he reasonably believes it necessary to thwart such

result.

- 5. A duly licensed physician, or a person acting under a physician's direction, may use physical force for the purpose of administering a recognized form of treatment which he or she reasonably believes to be adapted to promoting the physical or mental health of the patient if (a) the treatment is administered with the consent of the patient or, if the patient is under the age of eighteen years or an incompetent person, with the consent of the parent, guardian or other person entrusted with the patient's care and supervision, or (b) the treatment is administered in an emergency when the physician reasonably believes that no one competent to consent can be consulted and that a reasonable person, wishing to safeguard the welfare of the patient, would consent.
- 6. A person may, pursuant to the ensuing provisions of this article, use physical force upon another person in self-defense or defense of a third person, or in defense of premises, or in order to prevent larceny of or criminal mischief to property, or in order to effect an arrest or prevent an escape from custody. Whenever a person is authorized by any such provision to use deadly physical force in any given circumstance, nothing contained in any other such provision may be deemed to negate or qualify such authorization.

§ 35.15 Justification; use of physical force in defense of a person

- 1. A person may, subject to the provisions of subdivision two, use physical force upon another person when and to the extent he or she reasonably believes such to be necessary to defend himself, herself or a third person from what he or she reasonably believes to be the use or imminent use of unlawful physical force by such other person, unless:
- (a) The latter's conduct was provoked by the actor with intent to cause physical injury to another person; or
- (b) The actor was the initial aggressor; except that in such case the use of physical force is nevertheless justifiable if the actor has withdrawn from the encounter and effectively communicated such withdrawal to such other person but the latter persists in continuing the incident by the use or threatened imminent use of unlawful physical force; or
 - (c) The physical force involved is the product of a

combat by agreement not specifically authorized by law.

- 2. A person may not use deadly physical force upon another person under circumstances specified in subdivision one unless:
- (a) The actor reasonably believes that such other person is using or about to use deadly physical force. Even in such case, however, the actor may not use deadly physical force if he or she knows that with complete personal safety, to oneself and others he or she may avoid the necessity of so doing by retreating; except that the actor is under no duty to retreat if he or she is:
- (i) in his or her dwelling and not the initial aggressor; or
- (ii) a police officer or peace officer or a person assisting a police officer or a peace officer at the latter's direction, acting pursuant to section 35.30; or
- (b) He or she reasonably believes that such other person is committing or attempting to commit a kidnapping, forcible rape, forcible criminal sexual act or robbery; or
- (c) He or she reasonably believes that such other person is committing or attempting to commit a burglary, and the circumstances are such that the use of deadly physical force is authorized by subdivision three of section 35.20.

§ 35.20 Justification; use of physical force in defense of premises and in defense of a person in the course of burglary

- 1. Any person may use physical force upon another person when he or she reasonably believes such to be necessary to prevent or terminate what he or she reasonably believes to be the commission or attempted commission by such other person of a crime involving damage to premises. Such person may use any degree of physical force, other than deadly physical force, which he or she reasonably believes to be necessary for such purpose, and may use deadly physical force if he or she reasonably believes such to be necessary to prevent or terminate the commission or attempted commission of arson.
- 2. A person in possession or control of any premises, or a person licensed or privileged to be thereon or therein, may use physical force upon another person when he or she reasonably believes such to be necessary to prevent or

terminate what he or she reasonably believes to be the commission or attempted commission by such other person of a criminal trespass upon such premises. Such person may use any degree of physical force, other than deadly physical force, which he or she reasonably believes to be necessary for such purpose, and may use deadly physical force in order to prevent or terminate the commission or attempted commission of arson, as prescribed in subdivision one, or in the course of a burglary or attempted burglary, as prescribed in subdivision three.

- 3. A person in possession or control of, or licensed or privileged to be in, a dwelling or an occupied building, who reasonably believes that another person is committing or attempting to commit a burglary of such dwelling or building, may use deadly physical force upon such other person when he or she reasonably believes such to be necessary to prevent or terminate the commission or attempted commission of such burglary.
- 4. As used in this section, the following terms have the following meanings:
- (a) The terms "premises," "building" and "dwelling" have the meanings prescribed in section 140.00;
- (b) Persons "licensed or privileged" to be in buildings or upon other premises include, but are not limited to:
- (i) police officers or peace officers acting in the performance of their duties; and
- (ii) security personnel or employees of nuclear powered electric generating facilities located within the state who are employed as part of any security plan approved by the federal operating license agencies acting in the performance of their duties at such generating facilities. For purposes of this subparagraph, the term "nuclear powered electric generating facility" shall mean a facility that generates electricity using nuclear power for sale, directly or indirectly, to the public, including the land upon which the facility is located and the safety and security zones as defined under federal regulations....

Article 40. Other Defenses Involving Lack of Culpability

§ 40.00 Duress

1. In any prosecution for an offense, it is an

affirmative defense that the defendant engaged in the proscribed conduct because he was coerced to do so by the use or threatened imminent use of unlawful physical force upon him or a third person, which force or threatened force a person of reasonable firmness in his situation would have been unable to resist.

2. The defense of duress as defined in subdivision one of this section is not available when a person intentionally or recklessly places himself in a situation in which it is probable that he will be subjected to duress.

§ 40.05 Entrapment

In any prosecution for an offense, it is an affirmative defense that the defendant engaged in the proscribed conduct because he was induced or encouraged to do so by a public servant, or by a person acting in cooperation with a public servant, seeking to obtain evidence against him for purpose of criminal prosecution, and when the methods used to obtain such evidence were such as to create a substantial risk that the offense would be committed by a person not otherwise disposed to commit it. Inducement or encouragement to commit an offense means active inducement or encouragement. Conduct merely affording a person an opportunity to commit an offense does not constitute entrapment.

§ 40.10 Renunciation

- 1. In any prosecution for an offense, other than an attempt to commit a crime, in which the defendant's guilt depends upon his criminal liability for the conduct of another person pursuant to section 20.00, it is an affirmative defense that, under circumstances manifesting a voluntary and complete renunciation of his criminal purpose, the defendant withdrew from participation in such offense prior to the commission thereof and made a substantial effort to prevent the commission thereof.
- 2. In any prosecution for criminal facilitation pursuant to article one hundred fifteen, it is an affirmative defense that, prior to the commission of the felony which he facilitated, the defendant made a substantial effort to prevent the commission of such felony.
- 3. In any prosecution pursuant to section 110.00 for an attempt to commit a crime, it is an affirmative defense that, under circumstances manifesting a voluntary and complete renunciation of his criminal

purpose, the defendant avoided the commission of the crime attempted by abandoning his criminal effort and, if mere abandonment was insufficient to accomplish such avoidance, by taking further and affirmative steps which prevented the commission thereof.

- 4. In any prosecution for criminal solicitation pursuant to article one hundred or for conspiracy pursuant to article one hundred five in which the crime solicited or the crime contemplated by the conspiracy was not in fact committed, it is an affirmative defense that, under circumstances manifesting a voluntary and complete renunciation of his criminal purpose, the defendant prevented the commission of such crime.
- 5. A renunciation is not "voluntary and complete" within the meaning of this section if it is motivated in whole or in part by (a) a belief that circumstances exist which increase the probability of detection or apprehension of the defendant or another participant in the criminal enterprise, or which render more difficult the accomplishment of the criminal purpose, or (b) a decision to postpone the criminal conduct until another time or to transfer the criminal effort to another victim or another but similar objective.

§ 40.15 Mental disease or defect

In any prosecution for an offense, it is an affirmative defense that when the defendant engaged in the proscribed conduct, he lacked criminal responsibility by reason of mental disease or defect. Such lack of criminal responsibility means that at the time of such conduct, as a result of mental disease or defect, he lacked substantial capacity to know or appreciate either:

- 1. The nature and consequences of such conduct; or
 - 2. That such conduct was wrong.

PART TWO. SENTENCES

TITLE E. SENTENCES

Article 55. Classification and Designation of Offenses

§ 55.00 Applicability of article

The provisions of this article govern the classification and designation of every offense, whether defined within or outside of this chapter.

§ 55.05 Classifications of felonies and misdemeanors

- 1. Felonies. Felonies are classified, for the purpose of sentence, into five categories as follows:
 - (a) Class A felonies;
 - (b) Class B felonies;
 - (c) Class C felonies;
 - (d) Class D felonies; and
 - (e) Class E felonies.

Class A felonies are subclassified, for the purpose of sentence, into two categories as follows: subclass I and subclass II, to be known as class A-I and class A-II felonies, respectively.

- 2. Misdemeanors. Misdemeanors are classified, for the purpose of sentence, into three categories as follows:
 - (a) Class A misdemeanors;
 - (b) Class B misdemeanors; and
 - (c) Unclassified misdemeanors.

§ 55.10 Designation of offenses

- 1. Felonies.
- (a) The particular classification or subclassification of each felony defined in this chapter is expressly designated in the section or article defining it.
- (b) Any offense defined outside this chapter which is declared by law to be a felony without specification of the classification thereof, or for which a law outside this chapter provides a sentence to a term of imprisonment in excess of one year, shall be deemed a class E felony.
 - 2. Misdemeanors.
- (a) Each misdemeanor defined in this chapter is either a class A misdemeanor or a class B misdemeanor, as expressly designated in the section or article defining it.
- (b) Any offense defined outside this chapter which is declared by law to be a misdemeanor without specification of the classification thereof or of the sentence therefor shall be deemed a class A misdemeanor.
- (c) Except as provided in paragraph (b) of subdivision three, where an offense is defined

outside this chapter and a sentence to a term of imprisonment in excess of fifteen days but not in excess of one year is provided in the law or ordinance defining it, such offense shall be deemed an unclassified misdemeanor.

- 3. Violations. Every violation defined in this chapter is expressly designated as such. Any offense defined outside this chapter which is not expressly designated a violation shall be deemed a violation if:
- (a) Notwithstanding any other designation specified in the law or ordinance defining it, a sentence to a term of imprisonment which is not in excess of fifteen days is provided therein, or the only sentence provided therein is a fine; or
- (b) A sentence to a term of imprisonment in excess of fifteen days is provided for such offense in a law or ordinance enacted prior to the effective date of this chapter but the offense was not a crime prior to that date.
- 4. Traffic infraction. Notwithstanding any other provision of this section, an offense which is defined as a "traffic infraction" shall not be deemed a violation or a misdemeanor by virtue of the sentence prescribed therefor.

Article 60. Authorized Dispositions of Offenders

§ 60.00 Applicability of provisions

- 1. The sentences prescribed by this article shall apply in the case of every offense, whether defined within or outside of this chapter.
- 2. The sole provision of this article that shall apply in the case of an offense committed by a juvenile offender is section 60.10 of this article and no other provisions of this article shall be deemed or construed to apply in any such case.

§ 60.01 Authorized dispositions; generally

- 1. Applicability. Except as otherwise specified in this article, when the court imposes sentence upon a person convicted of an offense, the court must impose a sentence prescribed by this section.
 - 2. Revocable dispositions.
- (a) The court may impose a revocable sentence as herein specified:
- (i) the court, where authorized by article sixty-five, may sentence a person to a period of probation or to a period of conditional discharge as provided in

that article: or

- (ii) the court, where authorized by article eighty-five, may sentence a person to a term of intermittent imprisonment as provided in that article.
- (b) A revocable sentence shall be deemed a tentative one to the extent that it may be altered or revoked in accordance with the provisions of the article under which it was imposed, but for all other purposes shall be deemed to be a final judgment of conviction.
- (c) In any case where the court imposes a sentence of probation, conditional discharge, or a sentence of intermittent imprisonment, it may also impose a fine authorized by article eighty.
- (d) In any case where the court imposes a sentence of imprisonment not in excess of sixty days, for a misdemeanor or not in excess of six months for a felony or in the case of a sentence of intermittent imprisonment not in excess of four months, it may also impose a sentence of probation or conditional discharge provided that the term of probation or conditional discharge together with the term of imprisonment shall not exceed the term of probation or conditional discharge authorized by article sixty-five of this chapter. The sentence of imprisonment shall be a condition of and run concurrently with the sentence of probation or conditional discharge.
- 3. Other dispositions. When a person is not sentenced as specified in subdivision two, or when a sentence specified in subdivision two is revoked, the sentence of the court must be as follows:
 - (a) A term of imprisonment; or
- (b) A fine authorized by article eighty, provided, however, that when the conviction is of a class B felony or of any felony defined in article two hundred twenty, the sentence shall not consist solely of a fine; or
 - (c) Both imprisonment and a fine; or
- (d) Where authorized by section 65.20, unconditional discharge as provided in that section; or
- (e) Following revocation of a sentence of conditional discharge imposed pursuant to section 65.05 of this chapter or paragraph (d) of subdivision two of this section, probation as provided in section 65.00 of this chapter or to the sentence of imprisonment and probation as

provided for in paragraph (d) of subdivision two of this section.

4. In any case where a person has been sentenced to a period of probation imposed pursuant to section 65.00 of this chapter, if the part of the sentence that provides for probation is revoked, the court must sentence such person to imprisonment or to the sentence of imprisonment and probation as provided for in paragraph (d) of subdivision two of this section.

§ 60.02 Authorized disposition; youthful offender

When a person is to be sentenced upon a youthful offender finding, the court must impose a sentence as follows:

- (1) If the sentence is to be imposed upon a youthful offender finding which has been substituted for a conviction of an offense other than a felony, the court must impose a sentence authorized for the offense for which the youthful offender finding was substituted, except that if the youthful offender finding was entered pursuant to paragraph (b) of subdivision one of section 720.20 of the criminal procedure law, the court must not impose a definite or intermittent sentence of imprisonment with a term of more than six months; or
- (2) If the sentence is to be imposed upon a youthful offender finding which has been substituted for a conviction for any felony, the court must impose a sentence authorized to be imposed upon a person convicted of a class E felony provided, however, that the court must not impose a sentence of conditional discharge or unconditional discharge if the youthful offender finding was substituted for a conviction of a felony defined in article two hundred twenty of this chapter.
- (3) The provisions of section 60.35 of this article shall apply to a sentence imposed upon a youthful offender finding and the amount of the mandatory surcharge and crime victim assistance fee which shall be levied at sentencing shall be equal to the amount specified in such section for the offense of conviction for which the youthful offender finding was substituted; provided, however that the court shall not impose the sex offender registration fee, DNA databank fee or supplemental sex offender victim fee, as defined in subparagraphs (iv) and (v) of paragraph (a) and paragraph (b) of subdivision one of section 60.35 of this article, for an offense in

which the conviction was substituted with a youthful offender finding.

§ 60.04 Authorized disposition; controlled substances and marihuana felony offenses

- 1. Applicability. Notwithstanding the provisions of any law, this section shall govern the dispositions authorized when a person is to be sentenced upon a conviction of a felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter or when a person is to be sentenced upon a conviction of such a felony as a multiple felony offender as defined in subdivision five of this section.
- 2. Class A felony. Every person convicted of a class A felony must be sentenced to imprisonment in accordance with section 70.71 of this title, unless such person is convicted of a class A-II felony and is sentenced to probation for life in accordance with section 65.00 of this title.
- 3. Class B felonies. Every person convicted of a class B felony must be sentenced to imprisonment in accordance with the applicable provisions of section 70.70 of this chapter, a definite sentence of imprisonment with a term of one year or less or probation in accordance with section 65.00 of this chapter provided, however, a person convicted of criminal sale of a controlled substance to a child as defined in section 220.48 of this chapter must be sentenced to a determinate sentence of imprisonment in accordance with the applicable provisions of section 70.70 of this chapter or to a sentence of probation in accordance with the opening paragraph of paragraph (b) of subdivision one of section 65.00 of this chapter.
- 4. Alternative sentence. Where a sentence of imprisonment or a sentence of probation as an alternative to imprisonment is not required to be imposed pursuant to subdivision two, three or five of this section, the court may impose any other sentence authorized by section 60.01 of this article, provided that when the court imposes a sentence of imprisonment, such sentence must be in accordance with section 70.70 of this title. Where the court imposes a sentence of imprisonment in accordance with this section, the court may also impose a fine authorized by article eighty of this title and in such case the sentence shall be both imprisonment and a fine.
- 5. Multiple felony offender. Where the court imposes a sentence pursuant to subdivision three of section 70.70 of this chapter upon a second

felony drug offender, as defined in paragraph (b) of subdivision one of section 70.70 of this chapter, it must sentence such offender to imprisonment in accordance with the applicable provisions of section 70.70 of this chapter, a definite sentence of imprisonment with a term of one year or less, or probation in accordance with section 65.00 of this chapter, provided, however, that where the court imposes a sentence upon a class B second felony drug offender, it must sentence such offender to a determinate sentence of imprisonment in accordance with the applicable provisions of section 70.70 of this chapter or to a sentence of probation in accordance with the opening paragraph of paragraph (b) of subdivision one of section 65.00 of this chapter. When the court imposes sentence on a second felony drug offender pursuant to subdivision four of section 70.70 of this chapter, it must impose a determinate sentence of imprisonment in accordance with such subdivision.

- 6. Substance abuse treatment. When the court imposes a sentence of imprisonment which requires a commitment to the state department of corrections and community supervision upon a person who stands convicted of a controlled substance or marihuana offense, the court may, upon motion of the defendant in its discretion, issue an order directing that the department of corrections and community supervision enroll the defendant in the comprehensive alcohol and substance abuse treatment program in an alcohol and substance abuse correctional annex as defined in subdivision eighteen of section two of the correction law, provided that the defendant will satisfy the statutory eligibility criteria for participation in such program. Notwithstanding the foregoing provisions of this subdivision, any defendant to be enrolled in such program pursuant to this subdivision shall be governed by the same rules and regulations promulgated by the department of corrections and community supervision, including without limitation those rules and regulations establishing requirements for completion and those rules and regulations governing discipline and removal from the program. No such period of court ordered corrections based drug abuse treatment pursuant to this subdivision shall be required to extend beyond the defendant's conditional release date.
- 7. a. Shock incarceration participation. When the court imposes a sentence of imprisonment which requires a commitment to the department of

- corrections and community supervision upon a person who stands convicted of a controlled substance or marihuana offense, upon motion of the defendant, the court may issue an order directing that the department of corrections and community supervision enroll the defendant in the shock incarceration program as defined in article twenty-six-A of the correction law, provided that the defendant is an eligible inmate, as described in subdivision one of section eight hundred sixty-five of the correction law. Notwithstanding the foregoing provisions of this subdivision, any defendant to be enrolled in such program pursuant to this subdivision shall be governed by the same rules and regulations promulgated by the department of corrections and community supervision, including without limitation those rules and regulations establishing requirements for completion and such rules and regulations governing discipline and removal from the program.
- b. (i) In the event that an inmate designated by court order for enrollment in the shock incarceration program requires a degree of medical care or mental health care that cannot be provided at a shock incarceration facility, the department, in writing, shall notify the inmate, provide a proposal describing a proposed alternative-to-shock-incarceration program, and notify him or her that he or she may object in writing to placement in such alternative-to-shock-incarceration program. If the inmate objects in writing to placement in such alternative-to-shock-incarceration program, the department of corrections and community supervision shall notify the sentencing court, provide such proposal to the court, and arrange for the inmate's prompt appearance before the court. The court shall provide the proposal and notice of a court appearance to the people, the inmate and the appropriate defense attorney. After considering the proposal and any submissions by the parties, and after a reasonable opportunity for the people, the inmate and counsel to be heard, the court may modify its sentencing order accordingly, notwithstanding the provisions of section 430.10 of the criminal procedure law.
- (ii) An inmate who successfully completes an alternative-to-shock-incarceration program within the department of corrections and community supervision shall be treated in the same manner as a person who has successfully completed the shock incarceration program, as set forth in subdivision four of section eight hundred

sixty-seven of the correction law.

§ 60.05 Authorized dispositions; other class A, B, certain C and D felonies and multiple felony offenders

- 1. Applicability. Except as provided in section 60.04 of this article governing the authorized dispositions applicable to felony offenses defined in article two hundred twenty or two hundred twenty-one of this chapter or in section 60.13 of this article governing the authorized dispositions applicable to felony sex offenses defined in paragraph (a) of subdivision one of section 70.80 of this title, this section shall govern the dispositions authorized when a person is to be sentenced upon a conviction of a class A felony, a class B felony or a class C, class D or class E felony specified herein, or when a person is to be sentenced upon a conviction of a felony as a multiple felony offender.
- 2. Class A felony. Except as provided in subdivisions three and four of section 70.06 of this chapter, every person convicted of a class A felony must be sentenced to imprisonment in accordance with section 70.00 of this title, unless such person is convicted of murder in the first degree and is sentenced in accordance with section 60.06 of this article.
- 3. Class B felony. Except as provided in subdivision six of this section, every person convicted of a class B violent felony offense as defined in subdivision one of section 70.02 of this title, must be sentenced to imprisonment in accordance with such section 70.02; and, except as provided in subdivision six of this section, every person convicted of any other class B felony must be sentenced to imprisonment in accordance with section 70.00 of this title.
- 4. Certain class C felonies. Except as provided in subdivision six, every person convicted of a class C violent felony offense as defined in subdivision one of section 70.02 of this title, must be sentenced to imprisonment in accordance with section 70.02 of this title; and, except as provided in subdivision six of this section, every person convicted of the class C felonies of: attempt to commit any of the class B felonies of bribery in the first degree as defined in section 200.04, bribe receiving in the first degree as defined in section degree as defined in section 105.15 and criminal mischief in the first degree as defined in section 145.12; criminal usury in the first degree as

- defined in section 190.42, rewarding official misconduct in the first degree as defined in section 200.22, receiving reward for official misconduct in the first degree as defined in section 200.27, attempt to promote prostitution in the first degree as defined in section 230.32, promoting prostitution in the second degree as defined in section 230.30, arson in the third degree as defined in section 150.10 of this chapter, must be sentenced to imprisonment in accordance with section 70.00 of this title.
- 5. Certain class D felonies. Except as provided in subdivision six of this section, every person convicted of the class D felonies of assault in the second degree as defined in section 120.05, strangulation in the second degree as defined in section 121.12 or attempt to commit a class C felony as defined in section 230.30 of this chapter, must be sentenced in accordance with section 70.00 or 85.00 of this title.
- 6. Multiple felony offender. When the court imposes sentence upon a second violent felony offender, as defined in section 70.04, or a second felony offender, as defined in section 70.06, the court must impose a sentence of imprisonment in accordance with section 70.04 or 70.06, as the case may be, unless it imposes a sentence of imprisonment in accordance with section 70.08 or 70.10.
- 7. Fines. Where the court imposes a sentence of imprisonment in accordance with this section, the court also may impose a fine authorized by article eighty and in such case the sentence shall be both imprisonment and a fine.
- § 60.06 Authorized disposition; murder in the first degree offenders; aggravated murder offenders; certain murder in the second degree offenders; certain terrorism offenders; criminal possession of a chemical weapon or biological weapon offenders; criminal use of a chemical weapon or biological weapon offenders

When a defendant is convicted of murder in the first degree as defined in section 125.27 of this chapter, the court shall, in accordance with the provisions of section 400.27 of the criminal procedure law, sentence the defendant to death, to life imprisonment without parole in accordance with subdivision five of section 70.00 of this title, or to a term of imprisonment for a class A-I felony other than a sentence of life imprisonment without parole, in accordance with subdivisions one

through three of section 70.00 of this title. When a person is convicted of murder in the second degree as defined in subdivision five of section 125.25 of this chapter or of the crime of aggravated murder as defined in subdivision one of section 125.26 of this chapter, the court shall sentence the defendant to life imprisonment without parole in accordance with subdivision five of section 70.00 of this title. When a defendant is convicted of the crime of terrorism as defined in section 490.25 of this chapter, and the specified offense the defendant committed is a class A-I felony offense, or when a defendant is convicted of the crime of criminal possession of a chemical weapon or biological weapon in the first degree as defined in section 490.45 of this chapter, or when a defendant is convicted of the crime of criminal use of a chemical weapon or biological weapon in the first degree as defined in section 490.55 of this chapter, the court shall sentence the defendant to life imprisonment without parole in accordance with subdivision five of section 70.00 of this title; provided, however, that nothing in this section shall preclude or prevent a sentence of death when the defendant is also convicted of murder in the first degree as defined in section 125.27 of this chapter. When a defendant is convicted of aggravated murder as defined in subdivision two of section 125.26 of this chapter. the court shall sentence the defendant to life imprisonment without parole or to a term of imprisonment for a class A-I felony other than a sentence of life imprisonment without parole, in accordance with subdivisions one through three of section 70.00 of this title....

§ 60.10 Authorized disposition; juvenile offender

- 1. When a juvenile offender is convicted of a crime, the court shall sentence the defendant to imprisonment in accordance with section 70.05 or sentence him upon a youthful offender finding in accordance with section 60.02 of this chapter.
- 2. Subdivision one of this section shall apply when sentencing a juvenile offender notwithstanding the provisions of any other law that deals with the authorized sentence for persons who are not juvenile offenders. Provided, however, that the limitation prescribed by this section shall not be deemed or construed to bar use of a conviction of a juvenile offender, other than a juvenile offender who has been adjudicated a youthful offender pursuant to section 720.20 of the criminal procedure law, as a previous or predicate felony

offender under section 70.04, 70.06, 70.08 or 70.10, when sentencing a person who commits a felony after he has reached the age of sixteen.

§ 60.11 Authorized dispositions; criminal possession of a weapon in the fourth degree

When a person is to be sentenced upon a conviction of the crime of criminal possession of a weapon in the fourth degree as defined in subdivision one of section 265.01 as a result of a plea of guilty entered in satisfaction of an indictment or count thereof charging the defendant with the class D violent felony offense of criminal possession of a weapon in the third degree as defined in subdivision four of section 265.02, the court must sentence the defendant in accordance with the provisions of section 70.15....

§ 60.20 Authorized dispositions; traffic infraction

- 1. When a person is convicted of a traffic infraction, the sentence of the court shall be as follows:
- (a) A period of conditional discharge, as provided in article sixty-five; or
- (b) Unconditional discharge as provided in section 65.20; or
- (c) A fine or a sentence to a term of imprisonment, or both, as prescribed in and authorized by the provision that defines the infraction; or
- (d) A sentence of intermittent imprisonment, as provided in article eighty-five.
- 2. Where a sentence of conditional discharge is imposed for a traffic infraction, all incidents of the sentence shall be the same as would be applicable if the sentence were for a violation.

§ 60.21 Authorized dispositions; driving while intoxicated or aggravated driving while intoxicated

Notwithstanding paragraph (d) of subdivision two of section 60.01 of this article, when a person is to be sentenced upon a conviction for a violation of subdivision two, two-a or three of section eleven hundred ninety-two of the vehicle and traffic law, the court may sentence such person to a period of imprisonment authorized by article seventy of this title and shall sentence such person to a period of probation or conditional discharge in accordance with the provisions of section 65.00 of this title and

shall order the installation and maintenance of a functioning ignition interlock device. Such period of probation or conditional discharge shall run consecutively to any period of imprisonment and shall commence immediately upon such person's release from imprisonment....

§ 60.35 Mandatory surcharge, sex offender registration fee, DNA databank fee, supplemental sex offender victim fee and crime victim assistance fee required in certain cases

- 1. (a) Except as provided in section eighteen hundred nine of the vehicle and traffic law and section 27.12 of the parks, recreation and historic preservation law, whenever proceedings in an administrative tribunal or a court of this state result in a conviction for a felony, a misdemeanor, or a violation, as these terms are defined in section 10.00 of this chapter, there shall be levied at sentencing a mandatory surcharge, sex offender registration fee, DNA databank fee and a crime victim assistance fee in addition to any sentence required or permitted by law, in accordance with the following schedule:
- (i) a person convicted of a felony shall pay a mandatory surcharge of three hundred dollars and a crime victim assistance fee of twenty-five dollars;
- (ii) a person convicted of a misdemeanor shall pay a mandatory surcharge of one hundred seventy-five dollars and a crime victim assistance fee of twenty-five dollars;
- (iii) a person convicted of a violation shall pay a mandatory surcharge of ninety-five dollars and a crime victim assistance fee of twenty-five dollars;
- (iv) a person convicted of a sex offense as defined by subdivision two of section one hundred sixty-eight-a of the correction law or a sexually violent offense as defined by subdivision three of section one hundred sixty-eight-a of the correction law shall, in addition to a mandatory surcharge and crime victim assistance fee, pay a sex offender registration fee of fifty dollars.
- (v) a person convicted of a designated offense as defined by subdivision seven of section nine hundred ninety-five of the executive law shall, in addition to a mandatory surcharge and crime victim assistance fee, pay a DNA databank fee of fifty dollars
- (b) When the felony or misdemeanor conviction in subparagraphs (i), (ii) or (iv) of paragraph (a) of this subdivision results from an offense contained in

- article one hundred thirty of this chapter, incest in the third, second or first degree as defined in sections 255.25, 255.26 and 255.27 of this chapter or an offense contained in article two hundred sixty-three of this chapter, the person convicted shall pay a supplemental sex offender victim fee of one thousand dollars in addition to the mandatory surcharge and any other fee.
- 2. Where a person is convicted of two or more crimes or violations committed through a single act or omission, or through an act or omission which in itself constituted one of the crimes or violations and also was a material element of the other, the court shall impose a mandatory surcharge and a crime victim assistance fee, and where appropriate a supplemental sex offender victim fee, in accordance with the provisions of this section for the crime or violation which carries the highest classification, and no other sentence to pay a mandatory surcharge, crime victim assistance fee or supplemental sex offender victim fee required by this section shall be imposed. Where a person is convicted of two or more sex offenses or sexually violent offenses, as defined by subdivisions two and three of section one hundred sixty-eight-a of the correction law, committed through a single act or omission, or through an act or omission which in itself constituted one of the offenses and also was a material element of the other, the court shall impose only one sex offender registration fee. Where a person is convicted of two or more designated offenses, as defined by subdivision seven of section nine hundred ninety-five of the executive law, committed through a single act or omission, or through an act or omission which in itself constituted one of the offenses and also was a material element of the other, the court shall impose only one DNA databank fee.
- 3. The mandatory surcharge, sex offender registration fee, DNA databank fee, crime victim assistance fee, and supplemental sex offender victim fee provided for in subdivision one of this section shall be paid to the clerk of the court or administrative tribunal that rendered the conviction. Within the first ten days of the month following collection of the mandatory surcharge, crime victim assistance fee, and supplemental sex offender victim fee, the collecting authority shall determine the amount of mandatory surcharge, crime victim assistance fee, and supplemental sex offender victim fee collected and, if it is an administrative tribunal, or a town or village justice court, it shall then pay such money to the state comptroller who

shall deposit such money in the state treasury pursuant to section one hundred twenty-one of the state finance law to the credit of the criminal justice improvement account established by section ninety-seven-bb of the state finance law. Within the first ten days of the month following collection of the sex offender registration fee and DNA databank fee, the collecting authority shall determine the amount of the sex offender registration fee and DNA databank fee collected and, if it is an administrative tribunal, or a town or village justice court, it shall then pay such money to the state comptroller who shall deposit such money in the state treasury pursuant to section one hundred twenty-one of the state finance law to the credit of the general fund. If such collecting authority is any other court of the unified court system, it shall, within such period, pay such money attributable to the mandatory surcharge or crime victim assistance fee to the state commissioner of taxation and finance to the credit of the criminal justice improvement account established by section ninety-seven-bb of the state finance law. If such collecting authority is any other court of the unified court system, it shall, within such period, pay such money attributable to the sex offender registration fee and the DNA databank fee to the state commissioner of taxation and finance to the credit of the general fund.

- 4. Any person who has paid a mandatory surcharge, sex offender registration fee, DNA databank fee, a crime victim assistance fee or a supplemental sex offender victim fee under the authority of this section based upon a conviction that is subsequently reversed or who paid a mandatory surcharge, sex offender registration fee, DNA databank fee, a crime victim assistance fee or supplemental sex offender victim fee under the authority of this section which is ultimately determined not to be required by this section shall be entitled to a refund of such mandatory surcharge, sex offender registration fee, DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee upon application to the state comptroller. The state comptroller shall require such proof as is necessary in order to determine whether a refund is required by law.
- 5. When a person who is convicted of a crime or violation and sentenced to a term of imprisonment has failed to pay the mandatory surcharge, sex offender registration fee, DNA databank fee, crime victim assistance fee or supplemental sex offender
- victim fee required by this section, the clerk of the court that rendered the conviction shall notify the superintendent or the municipal official of the facility where the person is confined. The superintendent or the municipal official shall cause any amount owing to be collected from such person during his or her term of imprisonment from moneys to the credit of an inmates' fund or such moneys as may be earned by a person in a work release program pursuant to section eight hundred sixty of the correction law. Such moneys attributable to the mandatory surcharge or crime victim assistance fee shall be paid over to the state comptroller to the credit of the criminal justice improvement account established by section ninety-seven-bb of the state finance law and such moneys attributable to the sex offender registration fee or DNA databank fee shall be paid over to the state comptroller to the credit of the general fund, except that any such moneys collected which are surcharges, sex offender registration fees, DNA databank fees, crime victim assistance fees or supplemental sex offender victim fees levied in relation to convictions obtained in a town or village justice court shall be paid within thirty days after the receipt thereof by the superintendent or municipal official of the facility to the justice of the court in which the conviction was obtained. For the purposes of collecting such mandatory surcharge, sex offender registration fee, DNA databank fee, crime victim assistance fee, and supplemental sex offender victim fee, the state shall be legally entitled to the money to the credit of an inmates' fund or money which is earned by an inmate in a work release program. For purposes of this subdivision, the term "inmates' fund" shall mean moneys in the possession of an inmate at the time of his or her admission into such facility, funds earned by him or her as provided for in section one hundred eighty-seven of the correction law and any other funds received by him or her or on his or her behalf and deposited with such superintendent or municipal official.
- (b) The incarceration fee provided for in subdivision two of section one hundred eighty-nine of the correction law shall not be assessed or collected if any order of restitution or reparation, fine, mandatory surcharge, sex offender registration fee, DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee remains unpaid. In such circumstances, any monies which may lawfully be withheld from the compensation paid to a prisoner for work

performed while housed in a general confinement facility in satisfaction of such an obligation shall first be applied toward satisfaction of such obligation.

- 6. Notwithstanding any other provision of this section, where a person has made restitution or reparation pursuant to section 60.27 of this article, such person shall not be required to pay a mandatory surcharge or a crime victim assistance fee.
- 7. Notwithstanding the provisions of subdivision one of section 60.00 of this article, the provisions of subdivision one of this section shall not apply to a violation under any law other than this chapter.
- 8. Subdivision one of section 130.10 of the criminal procedure law notwithstanding, at the time that the mandatory surcharge, sex offender registration fee or DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee is imposed a town or village court may, and all other courts shall, issue and cause to be served upon the person required to pay the mandatory surcharge, sex offender registration fee or DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee, a summons directing that such person appear before the court regarding the payment of the mandatory surcharge, sex offender registration fee or DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee, if after sixty days from the date it was imposed it remains unpaid. The designated date of appearance on the summons shall be set for the first day court is in session falling after the sixtieth day from the imposition of the mandatory surcharge, sex offender registration fee or DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee. The summons shall contain the information required by subdivision two of section 130.10 of the criminal procedure law except that in substitution for the requirement of paragraph (c) of such subdivision the summons shall state that the person served must appear at a date, time and specific location specified in the summons if after sixty days from the date of issuance the mandatory surcharge, sex offender registration fee or DNA databank fee, crime victim assistance fee or supplemental sex offender victim fee remains unpaid. The court shall not issue a summons under this subdivision to a person who is being sentenced to a term of confinement in excess of sixty days in jail or in the department of corrections and community supervision. The mandatory surcharges, sex

offender registration fee and DNA databank fees, crime victim assistance fees and supplemental sex offender victim fees for those persons shall be governed by the provisions of section 60.30 of this article.

- 9. Notwithstanding the provisions of subdivision one of this section, in the event a proceeding is in a town or village court, such court shall add an additional five dollars to the surcharges imposed by such subdivision one.
- 10. The provisions of this section shall apply to sentences imposed upon a youthful offender finding; provided, however that the court shall not impose the sex offender registration fee, DNA databank fee or supplemental sex offender victim fee, as defined in subparagraphs (iv) and (v) of paragraph (a) and paragraph (b) of subdivision one of this section, for an offense in which the conviction was substituted with a youthful offender finding.

§ 60.36 Authorized dispositions; driving while intoxicated offenses

Where a court is imposing a sentence for a violation of subdivision two, two-a, or three of section eleven hundred ninety-two of the vehicle and traffic law pursuant to sections 65.00 or 65.05 of this title and, as a condition of such sentence, orders the installation and maintenance of an ignition interlock device, the court may impose any other penalty authorized pursuant to section eleven hundred ninety-three of the vehicle and traffic law.

§ 60.37. Authorized disposition; certain offenses

When a person has been charged with an offense and the elements of such offense meet the criteria of an "eligible offense" and such person qualifies as an "eligible person" as such terms are defined in section four hundred fifty-eight-I of the social services law, the court may, as a condition of adjournment in contemplation of dismissal in accordance with section 170.55 of the criminal procedure law, or a condition of probation or a conditional discharge, direct that the defendant participate in an education reform program pursuant to subdivision two of section four hundred fifty-eight-I of the social services law.

Article 65. Sentences of Probation, Conditional Discharge and Unconditional Discharge

§ 65.00 Sentence of probation

- 1. Criteria. (a) Except as otherwise required by section 60.04 or 60.05 of this title, and except as provided by paragraph (b) hereof, the court may sentence a person to a period of probation upon conviction of any crime if the court, having regard to the nature and circumstances of the crime and to the history, character and condition of the defendant, is of the opinion that:
- (i) Institutional confinement for the term authorized by law of the defendant is or may not be necessary for the protection of the public;
- (ii) the defendant is in need of guidance, training or other assistance which, in his case, can be effectively administered through probation supervision; and
- (iii) such disposition is not inconsistent with the ends of justice.
- (b) The court, with the concurrence of either the administrative judge of the court or of the judicial district within which the court is situated or such administrative judge as the presiding justice of the appropriate appellate division shall designate, may sentence a person to a period of probation upon conviction of a class A-II felony defined in article two hundred twenty, the class B felony defined in section 220.48 of this chapter or any other class B felony defined in article two hundred twenty of this chapter where the person is a second felony drug offender as defined in paragraph (b) of subdivision one of section 70.70 of this chapter, if the prosecutor either orally on the record or in a writing filed with the indictment recommends that the court sentence such person to a period of probation upon the ground that such person has or is providing material assistance in the investigation, apprehension or prosecution of any person for a felony defined in article two hundred twenty or the attempt or the conspiracy to commit any such felony, and if the court, having regard to the nature and circumstances of the crime and to the history, character and condition of the defendant is of the opinion that:
- (i) Institutional confinement of the defendant is not necessary for the protection of the public;
- (ii) The defendant is in need of guidance, training or other assistance which, in his case, can be effectively administered through probation supervision;
 - (iii) The defendant has or is providing material

assistance in the investigation, apprehension or prosecution of a person for a felony defined in article two hundred twenty or the attempt or conspiracy to commit any such felony; and

(iv) Such disposition is not inconsistent with the ends of justice.

Provided, however, that the court shall not, except to the extent authorized by paragraph (d) of subdivision two of section 60.01 of this chapter, impose a sentence of probation in any case where it sentences a defendant for more than one crime and imposes a sentence of imprisonment for any one of the crimes, or where the defendant is subject to an undischarged indeterminate or determinate sentence of imprisonment which was imposed at a previous time by a court of this state and has more than one year to run.

- 2. Sentence. When a person is sentenced to a period of probation the court shall, except to the extent authorized by paragraph (d) of subdivision two of section 60.01 of this chapter, impose the period authorized by subdivision three of this section and shall specify, in accordance with section 65.10, the conditions to be complied with. The court may modify or enlarge the conditions or, if the defendant commits an additional offense or violates a condition, revoke the sentence at any time prior to the expiration or termination of the period of probation.
- 3. Periods of probation. Unless terminated sooner in accordance with the criminal procedure law, the period of probation shall be as follows:
- (a)(i) For a felony, other than a class A-II felony defined in article two hundred twenty of this chapter or the class B felony defined in section 220.48 of this chapter, or any other class B felony defined in article two hundred twenty of this chapter committed by a second felony drug offender, or a sexual assault, the period of probation shall be five years;
- (ii) For a class A-II felony drug offender as defined in paragraph (a) of subdivision one of section 70.71 of this chapter as described in paragraph (b) of subdivision one of this section, or a class B felony committed by a second felony drug offender described in paragraph (b) of subdivision one of this section, the period of probation shall be life and for a class B felony defined in section 220.48 of this chapter, the period of probation shall be twenty-five years;

- (iii) For a felony sexual assault, the period of probation shall be ten years.
- (b) (i) For a class A misdemeanor, other than a sexual assault, the period of probation shall be three years;
- (ii) For a class A misdemeanor sexual assault, the period of probation shall be six years.
- (c) For a class B misdemeanor, the period of probation shall be one year, except the period of probation shall be no less than one year and no more than three years for the class B misdemeanor of public lewdness as defined in section 245.00 of this chapter;
- (d) For an unclassified misdemeanor, the period of probation shall be three years if the authorized sentence of imprisonment is in excess of three months, otherwise the period of probation shall be one year.

For the purposes of this section, the term "sexual assault" means an offense defined in article one hundred thirty or two hundred sixty-three, or in section 255.25, 255.26 or 255.27 of this chapter, or an attempt to commit any of the foregoing offenses.

4. In any case where a court pursuant to its authority under subdivision four of section 60.01 of this chapter revokes probation and sentences such person to imprisonment and probation, as provided in paragraph (d) of subdivision two of section 60.01 of this chapter, the period of probation shall be the remaining period of the original probation sentence or one year whichever is greater.

§ 65.05 Sentence of conditional discharge

- 1. Criteria. (a) Except as otherwise required by section 60.05, the court may impose a sentence of conditional discharge for an offense if the court, having regard to the nature and circumstances of the offense and to the history, character and condition of the defendant, is of the opinion that neither the public interest nor the ends of justice would be served by a sentence of imprisonment and that probation supervision is not appropriate.
- (b) When a sentence of conditional discharge is imposed for a felony, the court shall set forth in the record the reasons for its action.
- 2. Sentence. Except to the extent authorized by paragraph (d) of subdivision two of section 60.01 of this chapter, when the court imposes a sentence of conditional discharge the defendant shall be

released with respect to the conviction for which the sentence is imposed without imprisonment or probation supervision but subject, during the period of conditional discharge, to such conditions as the court may determine. The court shall impose the period of conditional discharge authorized by subdivision three of this section and shall specify, in accordance with section 65.10, the conditions to be complied with. If a defendant is sentenced pursuant to paragraph (e) of subdivision two of section 65.10 of this chapter, the court shall require the administrator of the program to provide written notice to the court of any violation of program participation by the defendant. The court may modify or enlarge the conditions or, if the defendant commits an additional offense or violates a condition, revoke the sentence at any time prior to the expiration or termination of the period of conditional discharge.

- 3. Periods of conditional discharge. Unless terminated sooner in accordance with the criminal procedure law, the period of conditional discharge shall be as follows:
 - (a) Three years in the case of a felony; and
- (b) One year in the case of a misdemeanor or a violation.

Where the court has required, as a condition of the sentence, that the defendant make restitution of the fruits of his or her offense or make reparation for the loss caused thereby and such condition has not been satisfied, the court, at any time prior to the expiration or termination of the period of conditional discharge, may impose an additional period. The length of the additional period shall be fixed by the court at the time it is imposed and shall not be more than two years. All of the incidents of the original sentence, including the authority of the court to modify or enlarge the conditions, shall continue to apply during such additional period.

Article 65. Sentences of Probation, Conditional Discharge and Unconditional Discharge

§ 65.10 Conditions of probation and of conditional discharge

1. In general. The conditions of probation and of conditional discharge shall be such as the court, in its discretion, deems reasonably necessary to insure that the defendant will lead a law-abiding life or to assist him to do so.

- 2. Conditions relating to conduct and rehabilitation. When imposing a sentence of probation or of conditional discharge, the court shall, as a condition of the sentence, consider restitution or reparation and may, as a condition of the sentence, require that the defendant:
 - (a) Avoid injurious or vicious habits;
- (b) Refrain from frequenting unlawful or disreputable places or consorting with disreputable persons;
- (c) Work faithfully at a suitable employment or faithfully pursue a course of study or of vocational training that will equip him for suitable employment;
- (d) Undergo available medical or psychiatric treatment and remain in a specified institution, when required for that purpose;
- (e) Participate in an alcohol or substance abuse program or an intervention program approved by the court after consultation with the local probation department having jurisdiction, or such other public or private agency as the court determines to be appropriate;
- (e-1) Participate in a motor vehicle accident prevention course. The court may require such condition where a person has been convicted of a traffic infraction for a violation of article twenty-six of the vehicle and traffic law where the commission of such violation caused the serious physical injury or death of another person. For purposes of this paragraph, the term "motor vehicle accident prevention course" shall mean a motor vehicle accident prevention course approved by the department of motor vehicles pursuant to article twelve-B of the vehicle and traffic law;
- (f) Support his dependents and meet other family responsibilities;
- (g) Make restitution of the fruits of his or her offense or make reparation, in an amount he can afford to pay, for the actual out-of-pocket loss caused thereby. When restitution or reparation is a condition of the sentence, the court shall fix the amount thereof, the manner of performance, specifically state the date when restitution is to be paid in full prior to the expiration of the sentence of probation and may establish provisions for the early termination of a sentence of probation or conditional discharge pursuant to the provisions of subdivision three of section 410.90 of the criminal procedure law after the restitution and reparation part of a sentence of probation or conditional

- discharge has been satisfied. The court shall provide that in the event the person to whom restitution or reparation is to be made dies prior to the completion of said restitution or reparation, the remaining payments shall be made to the estate of the deceased. [FN1]
- (g-1) Reimburse a consumer credit reporting agency for the amount of the fee or fees that could have been charged by such agency to a domestic violence victim, as defined in section three hundred eighty-t of the general business law, had such victim not been eligible to receive security freeze services without charge pursuant to subdivision (n) of such section;
- (h) Perform services for a public or not-for-profit corporation, association, institution or agency, including but not limited to services for the division of substance abuse services, services in an appropriate community program for removal of graffiti from public or private property, including any property damaged in the underlying offense, or services for the maintenance and repair of real or personal property maintained as a cemetery plot, grave, burial place or other place of interment of human remains. Provided however, that the performance of any such services shall not result in the displacement of employed workers or in the impairment of existing contracts for services, nor shall the performance of any such services be required or permitted in any establishment involved in any labor strike or lockout. The court may establish provisions for the early termination of a sentence of probation or conditional discharge pursuant to the provisions of subdivision three of section 410.90 of the criminal procedure law after such services have been completed. Such sentence may only be imposed upon conviction of a misdemeanor, violation, or class D or class E felony, or a youthful offender finding replacing any such conviction, where the defendant has consented to the amount and conditions of such service;
- (i) If a person under the age of twenty-one years, (i) resides with his parents or in a suitable foster home or hostel as referred to in section two hundred forty-four of the executive law, (ii) attends school, (iii) spends such part of the period of the sentence as the court may direct, but not exceeding two years, in a facility made available by the division for youth pursuant to article nineteen-G of the executive law, provided that admission to such facility may be made only with the prior

consent of the division for youth, (iv) attend a non-residential program for such hours and pursuant to a schedule prescribed by the court as suitable for a program of rehabilitation of youth, (v) contribute to his own support in any home, foster home or hostel;

- (j) Post a bond or other security for the performance of any or all conditions imposed;
- (k) Observe certain specified conditions of conduct as set forth in an order of protection issued pursuant to section 530.12 or 530.13 of the criminal procedure law.
- (k-1) Install and maintain a functioning ignition interlock device, as that term is defined in section one hundred nineteen-a of the vehicle and traffic law, in any vehicle owned or operated by the defendant if the court in its discretion determines that such a condition is necessary to ensure the public safety. The court may require such condition only where a person has been convicted of a violation of subdivision two, two-a or three of section eleven hundred ninety-two of the vehicle and traffic law, or any crime defined by the vehicle and traffic law or this chapter of which an alcohol-related violation of any provision of section eleven hundred ninety-two of the vehicle and traffic law is an essential element. The offender shall be required to install and operate the ignition interlock device only in accordance with section eleven hundred ninety-eight of the vehicle and traffic law.
- (I) Satisfy any other conditions reasonably related to his rehabilitation.
- 3. Conditions relating to supervision. When imposing a sentence of probation the court, in addition to any conditions imposed pursuant to subdivision two of this section, shall require as conditions of the sentence, that the defendant:
- (a) Report to a probation officer as directed by the court or the probation officer and permit the probation officer to visit him at his place of abode or elsewhere:
- (b) Remain within the jurisdiction of the court unless granted permission to leave by the court or the probation officer. Where a defendant is granted permission to move or travel outside the jurisdiction of the court, the defendant shall sign a written waiver of extradition agreeing to waive extradition proceedings where such proceedings are the result of the issuance of a warrant by the court pursuant to subdivision two of section 410.40 of the criminal

- procedure law based on an alleged violation of probation. Where any county or the city of New York incurs costs associated with the return of any probationer based on the issuance of a warrant by the court pursuant to subdivision two of section 410.40 of the criminal procedure law, the jurisdiction may collect the reasonable and necessary expenses involved in connection with his or her transport, from the probationer; provided that where the sentence of probation is not revoked pursuant to section 410.70 of the criminal procedure law no such expenses may be collected.
- (c) Answer all reasonable inquiries by the probation officer and notify the probation officer prior to any change in address or employment.
- 4. Electronic monitoring. When imposing a sentence of probation the court may, in addition to any conditions imposed pursuant to subdivisions two and three of this section, require the defendant to submit to the use of an electronic monitoring device and/or to follow a schedule that governs the defendant's daily movement. Such condition may be imposed only where the court, in its discretion, determines that requiring the defendant to comply with such condition will advance public safety, probationer control or probationer surveillance. Electronic monitoring shall be used in accordance with uniform procedures developed by the office of probation and correctional alternatives.
- 4-a. Mandatory conditions for sex offenders. (a) When imposing a sentence of probation or conditional discharge upon a person convicted of an offense defined in article one hundred thirty, two hundred thirty-five or two hundred sixty-three of this chapter, or section 255.25, 255.26 or 255.27 of this chapter, and the victim of such offense was under the age of eighteen at the time of such offense or such person has been designated a level three sex offender pursuant to subdivision six of section 168-I of the correction law, the court shall require, as a mandatory condition of such sentence, that such sentenced offender shall refrain from knowingly entering into or upon any school grounds, as that term is defined in subdivision fourteen of section 220.00 of this chapter, or any other facility or institution primarily used for the care or treatment of persons under the age of eighteen while one or more of such persons under the age of eighteen are present, provided however, that when such sentenced offender is a registered student or participant or an employee of such facility or institution or entity contracting therewith or has a

family member enrolled in such facility or institution, such sentenced offender may, with the written authorization of his or her probation officer or the court and the superintendent or chief administrator of such facility, institution or grounds, enter such facility, institution or upon such grounds for the limited purposes authorized by the probation officer or the court and superintendent or chief officer. Nothing in this subdivision shall be construed as restricting any lawful condition of supervision that may be imposed on such sentenced offender.

(b) When imposing a sentence of probation or conditional discharge upon a person convicted of an offense for which registration as a sex offender is required pursuant to subdivision two or three of section one hundred sixty-eight-a of the correction law, and the victim of such offense was under the age of eighteen at the time of such offense or such person has been designated a level three sex offender pursuant to subdivision six of section one hundred sixty-eight-I of the correction law or the internet was used to facilitate the commission of the crime, the court shall require, as mandatory conditions of such sentence, that such sentenced offender be prohibited from using the internet to access pornographic material, access a commercial social networking website, communicate with other individuals or groups for the purpose of promoting sexual relations with persons under the age of eighteen, and communicate with a person under the age of eighteen when such offender is over the age of eighteen, provided that the court may permit an offender to use the internet to communicate with a person under the age of eighteen when such offender is the parent of a minor child and is not otherwise prohibited from communicating with such child. Nothing in this subdivision shall be construed as restricting any other lawful condition of supervision that may be imposed on such sentenced offender. As used in this subdivision, a "commercial social networking website" shall mean any business, organization or other entity operating a website that permits persons under eighteen years of age to be registered users for the purpose of establishing personal relationships with other users, where such persons under eighteen years of age may: (i) create web pages or profiles that provide information about themselves where such web pages or profiles are available to the public or to other users; (ii) engage in direct or real time communication with other users, such as a chat

room or instant messenger; and (iii) communicate with persons over eighteen years of age; provided, however, that, for purposes of this subdivision, a commercial social networking website shall not include a website that permits users to engage in such other activities as are not enumerated herein.

- 5. Other conditions. When imposing a sentence of probation the court may, in addition to any conditions imposed pursuant to subdivisions two, three and four of this section, require that the defendant comply with any other reasonable condition as the court shall determine to be necessary or appropriate to ameliorate the conduct which gave rise to the offense or to prevent the incarceration of the defendant.
- 5-a. Other conditions for sex offenders. When imposing a sentence of probation upon a person convicted of an offense for which registration as a sex offender is required pursuant to subdivision two or three of section one hundred sixty-eight-a of the correction law, in addition to any conditions required under subdivisions two, three, four, four-a and five of this section, the court may require that the defendant comply with a reasonable limitation on his or her use of the internet that the court determines to be necessary or appropriate to ameliorate the conduct which gave rise to the offense or to protect public safety, provided that the court shall not prohibit such sentenced offender from using the internet in connection with education, lawful employment or search for lawful employment.

[FN1] So in original. Probably should be a semicolon....

§ 65.20 Sentence of unconditional discharge

1. Criteria. The court may impose a sentence of unconditional discharge in any case where it is authorized to impose a sentence of conditional discharge under section 65.05 if the court is of the opinion that no proper purpose would be served by imposing any condition upon the defendant's release.

When a sentence of unconditional discharge is imposed for a felony, the court shall set forth in the record the reasons for its action.

2. Sentence. When the court imposes a sentence of unconditional discharge, the defendant shall be released with respect to the conviction for which the sentence is imposed without imprisonment, fine or probation supervision. A sentence of

unconditional discharge is for all purposes a final judgment of conviction.

Article 70. Sentences of Imprisonment

§ 70.00 Sentence of imprisonment for felony

- 1. [Eff. until Sept. 1, 2013, pursuant to L.1995, c. 3, § 74, subd. d. See, also, subd. 1 below.] Indeterminate sentence. Except as provided in subdivisions four, five and six of this section or section 70.80 of this article, a sentence of imprisonment for a felony, other than a felony defined in article two hundred twenty or two hundred twenty-one of this chapter, shall be an indeterminate sentence. When such a sentence is imposed, the court shall impose a maximum term in accordance with the provisions of subdivision two of this section and the minimum period of imprisonment shall be as provided in subdivision three of this section.
- 1. [Eff. Sept. 1, 2013. See, also, subd. 1, above.] Indeterminate sentence. Except as provided in subdivisions four and five of this section or section 70.80 of this article, a sentence of imprisonment for a felony, other than a felony defined in article two hundred twenty or two hundred twenty-one of this chapter, shall be an indeterminate sentence. When such a sentence is imposed, the court shall impose a maximum term in accordance with the provisions of subdivision two of this section and the minimum period of imprisonment shall be as provided in subdivision three of this section.
- 2. Maximum term of sentence. The maximum term of an indeterminate sentence shall be at least three years and the term shall be fixed as follows:
- (a) For a class A felony, the term shall be life imprisonment;
- (b) For a class B felony, the term shall be fixed by the court, and shall not exceed twenty-five years;
- (c) For a class C felony, the term shall be fixed by the court, and shall not exceed fifteen years;
- (d) For a class D felony, the term shall be fixed by the court, and shall not exceed seven years; and
- (e) For a class E felony, the term shall be fixed by the court, and shall not exceed four years.
- 3. Minimum period of imprisonment. The minimum period of imprisonment under an indeterminate sentence shall be at least one year and shall be fixed as follows:
 - (a) In the case of a class A felony, the minimum

period shall be fixed by the court and specified in the sentence.

- (i) For a class A-I felony, such minimum period shall not be less than fifteen years nor more than twenty-five years; provided, however, that (A) where a sentence, other than a sentence of death or life imprisonment without parole, is imposed upon a defendant convicted of murder in the first degree as defined in section 125.27 of this chapter such minimum period shall be not less than twenty years nor more than twenty-five years, and, (B) where a sentence is imposed upon a defendant convicted of murder in the second degree as defined in subdivision five of section 125.25 of this chapter or convicted of aggravated murder as defined in section 125.26 of this chapter, the sentence shall be life imprisonment without parole, and, (C) where a sentence is imposed upon a defendant convicted of attempted murder in the first degree as defined in article one hundred ten of this chapter and subparagraph (i), (ii) or (iii) of paragraph (a) of subdivision one and paragraph (b) of subdivision one of section 125.27 of this chapter or attempted aggravated murder as defined in article one hundred ten of this chapter and section 125.26 of this chapter such minimum period shall be not less than twenty years nor more than forty years.
- (ii) For a class A-II felony, such minimum period shall not be less than three years nor more than eight years four months, except that for the class A-II felony of predatory sexual assault as defined in section 130.95 of this chapter or the class A-II felony of predatory sexual assault against a child as defined in section 130.96 of this chapter, such minimum period shall be not less than ten years nor more than twenty-five years.
- (b) For any other felony, the minimum period shall be fixed by the court and specified in the sentence and shall be not less than one year nor more than one-third of the maximum term imposed.
- 4. Alternative definite sentence for class D and E felonies. When a person, other than a second or persistent felony offender, is sentenced for a class D or class E felony, and the court, having regard to the nature and circumstances of the crime and to the history and character of the defendant, is of the opinion that a sentence of imprisonment is necessary but that it would be unduly harsh to impose an indeterminate or determinate sentence, the court may impose a definite sentence of imprisonment and fix a term of one year or less.

- 5. Life imprisonment without parole. Notwithstanding any other provision of law, a defendant sentenced to life imprisonment without parole shall not be or become eligible for parole or conditional release. For purposes of commitment and custody, other than parole and conditional release, such sentence shall be deemed to be an indeterminate sentence. A defendant may be sentenced to life imprisonment without parole upon conviction for the crime of murder in the first degree as defined in section 125.27 of this chapter and in accordance with the procedures provided by law for imposing a sentence for such crime. A defendant must be sentenced to life imprisonment without parole upon conviction for the crime of terrorism as defined in section 490.25 of this chapter, where the specified offense the defendant committed is a class A-I felony; the crime of criminal possession of a chemical weapon or biological weapon in the first degree as defined in section 490.45 of this chapter; or the crime of criminal use of a chemical weapon or biological weapon in the first degree as defined in section 490.55 of this chapter; provided, however, that nothing in this subdivision shall preclude or prevent a sentence of death when the defendant is also convicted of the crime of murder in the first degree as defined in section 125.27 of this chapter. A defendant must be sentenced to life imprisonment without parole upon conviction for the crime of murder in the second degree as defined in subdivision five of section 125.25 of this chapter or for the crime of aggravated murder as defined in subdivision one of section 125.26 of this chapter. A defendant may be sentenced to life imprisonment without parole upon conviction for the crime of aggravated murder as defined in subdivision two of section 125.26 of this chapter.
- 6. [Deemed repealed Sept. 1, 2013, pursuant to L.1995, c. 3, § 74, subd. d.] Determinate sentence. Except as provided in subdivision four of this section and subdivisions two and four of section 70.02, when a person is sentenced as a violent felony offender pursuant to section 70.02 or as a second violent felony offender pursuant to section 70.04 or as a second felony offender on a conviction for a violent felony offense pursuant to section 70.06, the court must impose a determinate sentence of imprisonment in accordance with the provisions of such sections and such sentence shall include, as a part thereof, a period of post-release supervision in accordance with section 70.45.

§ 70.02 Sentence of imprisonment for a violent felony offense

- 1. Definition of a violent felony offense. A violent felony offense is a class B violent felony offense, a class C violent felony offense, a class D violent felony offense, or a class E violent felony offense, defined as follows:
- (a) Class B violent felony offenses: an attempt to commit the class A-I felonies of murder in the second degree as defined in section 125.25. kidnapping in the first degree as defined in section 135.25, and arson in the first degree as defined in section 150.20; manslaughter in the first degree as defined in section 125.20, aggravated manslaughter in the first degree as defined in section 125.22, rape in the first degree as defined in section 130.35, criminal sexual act in the first degree as defined in section 130.50, aggravated sexual abuse in the first degree as defined in section 130.70, course of sexual conduct against a child in the first degree as defined in section 130.75; assault in the first degree as defined in section 120.10, kidnapping in the second degree as defined in section 135.20, burglary in the first degree as defined in section 140.30, arson in the second degree as defined in section 150.15, robbery in the first degree as defined in section 160.15, incest in the first degree as defined in section 255.27, criminal possession of a weapon in the first degree as defined in section 265.04, criminal use of a firearm in the first degree as defined in section 265.09, criminal sale of a firearm in the first degree as defined in section 265.13, aggravated assault upon a police officer or a peace officer as defined in section 120.11, gang assault in the first degree as defined in section 120.07, intimidating a victim or witness in the first degree as defined in section 215.17, hindering prosecution of terrorism in the first degree as defined in section 490.35, criminal possession of a chemical weapon or biological weapon in the second degree as defined in section 490.40, and criminal use of a chemical weapon or biological weapon in the third degree as defined in section 490.47.
- (b) Class C violent felony offenses: an attempt to commit any of the class B felonies set forth in paragraph (a) of this subdivision; aggravated criminally negligent homicide as defined in section 125.11, aggravated manslaughter in the second degree as defined in section 125.21, aggravated sexual abuse in the second degree as defined in section 130.67, assault on a peace officer, police

officer, fireman or emergency medical services professional as defined in section 120.08, assault on a judge as defined in section 120.09, gang assault in the second degree as defined in section 120.06, strangulation in the first degree as defined in section 121.13, burglary in the second degree as defined in section 140.25, robbery in the second degree as defined in section 160.10, criminal possession of a weapon in the second degree as defined in section 265.03, criminal use of a firearm in the second degree as defined in section 265.08, criminal sale of a firearm in the second degree as defined in section 265.12, criminal sale of a firearm with the aid of a minor as defined in section 265.14, soliciting or providing support for an act of terrorism in the first degree as defined in section 490.15, hindering prosecution of terrorism in the second degree as defined in section 490.30, and criminal possession of a chemical weapon or biological weapon in the third degree as defined in section 490.37.

(c) Class D violent felony offenses: an attempt to commit any of the class C felonies set forth in paragraph (b); reckless assault of a child as defined in section 120.02, assault in the second degree as defined in section 120.05, menacing a police officer or peace officer as defined in section 120.18, stalking in the first degree, as defined in subdivision one of section 120.60, strangulation in the second degree as defined in section 121.12, rape in the second degree as defined in section 130.30, criminal sexual act in the second degree as defined in section 130.45, sexual abuse in the first degree as defined in section 130.65, course of sexual conduct against a child in the second degree as defined in section 130.80, aggravated sexual abuse in the third degree as defined in section 130.66, facilitating a sex offense with a controlled substance as defined in section 130.90, criminal possession of a weapon in the third degree as defined in subdivision five, six, seven or eight of section 265.02, criminal sale of a firearm in the third degree as defined in section 265.11, intimidating a victim or witness in the second degree as defined in section 215.16, soliciting or providing support for an act of terrorism in the second degree as defined in section 490.10, and making a terroristic threat as defined in section 490.20, falsely reporting an incident in the first degree as defined in section 240.60, placing a false bomb or hazardous substance in the first degree as defined in section 240.62, placing a false bomb or hazardous substance in a sports stadium or

arena, mass transportation facility or enclosed shopping mall as defined in section 240.63, and aggravated unpermitted use of indoor pyrotechnics in the first degree as defined in section 405.18.

(d) Class E violent felony offenses: an attempt to commit any of the felonies of criminal possession of a weapon in the third degree as defined in subdivision five, six, seven or eight of section 265.02 as a lesser included offense of that section as defined in section 220.20 of the criminal procedure law, persistent sexual abuse as defined in section 130.53, aggravated sexual abuse in the fourth degree as defined in section 130.65-a, falsely reporting an incident in the second degree as defined in section 240.55 and placing a false bomb or hazardous substance in the second degree as defined in section 240.61.

2. Authorized sentence.

- (a) Except as provided in subdivision six of section 60.05, the sentence imposed upon a person who stands convicted of a class B or class C violent felony offense must be a determinate sentence of imprisonment which shall be in whole or half years. The term of such sentence must be in accordance with the provisions of subdivision three of this section.
- (b) Except as provided in paragraph (b-1) of this subdivision, subdivision six of section 60.05 and subdivision four of this section, the sentence imposed upon a person who stands convicted of a class D violent felony offense, other than the offense of criminal possession of a weapon in the third degree as defined in subdivision five, seven or eight of section 265.02 or criminal sale of a firearm in the third degree as defined in section 265.11, must be in accordance with the applicable provisions of this chapter relating to sentencing for class D felonies provided, however, that where a sentence of imprisonment is imposed which requires a commitment to the state department of corrections and community supervision, such sentence shall be a determinate sentence in accordance with paragraph (c) of subdivision three of this section.
- (b-1) Except as provided in subdivision six of section 60.05, the sentence imposed upon a person who stands convicted of the class D violent felony offense of menacing a police officer or peace officer as defined in section 120.18 of this chapter must be a determinate sentence of imprisonment.

- (c) Except as provided in subdivision six of section 60.05, the sentence imposed upon a person who stands convicted of the class D violent felony offenses of criminal possession of a weapon in the third degree as defined in subdivision four, five, seven or eight of section 265.02, criminal sale of a firearm in the third degree as defined in section 265.11 or the class E violent felonies of attempted criminal possession of a weapon in the third degree as defined in subdivision four, five, seven or eight of section 265.02 must be a sentence to a determinate period of imprisonment, or, in the alternative, a definite sentence of imprisonment for a period of no less than one year, except that:
- (i) the court may impose any other sentence authorized by law upon a person who has not been previously convicted in the five years immediately preceding the commission of the offense for a class A misdemeanor defined in this chapter, if the court having regard to the nature and circumstances of the crime and to the history and character of the defendant, finds on the record that such sentence would be unduly harsh and that the alternative sentence would be consistent with public safety and does not deprecate the seriousness of the crime; and
- (ii) the court may apply the provisions of paragraphs (b) and (c) of subdivision four of this section when imposing a sentence upon a person who has previously been convicted of a class A misdemeanor defined in this chapter in the five years immediately preceding the commission of the offense.
- 3. Term of sentence. The term of a determinate sentence for a violent felony offense must be fixed by the court as follows:
- (a) For a class B felony, the term must be at least five years and must not exceed twenty-five years, provided, however, that the term must be: (i) at least ten years and must not exceed thirty years where the sentence is for the crime of aggravated assault upon a police officer or peace officer as defined in section 120.11 of this chapter; and (ii) at least ten years and must not exceed thirty years where the sentence is for the crime of aggravated manslaughter in the first degree as defined in section 125.22 of this chapter;
- (b) For a class C felony, the term must be at least three and one-half years and must not exceed fifteen years, provided, however, that the term must

- be: (i) at least seven years and must not exceed twenty years where the sentence is for the crime of aggravated manslaughter in the second degree as defined in section 125.21 of this chapter; (ii) at least seven years and must not exceed twenty years where the sentence is for the crime of attempted aggravated assault upon a police officer or peace officer as defined in section 120.11 of this chapter; and (iii) at least three and one-half years and must not exceed twenty years where the sentence is for the crime of aggravated criminally negligent homicide as defined in section 125.11 of this chapter;
- (c) For a class D felony, the term must be at least two years and must not exceed seven years, provided, however, that the term must be at least two years and must not exceed eight years where the sentence is for the crime of menacing a police officer or peace officer as defined in section 120.18 of this chapter; and
- (d) For a class E felony, the term must be at least one and one-half years and must not exceed four years.
- 4. (a) Except as provided in paragraph (b) of this subdivision, where a plea of guilty to a class D violent felony offense is entered pursuant to section 220.10 or 220.30 of the criminal procedure law in satisfaction of an indictment charging the defendant with an armed felony, as defined in subdivision forty-one of section 1.20 of the criminal procedure law, the court must impose a determinate sentence of imprisonment.
- (b) In any case in which the provisions of paragraph (a) of this subdivision or the provisions of subparagraph (ii) of paragraph (c) of subdivision two of this section apply, the court may impose a sentence other than a determinate sentence of imprisonment, or a definite sentence of imprisonment for a period of no less than one year, if it finds that the alternate sentence is consistent with public safety and does not deprecate the seriousness of the crime and that one or more of the following factors exist:
- (i) mitigating circumstances that bear directly upon the manner in which the crime was committed; or
- (ii) where the defendant was not the sole participant in the crime, the defendant's participation was relatively minor although not so minor as to constitute a defense to the prosecution; or

- (iii) possible deficiencies in proof of the defendant's commission of an armed felony.
- (c) The defendant and the district attorney shall have an opportunity to present relevant information to assist the court in making a determination pursuant to paragraph (b) of this subdivision, and the court may, in its discretion, conduct a hearing with respect to any issue bearing upon such determination. If the court determines that a determinate sentence of imprisonment should not be imposed pursuant to the provisions of such paragraph (b), it shall make a statement on the record of the facts and circumstances upon which such determination is based. A transcript of the court's statement, which shall set forth the recommendation of the district attorney, shall be forwarded to the state division of criminal justice services along with a copy of the accusatory instrument.
 - 5. Renumbered.

[FN1] Now subd. 6.

§ 70.04 Sentence of imprisonment for second violent felony offender

- 1. Definition of second violent felony offender.
- (a) A second violent felony offender is a person who stands convicted of a violent felony offense as defined in subdivision one of section 70.02 after having previously been subjected to a predicate violent felony conviction as defined in paragraph (b) of this subdivision.
- (b) For the purpose of determining whether a prior conviction is a predicate violent felony conviction the following criteria shall apply:
- (i) The conviction must have been in this state of a class A felony (other than one defined in article two hundred twenty) or of a violent felony offense as defined in subdivision one of section 70.02, or of an offense defined by the penal law in effect prior to September first, nineteen hundred sixty-seven, which includes all of the essential elements of any such felony, or in any other jurisdiction of an offense which includes all of the essential elements of any such felony for which a sentence to a term of imprisonment in excess of one year or a sentence of death was authorized and is authorized in this state irrespective of whether such sentence was imposed:
- (ii) Sentence upon such prior conviction must have been imposed before commission of the

present felony;

- (iii) Suspended sentence, suspended execution of sentence, a sentence of probation, a sentence of conditional discharge or of unconditional discharge, and a sentence of certification to the care and custody of the division of substance abuse services, shall be deemed to be a sentence;
- (iv) Except as provided in subparagraph (v) of this paragraph, sentence must have been imposed not more than ten years before commission of the felony of which the defendant presently stands convicted:
- (v) In calculating the ten year period under subparagraph (iv), any period of time during which the person was incarcerated for any reason between the time of commission of the previous felony and the time of commission of the present felony shall be excluded and such ten year period shall be extended by a period or periods equal to the time served under such incarceration;
- (vi) An offense for which the defendant has been pardoned on the ground of innocence shall not be deemed a predicate violent felony conviction.
- 2. Authorized sentence. When the court has found, pursuant to the provisions of the criminal procedure law, that a person is a second violent felony offender the court must impose a determinate sentence of imprisonment which shall be in whole or half years. Except where sentence is imposed in accordance with the provisions of section 70.10, the term of such sentence must be in accordance with the provisions of subdivision three of this section.
- 3. Term of sentence. The term of a determinate sentence for a second violent felony offender must be fixed by the court as follows:
- (a) For a class B felony, the term must be at least ten years and must not exceed twenty-five years;
- (b) For a class C felony, the term must be at least seven years and must not exceed fifteen years; and
- (c) For a class D felony, the term must be at least five years and must not exceed seven years.
- (d) For a class E felony, the term must be at least three years and must not exceed four years.
- 4. Minimum period of imprisonment. The minimum period of imprisonment under an indeterminate sentence for a second violent felony

offender must be fixed by the court at one-half of the maximum term imposed and must be specified in the sentence.

§ 70.05 Sentence of imprisonment for juvenile offender

- 1. Indeterminate sentence. A sentence of imprisonment for a felony committed by a juvenile offender shall be an indeterminate sentence. When such a sentence is imposed, the court shall impose a maximum term in accordance with the provisions of subdivision two of this section and the minimum period of imprisonment shall be as provided in subdivision three of this section. The court shall further provide that where a juvenile offender is under placement pursuant to article three of the family court act, any sentence imposed pursuant to this section which is to be served consecutively with such placement shall be served in a facility designated pursuant to subdivision four of section 70.20 of this article prior to service of the placement in any previously designated facility.
- 2. Maximum term of sentence. The maximum term of an indeterminate sentence for a juvenile offender shall be at least three years and the term shall be fixed as follows:
- (a) For the class A felony of murder in the second degree, the term shall be life imprisonment;
- (b) For the class A felony of arson in the first degree, or for the class A felony of kidnapping in the first degree the term shall be fixed by the court, and shall be at least twelve years but shall not exceed fifteen years;
- (c) For a class B felony, the term shall be fixed by the court, and shall not exceed ten years;
- (d) For a class C felony, the term shall be fixed by the court, and shall not exceed seven years; and
- (e) For a class D felony, the term shall be fixed by the court and shall not exceed four years.
- 3. Minimum period of imprisonment. The minimum period of imprisonment under an indeterminate sentence for a juvenile offender shall be specified in the sentence as follows:
- (a) For the class A felony of murder in the second degree, the minimum period of imprisonment shall be fixed by the court and shall be not less than five years but shall not exceed nine years provided, however, that where the sentence is for an offense specified in subdivision one or two of section 125.25 of this chapter and the defendant was

- fourteen or fifteen years old at the time of such offense, the minimum period of imprisonment shall be not less than seven and one-half years but shall not exceed fifteen years;
- (b) For the class A felony of arson in the first degree, or for the class A felony of kidnapping in the first degree, the minimum period of imprisonment shall be fixed by the court and shall be not less than four years but shall not exceed six years; and
- (c) For a class B, C or D felony, the minimum period of imprisonment shall be fixed by the court at one-third of the maximum term imposed.

§ 70.06 Sentence of imprisonment for second felony offender

- 1. Definition of second felony offender.
- (a) A second felony offender is a person, other than a second violent felony offender as defined in section 70.04, who stands convicted of a felony defined in this chapter, other than a class A-I felony, after having previously been subjected to one or more predicate felony convictions as defined in paragraph (b) of this subdivision.
- (b) For the purpose of determining whether a prior conviction is a predicate felony conviction the following criteria shall apply:
- (i) The conviction must have been in this state of a felony, or in any other jurisdiction of an offense for which a sentence to a term of imprisonment in excess of one year or a sentence of death was authorized and is authorized in this state irrespective of whether such sentence was imposed;
- (ii) Sentence upon such prior conviction must have been imposed before commission of the present felony;
- (iii) Suspended sentence, suspended execution of sentence, a sentence of probation, a sentence of conditional discharge or of unconditional discharge, and a sentence of certification to the care and custody of the division of substance abuse services, shall be deemed to be a sentence;
- (iv) Except as provided in subparagraph (v) of this paragraph, sentence must have been imposed not more than ten years before commission of the felony of which the defendant presently stands convicted;
 - (v) In calculating the ten year period under

subparagraph (iv), any period of time during which the person was incarcerated for any reason between the time of commission of the previous felony and the time of commission of the present felony shall be excluded and such ten year period shall be extended by a period or periods equal to the time served under such incarceration;

- (vi) An offense for which the defendant has been pardoned on the ground of innocence shall not be deemed a predicate felony conviction.
- 2. Authorized sentence. Except as provided in subdivision five or six of this section, or as provided in subdivision five of section 70.80 of this article, when the court has found, pursuant to the provisions of the criminal procedure law, that a person is a second felony offender the court must impose an indeterminate sentence of imprisonment. The maximum term of such sentence must be in accordance with the provisions of subdivision three of this section and the minimum period of imprisonment under such sentence must be in accordance with subdivision four of this section.
- 3. Maximum term of sentence. Except as provided in subdivision five or six of this section, or as provided in subdivision five of section 70.80 of this article, the maximum term of an indeterminate sentence for a second felony offender must be fixed by the court as follows:
- (a) For a class A-II felony, the term must be life imprisonment;
- (b) For a class B felony, the term must be at least nine years and must not exceed twenty-five years;
- (c) For a class C felony, the term must be at least six years and must not exceed fifteen years:
- (d) For a class D felony, the term must be at least four years and must not exceed seven years; and
- (e) For a class E felony, the term must be at least three years and must not exceed four years; provided, however, that where the sentence is for the class E felony offense specified in section 240.32 of this chapter, the maximum term must be at least three years and must not exceed five years.
- 4. Minimum period of imprisonment. (a) The minimum period of imprisonment for a second felony offender convicted of a class A-II felony must be fixed by the court at no less than six years and not to exceed twelve and one-half years and

- must be specified in the sentence, except that for the class A-II felony of predatory sexual assault as defined in section 130.95 of this chapter or the class A-II felony of predatory sexual assault against a child as defined in section 130.96 of this chapter, such minimum period shall be not less than ten years nor more than twenty-five years.
- (b) Except as provided in paragraph (a), the minimum period of imprisonment under an indeterminate sentence for a second felony offender must be fixed by the court at one-half of the maximum term imposed and must be specified in the sentence.
- 5. Repealed by L.2004, c. 738, § 31, eff. Dec. 27, 2004.
- 6. Determinate sentence. When the court has found, pursuant to the provisions of the criminal procedure law, that a person is a second felony offender and the sentence to be imposed on such person is for a violent felony offense, as defined in subdivision one of section 70.02, the court must impose a determinate sentence of imprisonment the term of which must be fixed by the court as follows:
- (a) For a class B violent felony offense, the term must be at least eight years and must not exceed twenty-five years;
- (b) For a class C violent felony offense, the term must be at least five years and must not exceed fifteen years;
- (c) For a class D violent felony offense, the term must be at least three years and must not exceed seven years; and
- (d) For a class E violent felony offense, the term must be at least two years and must not exceed four years.
- 7. Notwithstanding any other provision of law, in the case of a person sentenced for a specified offense or offenses as defined in subdivision five of section 410.91 of the criminal procedure law, who stands convicted of no other felony offense, who has not previously been convicted of either a violent felony offense as defined in section 70.02 of this article, a class A felony offense or a class B felony offense, and is not under the jurisdiction of or awaiting delivery to the department of corrections and community supervision, the court may direct that such sentence be executed as a parole supervision sentence as defined in and pursuant to the procedures prescribed in section

410.91 of the criminal procedure law....

§ 70.08 Sentence of imprisonment for persistent violent felony offender; criteria

- 1. Definition of persistent violent felony offender.
- (a) A persistent violent felony offender is a person who stands convicted of a violent felony offense as defined in subdivision one of section 70.02 or the offense of predatory sexual assault as defined in section 130.95 of this chapter or the offense of predatory sexual assault against a child as defined in section 130.96 of this chapter, after having previously been subjected to two or more predicate violent felony convictions as defined in paragraph (b) of subdivision one of section 70.04 of this article.
- (b) For the purpose of determining whether a person has two or more predicate violent felony convictions, the criteria set forth in paragraph (b) of subdivision one of section 70.04 shall apply.
- 2. Authorized sentence. When the court has found, pursuant to the provisions of the criminal procedure law, that a person is a persistent violent felony offender the court must impose an indeterminate sentence of imprisonment, the maximum term of which shall be life imprisonment. The minimum period of imprisonment under such sentence must be in accordance with subdivision three of this section.
- 3. Minimum period of imprisonment. The minimum period of imprisonment under an indeterminate life sentence for a persistent violent felony offender must be fixed by the court as follows:
- (a) For the class A-II felony of predatory sexual assault as defined in section 130.95 of this chapter or the class A-II felony of predatory sexual assault against a child as defined in section 130.96 of this chapter, the minimum period must be twenty-five years;
- (a-1) For a class B felony, the minimum period must be at least twenty years and must not exceed twenty-five years;
- (b) For a class C felony, the minimum period must be at least sixteen years and must not exceed twenty-five years;
- (c) For a class D felony, the minimum period must be at least twelve years and must not exceed twenty-five years.

§ 70.10 Sentence of imprisonment for persistent felony offender

- 1. Definition of persistent felony offender.
- (a) A persistent felony offender is a person, other than a persistent violent felony offender as defined in section 70.08, who stands convicted of a felony after having previously been convicted of two or more felonies, as provided in paragraphs (b) and (c) of this subdivision.
- (b) A previous felony conviction within the meaning of paragraph (a) of this subdivision is a conviction of a felony in this state, or of a crime in any other jurisdiction, provided:
- (i) that a sentence to a term of imprisonment in excess of one year, or a sentence to death, was imposed therefor; and
- (ii) that the defendant was imprisoned under sentence for such conviction prior to the commission of the present felony; and
- (iii) that the defendant was not pardoned on the ground of innocence; and
- (iv) that such conviction was for a felony offense other than persistent sexual abuse, as defined in section 130.53 of this chapter.
- (c) For the purpose of determining whether a person has two or more previous felony convictions, two or more convictions of crimes that were committed prior to the time the defendant was imprisoned under sentence for any of such convictions shall be deemed to be only one conviction.
- 2. Authorized sentence. When the court has found, pursuant to the provisions of the criminal procedure law, that a person is a persistent felony offender, and when it is of the opinion that the history and character of the defendant and the nature and circumstances of his criminal conduct indicate that extended incarceration and life-time supervision will best serve the public interest, the court, in lieu of imposing the sentence of imprisonment authorized by section 70.00, 70.02, 70.04, 70.06 or subdivision five of section 70.80 for the crime of which such person presently stands convicted, may impose the sentence of imprisonment authorized by that section for a class A-I felony. In such event the reasons for the court's opinion shall be set forth in the record.

§ 70.15 Sentences of imprisonment for misdemeanors and violation

- 1. Class A misdemeanor, A sentence of imprisonment for a class A misdemeanor shall be a definite sentence. When such a sentence is imposed the term shall be fixed by the court, and shall not exceed one year; provided, however, that a sentence of imprisonment imposed upon a conviction of criminal possession of a weapon in the fourth degree as defined in subdivision one of section 265.01 must be for a period of no less than one year when the conviction was the result of a plea of guilty entered in satisfaction of an indictment or any count thereof charging the defendant with the class D violent felony offense of criminal possession of a weapon in the third degree as defined in subdivision four of section 265.02, except that the court may impose any other sentence authorized by law upon a person who has not been previously convicted in the five years immediately preceding the commission of the offense for a felony or a class A misdemeanor defined in this chapter, if the court having regard to the nature and circumstances of the crime and to the history and character of the defendant, finds on the record that such sentence would be unduly harsh and that the alternative sentence would be consistent with public safety and does not deprecate the seriousness of the crime.
- 2. Class B misdemeanor. A sentence of imprisonment for a class B misdemeanor shall be a definite sentence. When such a sentence is imposed the term shall be fixed by the court, and shall not exceed three months.
- 3. Unclassified misdemeanor. A sentence of imprisonment for an unclassified misdemeanor shall be a definite sentence. When such a sentence is imposed the term shall be fixed by the court, and shall be in accordance with the sentence specified in the law or ordinance that defines the crime.
- 4. Violation. A sentence of imprisonment for a violation shall be a definite sentence. When such a sentence is imposed the term shall be fixed by the court, and shall not exceed fifteen days.

In the case of a violation defined outside this chapter, if the sentence is expressly specified in the law or ordinance that defines the offense and consists solely of a fine, no term of imprisonment shall be imposed....

§ 70.25 Concurrent and consecutive terms of imprisonment

1. Except as provided in subdivisions two, two-a

- and five of this section, when multiple sentences of imprisonment are imposed on a person at the same time, or when a person who is subject to any undischarged term of imprisonment imposed at a previous time by a court of this state is sentenced to an additional term of imprisonment, the sentence or sentences imposed by the court shall run either concurrently or consecutively with respect to each other and the undischarged term or terms in such manner as the court directs at the time of sentence. If the court does not specify the manner in which a sentence imposed by it is to run, the sentence shall run as follows:
- (a) [Eff. until Sept. 1, 2013, pursuant to L.1995, c. 3, § 74, subd. d. See, also, par. (a) below.] An indeterminate or determinate sentence shall run concurrently with all other terms; and
- (a) [Eff. Sept. 1, 2013. See, also, par. (a) above.] An indeterminate sentence shall run concurrently with all other terms; and
- (b) A definite sentence shall run concurrently with any sentence imposed at the same time and shall be consecutive to any other term.
- 2. When more than one sentence of imprisonment is imposed on a person for two or more offenses committed through a single act or omission, or through an act or omission which in itself constituted one of the offenses and also was a material element of the other, the sentences, except if one or more of such sentences is for a violation of section 270.20 of this chapter, must run concurrently.
- 2-a. When an indeterminate or determinate sentence of imprisonment is imposed pursuant to section 70.04, 70.06, 70.07, 70.08, 70.10, subdivision three or four of section 70.70, subdivision three or four of section 70.71 or subdivision five of section 70.80 of this article, or is imposed for a class A-I felony pursuant to section 70.00 of this article, and such person is subject to an undischarged indeterminate or determinate sentence of imprisonment imposed prior to the date on which the present crime was committed, the court must impose a sentence to run consecutively with respect to such undischarged sentence.
- 2-b. When a person is convicted of a violent felony offense committed after arraignment and while released on recognizance or bail, but committed prior to the imposition of sentence on a pending felony charge, and if an indeterminate or

determinate sentence of imprisonment is imposed in each case, such sentences shall run consecutively. Provided, however, that the court may, in the interest of justice, order a sentence to run concurrently in a situation where consecutive sentences are required by this subdivision if it finds either mitigating circumstances that bear directly upon the manner in which the crime was committed or, where the defendant was not the sole participant in the crime, the defendant's participation was relatively minor although not so minor as to constitute a defense to the prosecution. The defendant and the district attorney shall have an opportunity to present relevant information to assist the court in making this determination and the court may, in its discretion, conduct a hearing with respect to any issue bearing upon such determination. If the court determines that consecutive sentences should not be ordered, it shall make a statement on the record of the facts and circumstances upon which such determination is based....

- 3. Where consecutive definite sentences of imprisonment are not prohibited by subdivision two of this section and are imposed on a person for offenses which were committed as parts of a single incident or transaction, the aggregate of the terms of such sentences shall not exceed one year.
- 4. When a person, who is subject to any undischarged term of imprisonment imposed at a previous time by a court of another jurisdiction, is sentenced to an additional term or terms of imprisonment by a court of this state, the sentence or sentences imposed by the court of this state, subject to the provisions of subdivisions one, two and three of this section, shall run either concurrently or consecutively with respect to such undischarged term in such manner as the court directs at the time of sentence. If the court of this state does not specify the manner in which a sentence imposed by it is to run, the sentence or sentences shall run consecutively.
- 5. (a) Except as provided in paragraph (c) of this subdivision, when a person is convicted of assault in the second degree, as defined in subdivision seven of section 120.05 of this chapter, any definite, indeterminate or determinate term of imprisonment which may be imposed as a sentence upon such conviction shall run consecutively to any undischarged term of imprisonment to which the defendant was subject and for which he was confined at the time of the

assault.

- (b) Except as provided in paragraph (c) of this subdivision, when a person is convicted of assault in the second degree, as defined in subdivision seven of section 120.05 of this chapter, any definite, indeterminate or determinate term of imprisonment which may be imposed as a sentence upon such conviction shall run consecutively to any term of imprisonment which was previously imposed or which may be prospectively imposed where the person was confined within a detention facility at the time of the assault upon a charge which culminated in such sentence of imprisonment.
- (c) Notwithstanding the provisions of paragraphs (a) and (b) of this subdivision, a term of imprisonment imposed upon a conviction to assault in the second degree as defined in subdivision seven of section 120.05 of this chapter may run concurrently to any other term of imprisonment, in the interest of justice, provided the court sets forth in the record its reasons for imposing a concurrent sentence. Nothing in this section shall require the imposition of a sentence of imprisonment where it is not otherwise required by law.

§ 70.30 Calculation of terms of imprisonment

- 1. An indeterminate or determinate sentence of imprisonment commences when the prisoner is received in an institution under the jurisdiction of the state department of corrections and community supervision. Where a person is under more than one indeterminate or determinate sentence, the sentences shall be calculated as follows:
- (a) If the sentences run concurrently, the time served under imprisonment on any of the sentences shall be credited against the minimum periods of all the concurrent indeterminate sentences and against the terms of all the concurrent determinate sentences. The maximum term or terms of the indeterminate sentences and the term or terms of the determinate sentences shall merge in and be satisfied by discharge of the term which has the longest unexpired time to run;
- (b) If the defendant is serving two or more indeterminate sentences which run consecutively, the minimum periods of imprisonment are added to arrive at an aggregate minimum period of imprisonment equal to the sum of all the minimum periods, and the maximum terms are added to arrive at an aggregate maximum term equal to the sum of all the maximum terms, provided, however,

that both the aggregate maximum term and the aggregate minimum period of imprisonment shall be subject to the limitations set forth in paragraphs (e) and (f) of this subdivision, where applicable;

- (c) If the defendant is serving two or more determinate sentences of imprisonment which run consecutively, the terms of the determinate sentences are added to arrive at an aggregate maximum term of imprisonment, provided, however, that the aggregate maximum term of imprisonment shall be subject to the limitations set forth in paragraphs (e) and (f) of this subdivision, where applicable.
- (d) If the defendant is serving one or more indeterminate sentences of imprisonment and one or more determinate sentence of imprisonment which run consecutively, the minimum term or terms of the indeterminate sentence or sentences and the term or terms of the determinate sentence or sentences are added to arrive at an aggregate maximum term of imprisonment, provided, however, (i) that in no event shall the aggregate maximum so calculated be less than the term or maximum term of imprisonment of the sentence which has the longest unexpired time to run; and (ii) that the aggregate maximum term of imprisonment shall be subject to the limitations set forth in paragraphs (e) and (f) of this subdivision, where applicable.
- (e)(i) Except as provided in subparagraph (ii), (iii), (iv), (v), (vi) or (vii) of this paragraph, the aggregate maximum term of consecutive sentences, all of which are indeterminate sentences or all of which are determinate sentences, imposed for two or more crimes, other than two or more crimes that include a class A felony, committed prior to the time the person was imprisoned under any of such sentences shall, if it exceeds twenty years, be deemed to be twenty years, unless one of the sentences was imposed for a class B felony, in which case the aggregate maximum term shall, if it exceeds thirty years, be deemed to be thirty years. Where the aggregate maximum term of two or more indeterminate consecutive sentences is reduced by calculation made pursuant to this paragraph, the aggregate minimum period of imprisonment, if it exceeds one-half of the aggregate maximum term as so reduced, shall be deemed to be one-half of the aggregate maximum term as so reduced;
- (ii) Where the aggregate maximum term of two or more consecutive sentences, one or more of which

- is a determinate sentence and one or more of which is an indeterminate sentence, imposed for two or more crimes, other than two or more crimes that include a class A felony, committed prior to the time the person was imprisoned under any of such sentences, exceeds twenty years, and none of the sentences was imposed for a class B felony, the following rules shall apply:
- (A) if the aggregate maximum term of the determinate sentence or sentences exceeds twenty years, the defendant shall be deemed to be serving to [FN1] a determinate sentence of twenty years.
- (B) if the aggregate maximum term of the determinate sentence or sentences is less than twenty years, the defendant shall be deemed to be serving an indeterminate sentence the maximum term of which shall be deemed to be twenty years. In such instances, the minimum sentence shall be deemed to be ten years or six-sevenths of the term or aggregate maximum term of the determinate sentence or sentences, whichever is greater.
- (iii) Where the aggregate maximum term of two or more consecutive sentences, one or more of which is a determinate sentence and one or more of which is an indeterminate sentence, imposed for two or more crimes, other than two or more crimes that include a class A felony, commmitted [FN2] prior to the time the person was imprisoned under any of such sentences, exceeds thirty years, and one of the sentences was imposed for a class B felony, the following rules shall apply:
- (A) if the aggregate maximum term of the determinate sentence or sentences exceeds thirty years, the defendant shall be deemed to be serving a determinate sentence of thirty years;
- (B) if the aggregate maximum term of the determinate sentence or sentences is less than thirty years, the defendant shall be deemed to be serving an indeterminate sentence the maximum term of which shall be deemed to be thirty years. In such instances, the minimum sentence shall be deemed to be fifteen years or six-sevenths of the term or aggregate maximum term of the determinate sentence or sentences, whichever is greater.
- (iv) Notwithstanding subparagraph (i) of this paragraph, the aggregate maximum term of consecutive sentences, all of which are indeterminate sentences or all of which are determinate sentences, imposed for the conviction of two violent felony offenses committed prior to

the time the person was imprisoned under any of such sentences and one of which is a class B violent felony offense, shall, if it exceeds forty years, be deemed to be forty years

- (v) Notwithstanding subparagraphs (ii) and (iii) of this paragraph, where the aggregate maximum term of two or more consecutive sentences, one or more of which is a determinate sentence and one or more of which is an indeterminate sentence, and where such sentences are imposed for the conviction of two violent felony offenses committed prior to the time the person was imprisoned under any such sentences and where one of which is a class B violent felony offense, the following rules shall apply:
- (A) if the aggregate maximum term of the determinate sentence or sentences exceeds forty years, the defendant shall be deemed to be serving a determinate sentence of forty years;
- (B) if the aggregate maximum term of the determinate sentence or sentences is less than forty years, the defendant shall be deemed to be serving an indeterminate sentence the maximum term of which shall be deemed to be forty years. In such instances, the minimum sentence shall be deemed to be twenty years or six-sevenths of the term or aggregate maximum term of the determinate sentence or sentences, whichever is greater.
- (vi) Notwithstanding subparagraphs (i) and (iv) of this paragraph, the aggregate maximum term of consecutive sentences, all of which are indeterminate or all of which are determinate sentences, imposed for the conviction of three or more violent felony offenses committed prior to the time the person was imprisoned under any of such sentences and one of which is a class B violent felony offense, shall, if it exceeds fifty years, be deemed to be fifty years;
- (vii) Notwithstanding subparagraphs (ii), (iii) and (v) of this paragraph, where the aggregate maximum term of two or more consecutive sentences, one or more of which is a determinate sentence and one or more of which is an indeterminate sentence, and where such sentences are imposed for the conviction of three or more violent felony offenses committed prior to the time the person was imprisoned under any such sentences and one of which is a class B violent felony offense, the following rules shall apply:

- (A) if the aggregate maximum term of the determinate sentence or sentences exceeds fifty years, the defendant shall be deemed to be serving a determinate sentence of fifty years.
- (B) if the aggregate maximum term of the determinate sentence or sentences is less than fifty years, the defendant shall be deemed to be serving an indeterminate sentence the maximum term of which shall be deemed to be fifty years. In such instances, the minimum sentence shall be deemed to be twenty-five years or six-sevenths of the term or aggregate maximum term of the determinate sentence or sentences, whichever is greater.
- (viii) Notwithstanding any provision of this subdivision to the contrary where a person is serving two or more consecutive sentences, one or more of which is an indeterminate sentence and one or more of which is a determinate sentence, and if he would be eligible for a reduction provision pursuant to this subdivision if the maximum term or aggregate maximum term of the indeterminate sentence or sentences were added to the term or aggregate maximum term of the determinate sentence or sentences, the person shall be deemed to be eligible for the applicable reduction provision and the rules set forth in this subdivision shall apply.
- (f) The aggregate maximum term of consecutive sentences imposed upon a juvenile offender for two or more crimes, not including a class A felony, committed before he has reached the age of sixteen, shall, if it exceeds ten years, be deemed to be ten years. If consecutive indeterminate sentences imposed upon a juvenile offender include a sentence for the class A felony of arson in the first degree or for the class A felony of kidnapping in the first degree, then the aggregate maximum term of such sentences shall, if it exceeds fifteen years, be deemed to be fifteen years. Where the aggregate maximum term of two or more consecutive sentences is reduced by a calculation made pursuant to this paragraph, the aggregate minimum period of imprisonment, if it exceeds one-half of the aggregate maximum term as so reduced, shall be deemed to be one-half of the aggregate maximum term as so reduced.
- 2. Definite sentences. A definite sentence of imprisonment commences when the prisoner is received in the institution named in the commitment. Where a person is under more than one definite sentence, the sentences shall be calculated as follows:

- (a) If the sentences run concurrently and are to be served in a single institution, the terms merge in and are satisfied by discharge of the term which has the longest unexpired time to run;
- (b) If the sentences run consecutively and are to be served in a single institution, the terms are added to arrive at an aggregate term and are satisfied by discharge of such aggregate term, or by service of two years imprisonment plus any term imposed for an offense committed while the person is under the sentences, whichever is less;
- (c) If the sentences run concurrently and are to be served in more than one institution, the term of each such sentence shall be credited with the portion of any concurrent term served after that sentence was imposed;
- (d) If the sentences run consecutively and are to be served in more than one institution, the aggregate of the time served in all of the institutions shall not exceed two years plus any term imposed for an offense committed while the person is under the sentences.
- 2-a. Undischarged imprisonment in other jurisdiction. Where a person who is subject to an undischarged term of imprisonment imposed at a previous time by a court of another jurisdiction is sentenced to an additional term or terms of imprisonment by a court of this state, to run concurrently with such undischarged term, such additional term or terms shall be deemed to commence when the said person is returned to the custody of the appropriate official of such other jurisdiction where the undischarged term of imprisonment is being served. If the additional term or terms imposed shall run consecutively to the said undischarged term, such additional term or terms shall commence when the prisoner is received in the appropriate institution as provided in subdivisions one and two of this section. The term or terms of such imprisonment shall be calculated and such other pertinent provisions of this section applied in the same manner as where a person is under more than one sentence in this state as provided in this section.
- 3. Jail time. The term of a definite sentence, a determinate sentence, or the maximum term of an indeterminate sentence imposed on a person shall be credited with and diminished by the amount of time the person spent in custody prior to the commencement of such sentence as a result of the charge that culminated in the sentence. In the case

- of an indeterminate sentence, if the minimum period of imprisonment has been fixed by the court or by the board of parole, the credit shall also be applied against the minimum period. The credit herein provided shall be calculated from the date custody under the charge commenced to the date the sentence commences and shall not include any time that is credited against the term or maximum term of any previously imposed sentence or period of post-release supervision to which the person is subject. Where the charge or charges culminate in more than one sentence, the credit shall be applied as follows:
- (a) If the sentences run concurrently, the credit shall be applied against each such sentence;
- (b) If the sentences run consecutively, the credit shall be applied against the aggregate term or aggregate maximum term of the sentences and against the aggregate minimum period of imprisonment.

In any case where a person has been in custody due to a charge that culminated in a dismissal or an acquittal, the amount of time that would have been credited against a sentence for such charge, had one been imposed, shall be credited against any sentence that is based on a charge for which a warrant or commitment was lodged during the pendency of such custody.

- 4. Good behavior time. Time allowances earned for good behavior, pursuant to the provisions of the correction law, shall be computed and applied as follows:
- (a) In the case of a person serving an indeterminate or determinate sentence, the total of such allowances shall be calculated as provided in section eight hundred three of the correction law and the allowances shall be applied as provided in paragraph (b) of subdivision one of section 70.40;
- (b) In the case of a person serving a definite sentence, the total of such allowances shall not exceed one-third of his term or aggregate term and the allowances shall be applied as a credit against such term.
- 5. Time served under vacated sentence. When a sentence of imprisonment that has been imposed on a person is vacated and a new sentence is imposed on such person for the same offense, or for an offense based upon the same act, the new sentence shall be calculated as if it had commenced at the time the vacated sentence

commenced, and all time credited against the vacated sentence shall be credited against the new sentence. In any case where a vacated sentence also includes a period of post-release supervision, all time credited against the period of post-release supervision shall be credited against the period of post-release supervision included with the new sentence. In the event a period of post-release supervision is not included with the new sentence, such period shall be credited against the new sentence.

- 6. Escape. When a person who is serving a sentence of imprisonment escapes from custody, the escape shall interrupt the sentence and such interruption shall continue until the return of the person to the institution in which the sentence was being served or, if the sentence was being served in an institution under the jurisdiction of the state department of corrections and community supervision, to an institution under the jurisdiction of that department. Any time spent by such person in custody from the date of escape to the date the sentence resumes shall be credited against the term or maximum term of the interrupted sentence, provided:
- (a) That such custody was due to an arrest or surrender based upon the escape; or
- (b) That such custody arose from an arrest on another charge which culminated in a dismissal or an acquittal; or
- (c) That such custody arose from an arrest on another charge which culminated in a conviction, but in such case, if a sentence of imprisonment was imposed, the credit allowed shall be limited to the portion of the time spent in custody that exceeds the period, term or maximum term of imprisonment imposed for such conviction.....

[FN1] So in original. Probably should be "serving a".

[FN2] So in original ("commmitted" probably should be "committed").

§ 70.35. Merger of certain definite and indeterminate or determinate sentences

The service of an indeterminate or determinate sentence of imprisonment shall satisfy any definite sentence of imprisonment imposed on a person for an offense committed prior to the time the indeterminate or determinate sentence was imposed, except as provided in paragraph (b) of subdivision five of section 70.25 of this article. A

person who is serving a definite sentence at the time an indeterminate or determinate sentence is imposed shall be delivered to the custody of the state department of corrections and community supervision to commence service of the indeterminate or determinate sentence immediately unless the person is serving a definite sentence pursuant to paragraph (b) of subdivision five of section 70.25 of this article. In any case where the indeterminate or determinate sentence is revoked or vacated, the person shall receive credit against the definite sentence for each day spent in the custody of the state department of corrections and community supervision.

§ 70.40 Release on parole; conditional release; presumptive release

- 1. Indeterminate sentence.
- (a) Release on parole shall be in the discretion of the state board of parole, and such person shall continue service of his or her sentence or sentences while on parole, in accordance with and subject to the provisions of the executive law and the correction law.
- (i) A person who is serving one or more than one indeterminate sentence of imprisonment may be paroled from the institution in which he or she is confined at any time after the expiration of the minimum or the aggregate minimum period of the sentence or sentences or, where applicable, the minimum or aggregate minimum period reduced by the merit time allowance granted pursuant to paragraph (d) of subdivision one of section eight hundred three of the correction law.
- (ii) A person who is serving one or more than one determinate sentence of imprisonment shall be ineligible for discretionary release on parole.
- (iii) A person who is serving one or more than one indeterminate sentence of imprisonment and one or more than one determinate sentence of imprisonment, which run concurrently may be paroled at any time after the expiration of the minimum period of imprisonment of the indeterminate sentence or sentences, or upon the expiration of six-sevenths of the term of imprisonment of the determinate sentence or sentences, whichever is later.
- (iv) A person who is serving one or more than one indeterminate sentence of imprisonment and one or more than one determinate sentence of imprisonment which run consecutively may be

paroled at any time after the expiration of the sum of the minimum or aggregate minimum period of the indeterminate sentence or sentences and six-sevenths of the term or aggregate term of imprisonment of the determinate sentence or sentences.

- (v) Notwithstanding any other subparagraph of this paragraph, a person may be paroled from the institution in which he or she is confined at any time on medical parole pursuant to section two hundred fifty-nine-r or section two hundred fifty-nine-s of the executive law or for deportation pursuant to paragraph (d) of subdivision two of section two hundred fifty-nine-i of the executive law or after the successful completion of a shock incarceration program pursuant to article twenty-six-A of the correction law.
- (ii) A person who is serving one or more than one indeterminate sentence of imprisonment may be paroled from the institution in which he or she is confined at any time after the expiration of the minimum or the aggregate minimum period of the sentence or sentences.
- (b) A person who is serving one or more than one indeterminate or determinate sentence of imprisonment shall, if he or she so requests, be conditionally released from the institution in which he or she is confined when the total good behavior time allowed to him or her, pursuant to the provisions of the correction law, is equal to the unserved portion of his or her term, maximum term or aggregate maximum term; provided, however, that (i) in no event shall a person serving one or more indeterminate sentence of imprisonment and one or more determinate sentence of imprisonment which run concurrently be conditionally released until serving at least six-sevenths of the determinate term of imprisonment which has the longest unexpired time to run and (ii) in no event shall a person be conditionally released prior to the date on which such person is first eligible for discretionary parole release. The conditions of release, including those governing post-release supervision, shall be such as may be imposed by the state board of parole in accordance with the provisions of the executive law.

Every person so released shall be under the supervision of the state department of corrections and community supervision for a period equal to the unserved portion of the term, maximum term, aggregate maximum term, or period of post-release supervision.

- (c) A person who is serving one or more than one indeterminate sentence of imprisonment shall, if he or she so requests, be released from the institution in which he or she is confined if granted presumptive release pursuant to section eight hundred six of the correction law. The conditions of release shall be such as may be imposed by the state board of parole in accordance with the provisions of the executive law. Every person so released shall be under the supervision of the department of corrections and community supervision for a period equal to the unserved portion of his or her maximum or aggregate maximum term unless discharged in accordance with law.
- 2. Definite sentence. A person who is serving one or more than one definite sentence of imprisonment with a term or aggregate term in excess of ninety days, and is eligible for release according to the criteria set forth in paragraphs (a), (b) and (c) of subdivision one of section two hundred seventy-three of the correction law, may, if he or she so requests, be conditionally released from the institution in which he or she is confined at any time after service of sixty days of that term. exclusive of credits allowed under subdivisions four and six of section 70.30. In computing service of sixty days, the credit allowed for jail time under subdivision three of section 70.30 shall be calculated as time served. Conditional release from such institution shall be in the discretion of the parole board, or a local conditional release commission established pursuant to article twelve of the correction law, provided, however that where such release is by a local conditional release commission, the person must be serving a definite sentence with a term in excess of one hundred twenty days and may only be released after service of ninety days of such term. In computing service of ninety days, the credit allowed for jail time under subdivision three of section 70.30 of this article shall be calculated as time served. A conditional release granted under this subdivision shall be upon such conditions as may be imposed by the parole board, in accordance with the provisions of the executive law, or a local conditional release commission in accordance with the provisions of the correction law.

Conditional release shall interrupt service of the sentence or sentences and the remaining portion of the term or aggregate term shall be held in abeyance. Every person so released shall be under the supervision of the department of corrections

and community supervision or a local probation department and in the custody of the local conditional release commission in accordance with article twelve of the correction law, for a period of one year. The local probation department shall cause complete records to be kept of every person released to its supervision pursuant to this subdivision. The department of corrections and community supervision may supply to a local probation department and the local conditional release commission custody information and records maintained on persons under the supervision of such local probation department to aid in the performance of its supervision responsibilities. Compliance with the conditions of release during the period of supervision shall satisfy the portion of the term or aggregate term that has been held in abeyance.

3. Delinquency.

- (a) When a person is alleged to have violated the terms of presumptive release or parole and the state board of parole has declared such person to be delinquent, the declaration of delinquency shall interrupt the person's sentence as of the date of the delinquency and such interruption shall continue until the return of the person to an institution under the jurisdiction of the state department of corrections and community supervision.
- (b) When a person is alleged to have violated the terms of his or her conditional release or post-release supervision and has been declared delinquent by the parole board or the local conditional release commission having supervision over such person, the declaration of delinquency shall interrupt the period of supervision or post-release supervision as of the date of the delinquency. For a conditional release, such interruption shall continue until the return of the person to the institution from which he or she was released or, if he or she was released from an institution under the jurisdiction of the state department of corrections and community supervision, to an institution under the jurisdiction of that department. Upon such return, the person shall resume service of his or her sentence. For a person released to post-release supervision, the provisions of section 70.45 shall apply.
- (c) Any time spent by a person in custody from the time of delinquency to the time service of the sentence resumes shall be credited against the term or maximum term of the interrupted sentence.

provided:

- (i) that such custody was due to an arrest or surrender based upon the delinquency; or
- (ii) that such custody arose from an arrest on another charge which culminated in a dismissal or an acquittal; or
- (iii) that such custody arose from an arrest on another charge which culminated in a conviction, but in such case, if a sentence of imprisonment was imposed, the credit allowed shall be limited to the portion of the time spent in custody that exceeds the period, term or maximum term of imprisonment imposed for such conviction.

§ 70.45 Determinate sentence; post-release supervision

- 1. In general. When a court imposes a determinate sentence it shall in each case state not only the term of imprisonment, but also an additional period of post-release supervision as determined pursuant to this article. Such period shall commence as provided in subdivision five of this section and a violation of any condition of supervision occurring at any time during such period of post-release supervision shall subject the defendant to a further period of imprisonment up to the balance of the remaining period of post-release supervision, not to exceed five years; provided, however, that a defendant serving a term of post-release supervision for a conviction of a felony sex offense, as defined in section 70.80 of this article, may be subject to a further period of imprisonment up to the balance of the remaining period of post-release supervision. Such maximum limits shall not preclude a longer period of further imprisonment for a violation where the defendant is subject to indeterminate and determinate sentences.
- 1-a. When, following a final hearing, a time assessment has been imposed upon a person convicted of a felony sex offense who owes three years or more on a period of post-release supervision, imposed pursuant to subdivision two-a of this section, such defendant, after serving three years of the time assessment, shall be reviewed by the board of parole and may be re-released to post-release supervision only upon a determination by the board of parole made in accordance with subdivision two of section two hundred fifty-nine-i of the executive law. If re-release is not granted, the board shall specify a date not more than twenty-four months from such determination for

reconsideration, and the procedures to be followed upon reconsideration shall be the same. If a time assessment of less than three years is imposed upon such a defendant, the defendant shall be released upon the expiration of such time assessment, unless he or she is subject to further imprisonment or confinement under any provision of law.

- 2. Period of post-release supervision for other than felony sex offenses. The period of post-release supervision for a determinate sentence, other than a determinate sentence imposed for a felony sex offense as defined in paragraph (a) of subdivision one of section 70.80 of this article, shall be five years except that:
- (a) such period shall be one year whenever a determinate sentence of imprisonment is imposed pursuant to subdivision two of section 70.70 of this article upon a conviction of a class D or class E felony offense;
- (b) such period shall be not less than one year nor more than two years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision two of section 70.70 of this article upon a conviction of a class B or class C felony offense;
- (c) such period shall be not less than one year nor more than two years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision three or four of section 70.70 of this article upon conviction of a class D or class E felony offense;
- (d) such period shall be not less than one and one-half years nor more than three years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision three or four of section 70.70 of this article upon conviction of a class B felony or class C felony offense;
- (e) such period shall be not less than one and one-half years nor more than three years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision three of section 70.02 of this article upon a conviction of a class D or class E violent felony offense;
- (f) such period shall be not less than two and one-half years nor more than five years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision three of section 70.02 of this article upon a conviction of a class B or class C violent felony offense.
 - 2-a. Periods of post-release supervision for felony

- sex offenses. The period of post-release supervision for a determinate sentence imposed for a felony sex offense as defined in paragraph (a) of subdivision one of section 70.80 of this article shall be as follows:
- (a) not less than three years nor more than ten years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision four of section 70.80 of this article upon a conviction of a class D or class E felony sex offense;
- (b) not less than five years nor more than fifteen years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision four of section 70.80 of this article upon a conviction of a class C felony sex offense;
- (c) not less than five years nor more than twenty years whenever a determinate sentence of imprisonment is imposed pursuant to subdivision four of section 70.80 of this article upon a conviction of a class B felony sex offense;
- (d) not less than three years nor more than ten years whenever a determinate sentence is imposed pursuant to subdivision three of section 70.02 of this article upon a conviction of a class D or class E violent felony sex offense as defined in paragraph (b) of subdivision one of section 70.80 of this article:
- (e) not less than five years nor more than fifteen years whenever a determinate sentence is imposed pursuant to subdivision three of section 70.02 of this article upon a conviction of a class C violent felony sex offense as defined in section 70.80 of this article;
- (f) not less than five years nor more than twenty years whenever a determinate sentence is imposed pursuant to subdivision three of section 70.02 of this article upon a conviction of a class B violent felony sex offense as defined in section 70.80 of this article;
- (g) not less than five years nor more than fifteen years whenever a determinate sentence of imprisonment is imposed pursuant to either section 70.04, section 70.06, or subdivision five of section 70.80 of this article upon a conviction of a class D or class E violent or non-violent felony sex offense as defined in section 70.80 of this article;
- (h) not less than seven years nor more than twenty years whenever a determinate sentence of imprisonment is imposed pursuant to either section

- 70.04, section 70.06, or subdivision five of section 70.80 of this article upon a conviction of a class C violent or non-violent felony sex offense as defined in section 70.80 of this article;
- (i) such period shall be not less than ten years nor more than twenty-five years whenever a determinate sentence of imprisonment is imposed pursuant to either section 70.04, section 70.06, or subdivision five of section 70.80 of this article upon a conviction of a class B violent or non-violent felony sex offense as defined in section 70.80 of this article; and
- (j) such period shall be not less than ten years nor more than twenty years whenever any determinate sentence of imprisonment is imposed pursuant to subdivision four of section 70.07 of this article.
- 3. Conditions of post-release supervision. The board of parole shall establish and impose conditions of post-release supervision in the same manner and to the same extent as it may establish and impose conditions in accordance with the executive law upon persons who are granted parole or conditional release; provided that, notwithstanding any other provision of law, the board of parole may impose as a condition of post-release supervision that for a period not exceeding six months immediately following release from the underlying term of imprisonment the person be transferred to and participate in the programs of a residential treatment facility as that term is defined in subdivision six of section two of the correction law. Upon release from the underlying term of imprisonment, the person shall be furnished with a written statement setting forth the conditions of post-release supervision in sufficient detail to provide for the person's conduct and supervision.
- 4. Revocation of post-release supervision. An alleged violation of any condition of post-release supervision shall be initiated, heard and determined in accordance with the provisions of subdivisions three and four of section two hundred fifty-nine-i of the executive law.
- 5. Calculation of service of period of post-release supervision. A period or periods of post-release supervision shall be calculated and served as follows:
- (a) A period of post-release supervision shall commence upon the person's release from imprisonment to supervision by the department of

- corrections and community supervision and shall interrupt the running of the determinate sentence or sentences of imprisonment and the indeterminate sentence or sentences of imprisonment, if any. The remaining portion of any maximum or aggregate maximum term shall then be held in abeyance until the successful completion of the period of post-release supervision or the person's return to the custody of the department of corrections and community supervision, whichever occurs first.
- (b) Upon the completion of the period of post-release supervision, the running of such sentence or sentences of imprisonment shall resume and only then shall the remaining portion of any maximum or aggregate maximum term previously held in abeyance be credited with and diminished by such period of post-release supervision. The person shall then be under the jurisdiction of the department of corrections and community supervision for the remaining portion of such maximum or aggregate maximum term.
- (c) When a person is subject to two or more periods of post-release supervision, such periods shall merge with and be satisfied by discharge of the period of post-release supervision having the longest unexpired time to run; provided, however, any time served upon one period of post-release supervision shall not be credited to any other period of post-release supervision except as provided in subdivision five of section 70.30 of this article.
- (d) When a person is alleged to have violated a condition of post-release supervision and the department of corrections and community supervision has declared such person to be delinquent: (i) the declaration of delinquency shall interrupt the period of post-release supervision; (ii) such interruption shall continue until the person is restored to post-release supervision; (iii) if the person is restored to post-release supervision without being returned to the department of corrections and community supervision, any time spent in custody from the date of delinquency until restoration to post-release supervision shall first be credited to the maximum or aggregate maximum term of the sentence or sentences of imprisonment, but only to the extent authorized by subdivision three of section 70.40 of this article. Any time spent in custody solely pursuant to such delinquency after completion of the maximum or aggregate maximum term of the sentence or

sentences of imprisonment shall be credited to the period of post-release supervision, if any; and (iv) if the person is ordered returned to the department of corrections and community supervision, the person shall be required to serve the time assessment before being re-released to post-release supervision. In the event the balance of the remaining period of post-release supervision is six months or less, such time assessment may be up to six months unless a longer period is authorized pursuant to subdivision one of this section. The time assessment shall commence upon the issuance of a determination after a final hearing that the person has violated one or more conditions of supervision. While serving such assessment, the person shall not receive any good behavior allowance pursuant to section eight hundred three of the correction law. Any time spent in custody from the date of delinquency until return to the department of corrections and community supervision shall first be credited to the maximum or aggregate maximum term of the sentence or sentences of imprisonment, but only to the extent authorized by subdivision three of section 70.40 of this article. The maximum or aggregate maximum term of the sentence or sentences of imprisonment shall run while the person is serving such time assessment in the custody of the department of corrections and community supervision. Any time spent in custody solely pursuant to such delinquency after completion of the maximum or aggregate maximum term of the sentence or sentences of imprisonment shall be credited to the period of post-release supervision, if any.

- (e) Notwithstanding paragraph (d) of this subdivision, in the event a person is sentenced to one or more additional indeterminate or determinate term or terms of imprisonment prior to the completion of the period of post-release supervision, such period of post-release supervision shall be held in abeyance and the person shall be committed to the custody of the department of corrections and community supervision in accordance with the requirements of the prior and additional terms of imprisonment.
- (f) When a person serving a period of post-release supervision is returned to the department of corrections and community supervision pursuant to an additional consecutive sentence of imprisonment and without a declaration of delinquency, such period of post-release supervision shall be held in abeyance while the person is in the custody of the

department of corrections and community supervision. Such period of post-release supervision shall resume running upon the person's re-release.

§ 70.70 Sentence of imprisonment for felony drug offender other than a class A felony

- 1. For the purposes of this section, the following terms shall mean:
- (a) "Felony drug offender" means a defendant who stands convicted of any felony, defined in article two hundred twenty or two hundred twenty-one of this chapter other than a class A felony.
- (b) "Second felony drug offender" means a second felony offender as that term is defined in subdivision one of section 70.06 of this article, who stands convicted of any felony, defined in article two hundred twenty or two hundred twenty-one of this chapter other than a class A felony.
- (c) "Violent felony" shall have the same meaning as that term is defined in subdivision one of section 70.02 of this article.
- 2. Except as provided in subdivision three or four of this section, a sentence of imprisonment for a felony drug offender shall be a determinate sentence as provided in paragraph (a) of this subdivision.
- (a) Term of determinate sentence. Except as provided in paragraph (b) or (c) of this subdivision, the court shall impose a determinate term of imprisonment upon a felony drug offender which shall be imposed by the court in whole or half years, which shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this article. The terms of imprisonment authorized for such determinate sentences are as follows:
- (i) for a class B felony, the term shall be at least one year and shall not exceed nine years, except that for the class B felony of criminal sale of a controlled substance in or near school grounds as defined in subdivision two of section 220.44 of this chapter or on a school bus as defined in subdivision seventeen of section 220.00 of this chapter or criminal sale of a controlled substance to a child as defined in section 220.48 of this chapter, the term shall be at least two years and shall not exceed nine years;
 - (ii) for a class C felony, the term shall be at least

one year and shall not exceed five and one-half years;

- (iii) for a class D felony, the term shall be at least one year and shall not exceed two and one-half years; and
- (iv) for a class E felony, the term shall be at least one year and shall not exceed one and one-half years.
- (b) Probation. Notwithstanding any other provision of law, the court may sentence a defendant convicted of a class B, class C, class D or class E felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter to probation in accordance with the provisions of sections 60.04 and 65.00 of this chapter.
- (c) Alternative definite sentence for class B, class C, class D, and class E felonies. If the court, having regard to the nature and circumstances of the crime and to the history and character of the defendant, is of the opinion that a sentence of imprisonment is necessary but that it would be unduly harsh to impose a determinate sentence upon a person convicted of a class C, class D or class E felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter, or a class B felony defined in article two hundred twenty of this chapter, other than the class B felony defined in section 220.48 of this chapter, as added by a chapter of the laws of two thousand nine the court may impose a definite sentence of imprisonment and fix a term of one year or less.
- (d) The court may direct that a determinate sentence imposed on a defendant convicted of a class B felony, other than the class B felony defined in section 220.48 of this chapter, pursuant to this subdivision be executed as a sentence of parole supervision in accordance with section 410.91 of the criminal procedure law.
- 3. Sentence of imprisonment for second felony drug offender.
- (a) Applicability. This subdivision shall apply to a second felony drug offender whose prior felony conviction was not a violent felony.
- (b) Authorized sentence. Except as provided in paragraphs (c), (d) and (e) of this subdivision, when the court has found pursuant to the provisions of section 400.21 of the criminal procedure law that a defendant is a second felony drug offender who stands convicted of a class B,

- class C, class D or class E felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter the court shall impose a determinate sentence of imprisonment. Such determinate sentence shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this article. The terms of such determinate sentence shall be imposed by the court in whole or half years as follows:
- (i) for a class B felony, the term shall be at least two years and shall not exceed twelve years;
- (ii) for a class C felony, the term shall be at least one and one-half years and shall not exceed eight years;
- (iii) for a class D felony, the term shall be at least one and one-half years and shall not exceed four years; and
- (iv) for a class E felony, the term shall be at least one and one-half years and shall not exceed two years.
- (c) Probation. Notwithstanding any other provision of law, the court may sentence a second felony drug offender convicted of a class B felony to lifetime probation in accordance with the provisions of section 65.00 of this chapter and may sentence a second felony drug offender convicted of a class C, class D or class E felony to probation in accordance with the provisions of section 65.00 of this chapter.
- (d) Sentence of parole supervision. In the case of a person sentenced for a specified offense or offenses as defined in subdivision five of section 410.91 of the criminal procedure law, who stands convicted of no other felony offense, who has not previously been convicted of either a violent felony offense as defined in section 70.02 of this article, a class A felony offense or a class B felony offense. and is not under the jurisdiction of or awaiting delivery to the department of corrections and community supervision, the court may direct that a determinate sentence imposed pursuant to this subdivision shall be executed as a parole supervision sentence as defined in and pursuant to the procedures prescribed in section 410.91 of the criminal procedure law.
- (e) Alternate definite sentence for class C, class D and class E felonies. If the court, having regard to the nature and circumstances of the crime and to the history and character of the defendant, is of

the opinion that a sentence of imprisonment is necessary but that it would be unduly harsh to impose a determinate sentence upon a person convicted of a class C, class D or class E felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter, the court may impose a definite sentence of imprisonment and fix a term of one year or less.

- 4. Sentence of imprisonment for second felony drug offender previously convicted of a violent felony.
- (a) Applicability. This subdivision shall apply to a second felony drug offender whose prior felony conviction was a violent felony.
- (b) Authorized sentence. When the court has found pursuant to the provisions of section 400.21 of the criminal procedure law that a defendant is a second felony drug offender whose prior felony conviction was a violent felony, who stands convicted of a class B, class C, class D or class E felony offense defined in article two hundred twenty or two hundred twenty-one of this chapter, the court shall impose a determinate sentence of imprisonment. Such determinate sentence shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this article. The terms of such determinate sentence shall be imposed by the court in whole or half years as follows:
- (i) for a class B felony, the term shall be at least six years and shall not exceed fifteen years;
- (ii) for a class C felony, the term shall be at least three and one-half years and shall not exceed nine years;
- (iii) for a class D felony, the term shall be at least two and one-half years and shall not exceed four and one-half years; and
- (iv) for a class E felony, the term shall be at least two years and shall not exceed two and one-half years.

§ 70.71 Sentence of imprisonment for a class A felony drug offender

- 1. For the purposes of this section, the following terms shall mean:
- (a) "Felony drug offender" means a defendant who stands convicted of any class A felony as defined in article two hundred twenty of this chapter.

- (b) "Second felony drug offender" means a second felony offender as that term is defined in subdivision one of section 70.06 of this article, who stands convicted of and is to be sentenced for any class A felony as defined in article two hundred twenty of this chapter.
- (c) "Violent felony offense" shall have the same meaning as that term is defined in subdivision one of section 70.02 of this article.
- 2. Sentence of imprisonment for a first felony drug offender.
- (a) Applicability. Except as provided in subdivision three, four or five of this section, this subdivision shall apply to a person convicted of a class A felony as defined in article two hundred twenty of this chapter.
- (b) Authorized sentence. The court shall impose a determinate term of imprisonment which shall be imposed by the court in whole or half years and which shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this article. The terms authorized for such determinate sentences are as follows:
- (i) for a class A-I felony, the term shall be at least eight years and shall not exceed twenty years;
- (ii) for a class A-II felony, the term shall be at least three years and shall not exceed ten years.
- (c) Lifetime probation. Notwithstanding any other provision of law, the court may sentence a defendant convicted of a class A-II felony defined in article two hundred twenty of this chapter to lifetime probation in accordance with the provisions of section 65.00 of this chapter.
- 3. Sentence of imprisonment for a second felony drug offender.
- (a) Applicability. This subdivision shall apply to a second felony drug offender whose prior felony conviction or convictions did not include one or more violent felony offenses.
- (b) Authorized sentence. When the court has found pursuant to the provisions of section 400.21 of the criminal procedure law that a defendant is a second felony drug offender who stands convicted of a class A felony as defined in article two hundred twenty or two hundred twenty-one of this chapter, the court shall impose a determinate sentence of imprisonment. Such determinate sentence shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this

article. Such determinate sentence shall be imposed by the court in whole or half years as follows:

- (i) for a class A-I felony, the term shall be at least twelve years and shall not exceed twenty-four years:
- (ii) for a class A-II felony, the term shall be at least six years and shall not exceed fourteen years.
- (c) Lifetime probation. Notwithstanding any other provision of law, the court may sentence a defendant convicted of a class A-II felony defined in article two hundred twenty of this chapter to lifetime probation in accordance with the provisions of section 65.00 of this chapter.
- 4. Sentence of imprisonment for a second felony drug offender previously convicted of a violent felony offense.
- (a) Applicability. This subdivision shall apply to a second felony drug offender whose prior felony conviction was a violent felony.
- (b) Authorized sentence. When the court has found pursuant to the provisions of section 400.21 of the criminal procedure law that a defendant is a second felony drug offender whose prior felony conviction was a violent felony, who stands convicted of a class A felony as defined in article two hundred twenty or two hundred twenty-one of this chapter, the court shall impose a determinate sentence of imprisonment. Such determinate sentence shall include as a part thereof a period of post-release supervision in accordance with section 70.45 of this article. Such determinate sentence shall be imposed by the court in whole or half years as follows:
- (i) for a class A-I felony, the term shall be at least fifteen years and shall not exceed thirty years;
- (ii) for a class A-II felony, the term shall be at least eight years and shall not exceed seventeen years....

Article 80. Fines

§ 80.00 Fine for felony

- 1. A sentence to pay a fine for a felony shall be a sentence to pay an amount, fixed by the court, not exceeding the higher of
 - a. five thousand dollars; or
- b. double the amount of the defendant's gain from the commission of the crime; or

- c. if the conviction is for any felony defined in article two hundred twenty or two hundred twenty-one of this chapter, according to the following schedule:
 - (i) for A-I felonies, one hundred thousand dollars;
 - (ii) for A-II felonies, fifty thousand dollars;
 - (iii) for B felonies, thirty thousand dollars;
 - (iv) for C felonies, fifteen thousand dollars.

When imposing a fine pursuant to the provisions of this paragraph, the court shall consider the profit gained by defendant's conduct, whether the amount of the fine is disproportionate to the conduct in which defendant engaged, its impact on any victims, and defendant's economic circumstances, including the defendant's ability to pay, the effect of the fine upon his or her immediate family or any other persons to whom the defendant owes an obligation of support.

- 2. As used in this section the term "gain" means the amount of money or the value of property derived from the commission of the crime, less the amount of money or the value of property returned to the victim of the crime or seized by or surrendered to lawful authority prior to the time sentence is imposed.
- 3. When the court imposes a fine for a felony pursuant to paragraph b of subdivision one of this section, the court shall make a finding as to the amount of the defendant's gain from the crime. If the record does not contain sufficient evidence to support such a finding or to permit adequate consideration of the matters specified in paragraph c of subdivision one of this section, the court may conduct a hearing upon such issues.
- 4. Exception. The provisions of this section shall not apply to a corporation.
- 5. All moneys in excess of five thousand dollars received or collected in payment of a fine imposed pursuant to paragraph c of subdivision one of this section are the property of the state and the state comptroller shall deposit all such fines to the rehabilitative alcohol and substance treatment fund established pursuant to section ninety-seven-cc of the state finance law.
- 6. Notwithstanding any inconsistent provision of subdivision one of this section a sentence to pay a fine for a felony set forth in the vehicle and traffic law shall be a sentence to pay an amount fixed by the court in accordance with the provisions of the

law that defines the crime.

7. When the court imposes a fine pursuant to section 145.22 or 145.23 of this chapter, the court shall direct that no less than ten percent of such fine be credited to the state cemetery vandalism restoration and administration fund created pursuant to section ninety-seven-r of the state finance law.

§ 80.05 Fines for misdemeanors and violation

- 1. Class A misdemeanor. A sentence to pay a fine for a class A misdemeanor shall be a sentence to pay an amount, fixed by the court, not exceeding one thousand dollars, provided, however, that a sentence imposed for a violation of section 215.80 of this chapter may include a fine in an amount equivalent to double the value of the property unlawfully disposed of in the commission of the crime.
- 2. Class B misdemeanor. A sentence to pay a fine for a class B misdemeanor shall be a sentence to pay an amount, fixed by the court, not exceeding five hundred dollars.
- 3. Unclassified misdemeanor. A sentence to pay a fine for an unclassified misdemeanor shall be a sentence to pay an amount, fixed by the court, in accordance with the provisions of the law or ordinance that defines the crime.
- 4. Violation. A sentence to pay a fine for a violation shall be a sentence to pay an amount, fixed by the court, not exceeding two hundred fifty dollars.

In the case of a violation defined outside this chapter, if the amount of the fine is expressly specified in the law or ordinance that defines the offense, the amount of the fine shall be fixed in accordance with that law or ordinance.

5. Alternative sentence. If a person has gained money or property through the commission of any misdemeanor or violation then upon conviction thereof, the court, in lieu of imposing the fine authorized for the offense under one of the above subdivisions, may sentence the defendant to pay an amount, fixed by the court, not exceeding double the amount of the defendant's gain from the commission of the offense; provided, however, that the amount fixed by the court pursuant to this subdivision upon a conviction under section 11-1904 of the environmental conservation law shall not exceed five thousand dollars. In such event the provisions of subdivisions two and three

of section 80.00 shall be applicable to the sentence.

6. Exception. The provisions of this section shall not apply to a corporation....

§ 80.15 Multiple offenses

Where a person is convicted of two or more offenses committed through a single act or omission, or through an act or omission which in itself constituted one of the offenses and also was a material element of the other, and the court imposes a sentence of imprisonment or a fine or both for one of the offenses, a fine shall not be imposed for the other. The provisions of this section shall not apply to any offense or offenses set forth in the vehicle and traffic law....

PART THREE. SPECIFIC OFFENSES TITLE G. ANTICIPATORY OFFENSES

Article 100. Criminal Solicitation

§ 100.00 Criminal solicitation in the fifth degree

A person is guilty of criminal solicitation in the fifth degree when, with intent that another person engage in conduct constituting a crime, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct.

Criminal solicitation in the fifth degree is a violation.

§ 100.05 Criminal solicitation in the fourth degree

A person is guilty of criminal solicitation in the fourth degree when:

- 1. with intent that another person engage in conduct constituting a felony, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct; or
- 2. being over eighteen years of age, with intent that another person under sixteen years of age engage in conduct that would constitute a crime, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct.

Criminal solicitation in the fourth degree is a class A misdemeanor.

§ 100.08 Criminal solicitation in the third degree

A person is guilty of criminal solicitation in the third degree when, being over eighteen years of age, with intent that another person under sixteen years of age engage in conduct that would constitute a felony, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct.

Criminal solicitation in the third degree is a class E felony.

§ 100.10 Criminal solicitation in the second degree

A person is guilty of criminal solicitation in the second degree when, with intent that another person engage in conduct constituting a class A felony, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct.

Criminal solicitation in the second degree is a class D felony.

§ 100.13 Criminal solicitation in the first degree

A person is guilty of criminal solicitation in the first degree when, being over eighteen years of age, with intent that another person under sixteen years of age engage in conduct that would constitute a class A felony, he solicits, requests, commands, importunes or otherwise attempts to cause such other person to engage in such conduct.

Criminal solicitation in the first degree is a class C felony.

§ 100.15 Criminal solicitation; no defense

It is no defense to a prosecution for criminal solicitation that the person solicited could not be guilty of the crime solicited owing to criminal irresponsibility or other legal incapacity or exemption, or to unawareness of the criminal nature of the conduct solicited or of the defendant's criminal purpose or to other factors precluding the mental state required for the commission of the crime in question.

§ 100.20 Criminal solicitation; exemption

A person is not guilty of criminal solicitation when his solicitation constitutes conduct of a kind that is necessarily incidental to the commission of the crime solicited. When under such circumstances the solicitation constitutes an offense other than criminal solicitation which is related to but separate from the crime solicited, the actor is guilty of such related and separate offense only and not of criminal solicitation.

Article 105. Conspiracy

§ 105.00 Conspiracy in the sixth degree

A person is guilty of conspiracy in the sixth degree when, with intent that conduct constituting a crime be performed, he agrees with one or more persons to engage in or cause the performance of such conduct.

Conspiracy in the sixth degree is a class B misdemeanor.

§ 105.05 Conspiracy in the fifth degree

A person is guilty of conspiracy in the fifth degree when, with intent that conduct constituting:

- 1. a felony be performed, he agrees with one or more persons to engage in or cause the performance of such conduct; or
- 2. a crime be performed, he, being over eighteen years of age, agrees with one or more persons under sixteen years of age to engage in or cause the performance of such conduct.

Conspiracy in the fifth degree is a class A misdemeanor.

§ 105.10 Conspiracy in the fourth degree

A person is guilty of conspiracy in the fourth degree when, with intent that conduct constituting:

- 1. a class B or class C felony be performed, he or she agrees with one or more persons to engage in or cause the performance of such conduct; or
- 2. a felony be performed, he or she, being over eighteen years of age, agrees with one or more persons under sixteen years of age to engage in or cause the performance of such conduct; or
- 3. the felony of money laundering in the third degree as defined in section 470.10 of this chapter, be performed, he or she agrees with one or more persons to engage in or cause the performance of such conduct.

Conspiracy in the fourth degree is a class E felony.

§ 105.13 Conspiracy in the third degree

A person is guilty of conspiracy in the third degree when, with intent that conduct constituting a class B or a class C felony be performed, he, being over

eighteen years of age, agrees with one or more persons under sixteen years of age to engage in or cause the performance of such conduct.

Conspiracy in the third degree is a class D felony.

§ 105.15 Conspiracy in the second degree

A person is guilty of conspiracy in the second degree when, with intent that conduct constituting a class A felony be performed, he agrees with one or more persons to engage in or cause the performance of such conduct.

Conspiracy in the second degree is a class B felony.

§ 105.17 Conspiracy in the first degree

A person is guilty of conspiracy in the first degree when, with intent that conduct constituting a class A felony be performed, he, being over eighteen years of age, agrees with one or more persons under sixteen years of age to engage in or cause the performance of such conduct.

Conspiracy in the first degree is a class A-I felony.

§ 105.20 Conspiracy; pleading and proof; necessity of overt act

A person shall not be convicted of conspiracy unless an overt act is alleged and proved to have been committed by one of the conspirators in furtherance of the conspiracy.

§ 105.25 Conspiracy; jurisdiction and venue

- 1. A person may be prosecuted for conspiracy in the county in which he entered into such conspiracy or in any county in which an overt act in furtherance thereof was committed.
- 2. An agreement made within this state to engage in or cause the performance of conduct in another jurisdiction is punishable herein as a conspiracy only when such conduct would constitute a crime both under the laws of this state if performed herein and under the laws of the other jurisdiction if performed therein.
- 3. An agreement made in another jurisdiction to engage in or cause the performance of conduct within this state, which would constitute a crime herein, is punishable herein only when an overt act in furtherance of such conspiracy is committed within this state. Under such circumstances, it is no defense to a prosecution for conspiracy that the conduct which is the objective of the conspiracy

would not constitute a crime under the laws of the other jurisdiction if performed therein.

§ 105.30 Conspiracy; no defense

It is no defense to a prosecution for conspiracy that, owing to criminal irresponsibility or other legal incapacity or exemption, or to unawareness of the criminal nature of the agreement or the object conduct or of the defendant's criminal purpose or to other factors precluding the mental state required for the commission of conspiracy or the object crime, one or more of the defendant's co-conspirators could not be guilty of conspiracy or the object crime.

§ 105.35 Conspiracy; enterprise corruption: applicability

For purposes of this article, conspiracy to commit the crime of enterprise corruption in violation of section 460.20 of this chapter shall not constitute an offense.

Article 110. Attempt

§ 110.00 Attempt to commit a crime

A person is guilty of an attempt to commit a crime when, with intent to commit a crime, he engages in conduct which tends to effect the commission of such crime.

§ 110.05 Attempt to commit a crime; punishment

An attempt to commit a crime is a:

- 1. Class A-I felony when the crime attempted is the A-I felony of murder in the first degree, aggravated murder as defined in subdivision one of section 125.26 of this chapter, criminal possession of a controlled substance in the first degree, criminal sale of a controlled substance in the first degree, criminal possession of a chemical or biological weapon in the first degree or criminal use of a chemical or biological weapon in the first degree;
- 2. Class A-II felony when the crime attempted is a class A-II felony;
- 3. Class B felony when the crime attempted is a class A-I felony except as provided in subdivision one hereof;
- 4. Class C felony when the crime attempted is a class B felony;
 - 5. Class D felony when the crime attempted is a

class C felony;

- 6. Class E felony when the crime attempted is a class D felony;
- 7. Class A misdemeanor when the crime attempted is a class E felony;
- 8. Class B misdemeanor when the crime attempted is a misdemeanor.

§ 110.10 Attempt to commit a crime; no defense

If the conduct in which a person engages otherwise constitutes an attempt to commit a crime pursuant to section 110.00, it is no defense to a prosecution for such attempt that the crime charged to have been attempted was, under the attendant circumstances, factually or legally impossible of commission, if such crime could have been committed had the attendant circumstances been as such person believed them to be.

Article 115. Criminal Facilitation

§ 115.00 Criminal facilitation in the fourth degree

A person is guilty of criminal facilitation in the fourth degree when, believing it probable that he is rendering aid:

- 1. to a person who intends to commit a crime, he engages in conduct which provides such person with means or opportunity for the commission thereof and which in fact aids such person to commit a felony; or
- 2. to a person under sixteen years of age who intends to engage in conduct which would constitute a crime, he, being over eighteen years of age, engages in conduct which provides such person with means or opportunity for the commission thereof and which in fact aids such person to commit a crime.

Criminal facilitation in the fourth degree is a class A misdemeanor.

§ 115.01 Criminal facilitation in the third degree

A person guilty of criminal facilitation in the third degree, when believing it probable that he is rendering aid to a person under sixteen years of age who intends to engage in conduct that would constitute a felony, he, being over eighteen years of age, engages in conduct which provides such person with means or opportunity for the commission thereof and which in fact aids such person to commit a felony.

Criminal facilitation in the third degree is a class E felony.

§ 115.05 Criminal facilitation in the second degree

A person is guilty of criminal facilitation in the second degree when, believing it probable that he is rendering aid to a person who intends to commit a class A felony, he engages in conduct which provides such person with means or opportunity for the commission thereof and which in fact aids such person to commit such class A felony.

Criminal facilitation in the second degree is a class C felony.

§ 115.08 Criminal facilitation in the first degree

A person is guilty of criminal facilitation in the first degree when, believing it probable that he is rendering aid to a person under sixteen years of age who intends to engage in conduct that would constitute a class A felony, he, being over eighteen years of age, engages in conduct which provides such person with means or opportunity for the commission thereof and which in fact aids such person to commit such a class A felony.

Criminal facilitation in the first degree is a class B felony.

§ 115.10 Criminal facilitation; no defense

It is no defense to a prosecution for criminal facilitation that:

- 1. The person facilitated was not guilty of the underlying felony owing to criminal irresponsibility or other legal incapacity or exemption, or to unawareness of the criminal nature of the conduct in question or to other factors precluding the mental state required for the commission of such felony; or
- 2. The person facilitated has not been prosecuted for or convicted of the underlying felony, or has previously been acquitted thereof; or
- 3. The defendant himself is not guilty of the felony which he facilitated because he did not act with the intent or other culpable mental state required for the commission thereof.

§ 115.15 Criminal facilitation; corroboration

A person shall not be convicted of criminal facilitation upon the testimony of a person who has committed the felony charged to have been facilitated unless such testimony be corroborated

by such other evidence as tends to connect the defendant with such facilitation.

TITLE H. OFFENSES AGAINST THE PERSON INVOLVING PHYSICAL INJURY, SEXUAL CONDUCT, RESTRAINT AND INTIMIDATION

Article 120. Assault and Related Offenses

§ 120.00 Assault in the third degree

A person is guilty of assault in the third degree when:

- 1. With intent to cause physical injury to another person, he causes such injury to such person or to a third person; or
- 2. He recklessly causes physical injury to another person; or
- 3. With criminal negligence, he causes physical injury to another person by means of a deadly weapon or a dangerous instrument.

Assault in the third degree is a class A misdemeanor.

- § 120.01 Reckless assault of a child by a child day care provider...
- § 120.02 Reckless assault of a child...
- § 120.03 Vehicular assault in the second degree...
- § 120.04 Vehicular assault in the first degree...
- § 120.04-a Aggravated vehicular assault...
- § 120.05 Assault in the second degree

A person is guilty of assault in the second degree when:

- 1. With intent to cause serious physical injury to another person, he causes such injury to such person or to a third person; or
- 2. With intent to cause physical injury to another person, he causes such injury to such person or to a third person by means of a deadly weapon or a dangerous instrument; or
- 3. With intent to prevent a peace officer, a police officer, registered nurse, licensed practical nurse, sanitation enforcement agent, New York city sanitation worker, a firefighter, including a firefighter acting as a paramedic or emergency

medical technician administering first aid in the course of performance of duty as such firefighter, an emergency medical service paramedic or emergency medical service technician, or medical or related personnel in a hospital emergency department, a city marshal, a traffic enforcement officer or traffic enforcement agent, from performing a lawful duty, by means including releasing or failing to control an animal under circumstances evincing the actor's intent that the animal obstruct the lawful activity of such peace officer, police officer, registered nurse, licensed practical nurse, sanitation enforcement agent, New York city sanitation worker, firefighter, paramedic, technician, city marshal, traffic enforcement officer or traffic enforcement agent, he or she causes physical injury to such peace officer, police officer, registered nurse, licensed practical nurse, sanitation enforcement agent, New York city sanitation worker, firefighter, paramedic, technician or medical or related personnel in a hospital emergency department, city marshal, traffic enforcement officer or traffic enforcement agent; or...

- 4. He recklessly causes serious physical injury to another person by means of a deadly weapon or a dangerous instrument; or...
- 6. In the course of and in furtherance of the commission or attempted commission of a felony, other than a felony defined in article one hundred thirty which requires corroboration for conviction, or of immediate flight therefrom, he, or another participant if there be any, causes physical injury to a person other than one of the participants; or...
- 8. Being eighteen years old or more and with intent to cause physical injury to a person less than eleven years old, the defendant recklessly causes serious physical injury to such person; or
- 9. Being eighteen years old or more and with intent to cause physical injury to a person less than seven years old, the defendant causes such injury to such person...

Assault in the second degree is a class D felony.

- § 120.06 Gang assault in the second degree...
- § 120.07 Gang assault in the first degree...
- § 120.08 Assault on a peace officer, police officer, fireman or emergency medical services professional

A person is guilty of assault on a peace officer,

police officer, fireman or emergency medical services professional when, with intent to prevent a peace officer, police officer, a fireman, including a fireman acting as a paramedic or emergency medical technician administering first aid in the course of performance of duty as such fireman, or an emergency medical service paramedic or emergency medical service technician, from performing a lawful duty, he causes serious physical injury to such peace officer, police officer, fireman, paramedic or technician.

Assault on a peace officer, police officer, fireman or emergency medical services professional is a class C felony.

§ 120.09. Assault on a judge....

§ 120.10 Assault in the first degree

A person is guilty of assault in the first degree when:

- 1. With intent to cause serious physical injury to another person, he causes such injury to such person or to a third person by means of a deadly weapon or a dangerous instrument; or
- 2. With intent to disfigure another person seriously and permanently, or to destroy, amputate or disable permanently a member or organ of his body, he causes such injury to such person or to a third person; or
- 3. Under circumstances evincing a depraved indifference to human life, he recklessly engages in conduct which creates a grave risk of death to another person, and thereby causes serious physical injury to another person; or
- 4. In the course of and in furtherance of the commission or attempted commission of a felony or of immediate flight therefrom, he, or another participant if there be any, causes serious physical injury to a person other than one of the participants.

Assault in the first degree is a class B felony.

§ 120.11 Aggravated assault upon a police officer or a peace officer...

§ 120.12 Aggravated assault upon a person less than eleven years old...

§ 120.13 Menacing in the first degree

A person is guilty of menacing in the first degree when he or she commits the crime of menacing in the second degree and has been previously convicted of the crime of menacing in the second degree or the crime of menacing a police officer or peace officer within the preceding ten years.

Menacing in the first degree is a class E felony.

§ 120.14 Menacing in the second degree

A person is guilty of menacing in the second degree when:

- 1. He or she intentionally places or attempts to place another person in reasonable fear of physical injury, serious physical injury or death by displaying a deadly weapon, dangerous instrument or what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm; or
- 2. He or she repeatedly follows a person or engages in a course of conduct or repeatedly commits acts over a period of time intentionally placing or attempting to place another person in reasonable fear of physical injury, serious physical injury or death; or
- 3. He or she commits the crime of menacing in the third degree in violation of that part of a duly served order of protection, or such order which the defendant has actual knowledge of because he or she was present in court when such order was issued, pursuant to article eight of the family court act, section 530.12 of the criminal procedure law, or an order of protection issued by a court of competent jurisdiction in another state, territorial or tribal jurisdiction, which directed the respondent or defendant to stay away from the person or persons on whose behalf the order was issued.

Menacing in the second degree is a class A misdemeanor.

§ 120.15 Menacing in the third degree

A person is guilty of menacing in the third degree when, by physical menace, he or she intentionally places or attempts to place another person in fear of death, imminent serious physical injury or physical injury.

Menacing in the third degree is a class B misdemeanor.

§ 120.16 Hazing in the first degree...

§ 120.17 Hazing in the second degree...

§ 120.18 Menacing a police officer or peace officer...

§ 120.20 Reckless endangerment in the second degree

A person is guilty of reckless endangerment in the second degree when he recklessly engages in conduct which creates a substantial risk of serious physical injury to another person.

Reckless endangerment in the second degree is a class A misdemeanor.

§ 120.25 Reckless endangerment in the first degree

A person is guilty of reckless endangerment in the first degree when, under circumstances evincing a depraved indifference to human life, he recklessly engages in conduct which creates a grave risk of death to another person.

Reckless endangerment in the first degree is a class D felony.

§ 120.30 Promoting a suicide attempt...

§ 120.35 Promoting a suicide attempt; when punishable as attempt to commit murder...

§ 120.40 Definitions

For purposes of sections 120.45, 120.50, 120.55 and 120.60 of this article:

- 1. "Kidnapping" shall mean a kidnapping crime defined in article one hundred thirty-five of this chapter.
- 2. "Unlawful imprisonment" shall mean an unlawful imprisonment felony crime defined in article one hundred thirty-five of this chapter.
- 3. "Sex offense" shall mean a felony defined in article one hundred thirty of this chapter, sexual misconduct, as defined in section 130.20 of this chapter, sexual abuse in the third degree as defined in section 130.55 of this chapter or sexual abuse in the second degree as defined in section 130.60 of this chapter.
- 4. "Immediate family" means the spouse, former spouse, parent, child, sibling, or any other person who regularly resides or has regularly resided in the household of a person.
 - 5. "Specified predicate crime" means:
 - a. a violent felony offense;
- b. a crime defined in section 130.20, 130.25, 130.30, 130.40, 130.45, 130.55, 130.60, 130.70, 255.25, 255.26 or 255.27;
- c. assault in the third degree, as defined in section 120.00; menacing in the first degree, as defined in section 120.13; menacing in the second

degree, as defined in section 120.14; coercion in the first degree, as defined in section 135.65; coercion in the second degree, as defined in section 135.60; aggravated harassment in the second degree, as defined in section 240.30; harassment in the first degree, as defined in section 240.25; menacing in the third degree, as defined in section 120.15; criminal mischief in the third degree, as defined in section 145.05; criminal mischief in the second degree, as defined in section 145.10, criminal mischief in the first degree, as defined in section 145.12; criminal tampering in the first degree, as defined in section 145.20; arson in the fourth degree, as defined in section 150.05; arson in the third degree, as defined in section 150.10; criminal contempt in the first degree, as defined in section 215.51; endangering the welfare of a child, as defined in section 260.10; or

- d. stalking in the fourth degree, as defined in section 120.45; stalking in the third degree, as defined in section 120.50; stalking in the second degree, as defined in section 120.55; or
- e. an offense in any other jurisdiction which includes all of the essential elements of any such crime for which a sentence to a term of imprisonment in excess of one year or a sentence of death was authorized and is authorized in this state irrespective of whether such sentence was imposed.

§ 120.45 Stalking in the fourth degree

A person is guilty of stalking in the fourth degree when he or she intentionally, and for no legitimate purpose, engages in a course of conduct directed at a specific person, and knows or reasonably should know that such conduct:

- 1. is likely to cause reasonable fear of material harm to the physical health, safety or property of such person, a member of such person's immediate family or a third party with whom such person is acquainted; or
- 2. causes material harm to the mental or emotional health of such person, where such conduct consists of following, telephoning or initiating communication or contact with such person, a member of such person's immediate family or a third party with whom such person is acquainted, and the actor was previously clearly informed to cease that conduct; or
- 3. is likely to cause such person to reasonably fear that his or her employment, business or career

is threatened, where such conduct consists of appearing, telephoning or initiating communication or contact at such person's place of employment or business, and the actor was previously clearly informed to cease that conduct.

Stalking in the fourth degree is a class B misdemeanor.

§ 120.50 Stalking in the third degree

A person is guilty of stalking in the third degree when he or she:

- 1. Commits the crime of stalking in the fourth degree in violation of section 120.45 of this article against three or more persons, in three or more separate transactions, for which the actor has not been previously convicted; or
- 2. Commits the crime of stalking in the fourth degree in violation of section 120.45 of this article against any person, and has previously been convicted, within the preceding ten years of a specified predicate crime, as defined in subdivision five of section 120.40 of this article, and the victim of such specified predicate crime is the victim, or an immediate family member of the victim, of the present offense; or
- 3. With intent to harass, annoy or alarm a specific person, intentionally engages in a course of conduct directed at such person which is likely to cause such person to reasonably fear physical injury or serious physical injury, the commission of a sex offense against, or the kidnapping, unlawful imprisonment or death of such person or a member of such person's immediate family; or
- 4. Commits the crime of stalking in the fourth degree and has previously been convicted within the preceding ten years of stalking in the fourth degree.

Stalking in the third degree is a class A misdemeanor.

§ 120.55 Stalking in the second degree

A person is guilty of stalking in the second degree when he or she:

1. Commits the crime of stalking in the third degree as defined in subdivision three of section 120.50 of this article and in the course of and in furtherance of the commission of such offense: (i) displays, or possesses and threatens the use of, a firearm, pistol, revolver, rifle, shotgun, machine gun, electronic dart gun, electronic stun gun, cane

sword, billy, blackjack, bludgeon, plastic knuckles, metal knuckles, chuka stick, sand bag, sandclub, slingshot, slungshot, shirken, "Kung Fu Star", dagger, dangerous knife, dirk, razor, stiletto, imitation pistol, dangerous instrument, deadly instrument or deadly weapon; or (ii) displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm; or

- 2. Commits the crime of stalking in the third degree in violation of subdivision three of section 120.50 of this article against any person, and has previously been convicted, within the preceding five years, of a specified predicate crime as defined in subdivision five of section 120.40 of this article, and the victim of such specified predicate crime is the victim, or an immediate family member of the victim, of the present offense; or
- 3. Commits the crime of stalking in the fourth degree and has previously been convicted of stalking in the third degree as defined in subdivision four of section 120.50 of this article against any person; or
- 4. Being twenty-one years of age or older, repeatedly follows a person under the age of fourteen or engages in a course of conduct or repeatedly commits acts over a period of time intentionally placing or attempting to place such person who is under the age of fourteen in reasonable fear of physical injury, serious physical injury or death; or
- 5. Commits the crime of stalking in the third degree, as defined in subdivision three of section 120.50 of this article, against ten or more persons, in ten or more separate transactions, for which the actor has not been previously convicted.

Stalking in the second degree is a class E felony.

§ 120.60 Stalking in the first degree

A person is guilty of stalking in the first degree when he or she commits the crime of stalking in the third degree as defined in subdivision three of section 120.50 or stalking in the second degree as defined in section 120.55 of this article and, in the course and furtherance thereof, he or she:

- 1. intentionally or recklessly causes physical injury to the victim of such crime; or
- 2. commits a class A misdemeanor defined in article one hundred thirty of this chapter, or a class E felony defined in section 130.25, 130.40 or 130.85 of this chapter, or a class D felony defined

in section 130.30 or 130.45 of this chapter.

Stalking in the first degree is a class D felony.

§ 120.70 Luring a child...

Article 121. Strangulation and Related Offenses

§ 121.11 Criminal obstruction of breathing or blood circulation

A person is guilty of criminal obstruction of breathing or blood circulation when, with intent to impede the normal breathing or circulation of the blood of another person, he or she:

- a. applies pressure on the throat or neck of such person; or
 - b. blocks the nose or mouth of such person.

Criminal obstruction of breathing or blood circulation is a class A misdemeanor.

§ 121.12 Strangulation in the second degree

A person is guilty of strangulation in the second degree when he or she commits the crime of criminal obstruction of breathing or blood circulation, as defined in section 121.11 of this article, and thereby causes stupor, loss of consciousness for any period of time, or any other physical injury or impairment.

Strangulation in the second degree is a class D felony.

§ 121.13 Strangulation in the first degree

A person is guilty of strangulation in the first degree when he or she commits the crime of criminal obstruction of breathing or blood circulation, as defined in section 121.11 of this article, and thereby causes serious physical injury to such other person.

Strangulation in the first degree is a class C felony.

§ 121.14 Medical or dental purpose...

Article 125. Homicide, Abortion and Related Offenses

§ 125.00 Homicide defined

Homicide means conduct which causes the death of a person or an unborn child with which a female has been pregnant for more than twenty-four weeks under circumstances constituting murder,

manslaughter in the first degree, manslaughter in the second degree, criminally negligent homicide, abortion in the first degree or self-abortion in the first degree.

§ 125.05 Homicide, abortion and related offenses; definitions of terms

The following definitions are applicable to this article:

1. "Person," when referring to the victim of a homicide, means a human being who has been born and is alive....

§ 125.10 Criminally negligent homicide

A person is guilty of criminally negligent homicide when, with criminal negligence, he causes the death of another person.

Criminally negligent homicide is a class E felony.

§ 125.11 Aggravated criminally negligent homicide...

§ 125.12 Vehicular manslaughter in the second degree

A person is guilty of vehicular manslaughter in the second degree when he or she causes the death of another person, and either:

- (1) operates a motor vehicle in violation of subdivision two, three, four or four-a of section eleven hundred ninety-two of the vehicle and traffic law or operates a vessel or public vessel in violation of paragraph (b), (c), (d) or (e) of subdivision two of section forty-nine-a of the navigation law, and as a result of such intoxication or impairment by the use of a drug, or by the combined influence of drugs or of alcohol and any drug or drugs, operates such motor vehicle, vessel or public vessel in a manner that causes the death of such other person, or
- (2) operates a motor vehicle with a gross vehicle weight rating of more than eighteen thousand pounds which contains flammable gas, radioactive materials or explosives in violation of subdivision one of section eleven hundred ninety-two of the vehicle and traffic law, and such flammable gas, radioactive materials or explosives is the cause of such death, and as a result of such impairment by the use of alcohol, operates such motor vehicle in a manner that causes the death of such other person, or
- (3) operates a snowmobile in violation of paragraph (b), (c) or (d) of subdivision one of

section 25.24 of the parks, recreation and historic preservation law or operates an all terrain vehicle as defined in paragraph (a) of subdivision one of section twenty-two hundred eighty-one of the vehicle and traffic law in violation of subdivision two, three, four, or four-a of section eleven hundred ninety-two of the vehicle and traffic law, and as a result of such intoxication or impairment by the use of a drug, or by the combined influence of drugs or of alcohol and any drug or drugs, operates such snowmobile or all terrain vehicle in a manner that causes the death of such other person.

If it is established that the person operating such motor vehicle, vessel, public vessel, snowmobile or all terrain vehicle caused such death while unlawfully intoxicated or impaired by the use of alcohol or a drug, then there shall be a rebuttable presumption that, as a result of such intoxication or impairment by the use of alcohol or a drug, or by the combined influence of drugs or of alcohol and any drug or drugs, such person operated the motor vehicle, vessel, public vessel, snowmobile or all terrain vehicle in a manner that caused such death, as required by this section.

Vehicular manslaughter in the second degree is a class D felony.

§ 125.13 Vehicular manslaughter in the first degree

A person is guilty of vehicular manslaughter in the first degree when he or she commits the crime of vehicular manslaughter in the second degree as defined in section 125.12 of this article, and either:

- (1) commits such crime while operating a motor vehicle while such person has .18 of one per centum or more by weight of alcohol in such person's blood as shown by chemical analysis of such person's blood, breath, urine or saliva made pursuant to the provisions of section eleven hundred ninety-four of the vehicle and traffic law;
- (2) commits such crime while knowing or having reason to know that: (a) his or her license or his or her privilege of operating a motor vehicle in another state or his or her privilege of obtaining a license to operate a motor vehicle in another state is suspended or revoked and such suspension or revocation is based upon a conviction in such other state for an offense which would, if committed in this state, constitute a violation of any of the provisions of section eleven hundred ninety-two of the vehicle and traffic law; or (b) his or her license or his or her privilege of operating a motor vehicle

in the state or his or her privilege of obtaining a license issued by the commissioner of motor vehicles is suspended or revoked and such suspension or revocation is based upon either a refusal to submit to a chemical test pursuant to section eleven hundred ninety-four of the vehicle and traffic law or following a conviction for a violation of any of the provisions of section eleven hundred ninety-two of the vehicle and traffic law;

- (3) has previously been convicted of violating any of the provisions of section eleven hundred ninety-two of the vehicle and traffic law within the preceding ten years, provided that, for the purposes of this subdivision, a conviction in any other state or jurisdiction for an offense which, if committed in this state, would constitute a violation of section eleven hundred ninety-two of the vehicle and traffic law, shall be treated as a violation of such law:
- (4) causes the death of more than one other person;
- (5) has previously been convicted of violating any provision of this article or article one hundred twenty of this title involving the operation of a motor vehicle, or was convicted in any other state or jurisdiction of an offense involving the operation of a motor vehicle which, if committed in this state, would constitute a violation of this article or article one hundred twenty of this title; or
- (6) commits such crime while operating a motor vehicle while a child who is fifteen years of age or less is a passenger in such motor vehicle and causes the death of such child.

If it is established that the person operating such motor vehicle caused such death or deaths while unlawfully intoxicated or impaired by the use of alcohol or a drug, or by the combined influence of drugs or of alcohol and any drug or drugs, then there shall be a rebuttable presumption that, as a result of such intoxication or impairment by the use of alcohol or a drug, or by the combined influence of drugs or of alcohol and any drug or drugs, such person operated the motor vehicle in a manner that caused such death or deaths, as required by this section and section 125.12 of this article.

Vehicular manslaughter in the first degree is a class C felony.

§ 125.14 Aggravated vehicular homicide...

§ 125.15 Manslaughter in the second degree

A person is guilty of manslaughter in the second degree when:

- 1. He recklessly causes the death of another person; or
- 2. He commits upon a female an abortional act which causes her death, unless such abortional act is justifiable pursuant to subdivision three of section 125.05; or
- 3. He intentionally causes or aids another person to commit suicide.

Manslaughter in the second degree is a class C felony.

§ 125.20 Manslaughter in the first degree

A person is guilty of manslaughter in the first degree when:

- 1. With intent to cause serious physical injury to another person, he causes the death of such person or of a third person; or
- 2. With intent to cause the death of another person, he causes the death of such person or of a third person under circumstances which do not constitute murder because he acts under the influence of extreme emotional disturbance, as defined in paragraph (a) of subdivision one of section 125.25. The fact that homicide was committed under the influence of extreme emotional disturbance constitutes a mitigating circumstance reducing murder to manslaughter in the first degree and need not be proved in any prosecution initiated under this subdivision; or
- 3. He commits upon a female pregnant for more than twenty-four weeks an abortional act which causes her death, unless such abortional act is justifiable pursuant to subdivision three of section 125.05; or
- 4. Being eighteen years old or more and with intent to cause physical injury to a person less than eleven years old, the defendant recklessly engages in conduct which creates a grave risk of serious physical injury to such person and thereby causes the death of such person.

Manslaughter in the first degree is a class B felony.

§ 125.21 Aggravated manslaughter in the second degree...

§ 125.22 Aggravated manslaughter in the first degree....

§ 125.25 Murder in the second degree

A person is guilty of murder in the second degree when:

- 1. With intent to cause the death of another person, he causes the death of such person or of a third person; except that in any prosecution under this subdivision, it is an affirmative defense that:
- (a) The defendant acted under the influence of extreme emotional disturbance for which there was a reasonable explanation or excuse, the reasonableness of which is to be determined from the viewpoint of a person in the defendant's situation under the circumstances as the defendant believed them to be. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, manslaughter in the first degree or any other crime; or
- (b) The defendant's conduct consisted of causing or aiding, without the use of duress or deception, another person to commit suicide. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, manslaughter in the second degree or any other crime; or
- 2. Under circumstances evincing a depraved indifference to human life, he recklessly engages in conduct which creates a grave risk of death to another person, and thereby causes the death of another person; or
- 3. Acting either alone or with one or more other persons, he commits or attempts to commit robbery, burglary, kidnapping, arson, rape in the first degree, criminal sexual act in the first degree, sexual abuse in the first degree, aggravated sexual abuse, escape in the first degree, or escape in the second degree, and, in the course of and in furtherance of such crime or of immediate flight therefrom, he, or another participant, if there be any, causes the death of a person other than one of the participants; except that in any prosecution under this subdivision, in which the defendant was not the only participant in the underlying crime, it is an affirmative defense that the defendant:
- (a) Did not commit the homicidal act or in any way solicit, request, command, importune, cause or aid the commission thereof; and
- (b) Was not armed with a deadly weapon, or any instrument, article or substance readily capable of causing death or serious physical injury and of a

sort not ordinarily carried in public places by law-abiding persons; and

- (c) Had no reasonable ground to believe that any other participant was armed with such a weapon, instrument, article or substance; and
- (d) Had no reasonable ground to believe that any other participant intended to engage in conduct likely to result in death or serious physical injury; or
- 4. Under circumstances evincing a depraved indifference to human life, and being eighteen years old or more the defendant recklessly engages in conduct which creates a grave risk of serious physical injury or death to another person less than eleven years old and thereby causes the death of such person; or
- 5. Being eighteen years old or more, while in the course of committing rape in the first, second or third degree, criminal sexual act in the first, second or third degree, sexual abuse in the first degree, aggravated sexual abuse in the first, second, third or fourth degree, or incest in the first, second or third degree, against a person less than fourteen years old, he or she intentionally causes the death of such person.

Murder in the second degree is a class A-I felony.

§ 125.26 Aggravated murder

A person is guilty of aggravated murder when:

- 1. With intent to cause the death of another person, he or she causes the death of such person, or of a third person who was a person described in subparagraph (i), (ii) or (iii) of paragraph (a) of this subdivision engaged at the time of the killing in the course of performing his or her official duties: and
 - (a) Either:
- (i) the intended victim was a police officer as defined in subdivision thirty-four of section 1.20 of the criminal procedure law who was at the time of the killing engaged in the course of performing his or her official duties, and the defendant knew or reasonably should have known that the victim was a police officer; or
- (ii) the intended victim was a peace officer as defined in paragraph a of subdivision twenty-one, subdivision twenty-three, twenty-four or sixty-two (employees of the division for youth) of section 2.10 of the criminal procedure law who was at the time of the killing engaged in the course of

- performing his or her official duties, and the defendant knew or reasonably should have known that the victim was such a uniformed court officer, parole officer, probation officer, or employee of the division for youth; or
- (iii) the intended victim was an employee of a state correctional institution or was an employee of a local correctional facility as defined in subdivision two of section forty of the correction law, who was at the time of the killing engaged in the course of performing his or her official duties, and the defendant knew or reasonably should have known that the victim was an employee of a state correctional institution or a local correctional facility; and
- (b) The defendant was more than eighteen years old at the time of the commission of the crime; or
- 2. (a) With intent to cause the death of a person less than fourteen years old, he or she causes the death of such person, and the defendant acted in an especially cruel and wanton manner pursuant to a course of conduct intended to inflict and inflicting torture upon the victim prior to the victim's death. As used in this subdivision, "torture" means the intentional and depraved infliction of extreme physical pain that is separate and apart from the pain which otherwise would have been associated with such cause of death; and
- (b) The defendant was more than eighteen years old at the time of the commission of the crime.
- 3. In any prosecution under subdivision one or two of this section, it is an affirmative defense that:
- (a) The defendant acted under the influence of extreme emotional disturbance for which there was a reasonable explanation or excuse, the reasonableness of which is to be determined from the viewpoint of a person in the defendant's situation under the circumstances as the defendant believed them to be. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, aggravated manslaughter in the first degree, manslaughter in the first degree or any other crime except murder in the second degree; or
- (b) The defendant's conduct consisted of causing or aiding, without the use of duress or deception, another person to commit suicide. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, aggravated manslaughter in the

second degree, manslaughter in the second degree or any other crime except murder in the second degree.

Aggravated murder is a class A-I felony.

§ 125.27 Murder in the first degree

A person is guilty of murder in the first degree when:

- 1. With intent to cause the death of another person, he causes the death of such person or of a third person; and
 - (a) Either:
- (i) the intended victim was a police officer as defined in subdivision 34 of section 1.20 of the criminal procedure law who was at the time of the killing engaged in the course of performing his official duties, and the defendant knew or reasonably should have known that the intended victim was a police officer; or
- (ii) the intended victim was a peace officer as defined in paragraph a of subdivision twenty-one, subdivision twenty-three, twenty-four or sixty-two (employees of the division for youth) of section 2.10 of the criminal procedure law who was at the time of the killing engaged in the course of performing his official duties, and the defendant knew or reasonably should have known that the intended victim was such a uniformed court officer, parole officer, probation officer, or employee of the division for youth; or
- (iii) the intended victim was an employee of a state correctional institution or was an employee of a local correctional facility as defined in subdivision two of section forty of the correction law, who was at the time of the killing engaged in the course of performing his official duties, and the defendant knew or reasonably should have known that the intended victim was an employee of a state correctional institution or a local correctional facility; or
- (iv) at the time of the commission of the killing, the defendant was confined in a state correctional institution or was otherwise in custody upon a sentence for the term of his natural life, or upon a sentence commuted to one of natural life, or upon a sentence for an indeterminate term the minimum of which was at least fifteen years and the maximum of which was natural life, or at the time of the commission of the killing, the defendant had escaped from such confinement or custody while

- serving such a sentence and had not yet been returned to such confinement or custody; or
- (v) the intended victim was a witness to a crime committed on a prior occasion and the death was caused for the purpose of preventing the intended victim's testimony in any criminal action or proceeding whether or not such action or proceeding had been commenced, or the intended victim had previously testified in a criminal action or proceeding and the killing was committed for the purpose of exacting retribution for such prior testimony, or the intended victim was an immediate family member of a witness to a crime committed on a prior occasion and the killing was committed for the purpose of preventing or influencing the testimony of such witness, or the intended victim was an immediate family member of a witness who had previously testified in a criminal action or proceeding and the killing was committed for the purpose of exacting retribution upon such witness for such prior testimony. As used in this subparagraph "immediate family member" means a husband, wife, father, mother, daughter, son, brother, sister, stepparent, grandparent, stepchild or grandchild; or
- (vi) the defendant committed the killing or procured commission of the killing pursuant to an agreement with a person other than the intended victim to commit the same for the receipt, or in expectation of the receipt, of anything of pecuniary value from a party to the agreement or from a person other than the intended victim acting at the direction of a party to such agreement; or
- (vii) the victim was killed while the defendant was in the course of committing or attempting to commit and in furtherance of robbery, burglary in the first degree or second degree, kidnapping in the first degree, arson in the first degree or second degree, rape in the first degree, criminal sexual act in the first degree, sexual abuse in the first degree, aggravated sexual abuse in the first degree or escape in the first degree, or in the course of and furtherance of immediate flight after committing or attempting to commit any such crime or in the course of and furtherance of immediate flight after attempting to commit the crime of murder in the second degree; provided however, the victim is not a participant in one of the aforementioned crimes and, provided further that, unless the defendant's criminal liability under this subparagraph is based upon the defendant having commanded another person to cause the death of the victim or intended

victim pursuant to section 20.00 of this chapter, this subparagraph shall not apply where the defendant's criminal liability is based upon the conduct of another pursuant to section 20.00 of this chapter; or

- (viii) as part of the same criminal transaction, the defendant, with intent to cause serious physical injury to or the death of an additional person or persons, causes the death of an additional person or persons; provided, however, the victim is not a participant in the criminal transaction; or
- (ix) prior to committing the killing, the defendant had been convicted of murder as defined in this section or section 125.25 of this article, or had been convicted in another jurisdiction of an offense which, if committed in this state, would constitute a violation of either of such sections; or
- (x) the defendant acted in an especially cruel and wanton manner pursuant to a course of conduct intended to inflict and inflicting torture upon the victim prior to the victim's death. As used in this subparagraph, "torture" means the intentional and depraved infliction of extreme physical pain; "depraved" means the defendant relished the infliction of extreme physical pain upon the victim evidencing debasement or perversion or that the defendant evidenced a sense of pleasure in the infliction of extreme physical pain; or
- (xi) the defendant intentionally caused the death of two or more additional persons within the state in separate criminal transactions within a period of twenty-four months when committed in a similar fashion or pursuant to a common scheme or plan; or
- (xii) the intended victim was a judge as defined in subdivision twenty-three of section 1.20 of the criminal procedure law and the defendant killed such victim because such victim was, at the time of the killing, a judge; or
- (xiii) the victim was killed in furtherance of an act of terrorism, as defined in paragraph (b) of subdivision one of section 490.05 of this chapter; and
- (b) The defendant was more than eighteen years old at the time of the commission of the crime.
- 2. In any prosecution under subdivision one, it is an affirmative defense that:
- (a) The defendant acted under the influence of extreme emotional disturbance for which there was

- a reasonable explanation or excuse, the reasonableness of which is to be determined from the viewpoint of a person in the defendant's situation under the circumstances as the defendant believed them to be. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, manslaughter in the first degree or any other crime except murder in the second degree; or
- (b) The defendant's conduct consisted of causing or aiding, without the use of duress or deception, another person to commit suicide. Nothing contained in this paragraph shall constitute a defense to a prosecution for, or preclude a conviction of, manslaughter in the second degree or any other crime except murder in the second degree.

Murder in the first degree is a class A-I felony.

- § 125.40 Abortion in the second degree...
- § 125.45 Abortion in the first degree...
- § 125.50 Self-abortion in the second degree...
- § 125.55 Self-abortion in the first degree...
- § 125.60 Issuing abortional articles...

Article 130. Sex Offenses

§ 130.00 Sex offenses; definitions of terms

The following definitions are applicable to this article:

- 1. "Sexual intercourse" has its ordinary meaning and occurs upon any penetration, however slight.
- 2. (a) "Oral sexual conduct" means conduct between persons consisting of contact between the mouth and the penis, the mouth and the anus, or the mouth and the vulva or vagina.
- (b) "Anal sexual conduct" means conduct between persons consisting of contact between the penis and anus.
- 3. "Sexual contact" means any touching of the sexual or other intimate parts of a person for the purpose of gratifying sexual desire of either party. It includes the touching of the actor by the victim, as well as the touching of the victim by the actor, whether directly or through clothing, as well as the emission of ejaculate by the actor upon any part of the victim, clothed or unclothed.
- 4. For the purposes of this article "married" means the existence of the relationship between

the actor and the victim as spouses which is recognized by law at the time the actor commits an offense proscribed by this article against the victim.

- 5. "Mentally disabled" means that a person suffers from a mental disease or defect which renders him or her incapable of appraising the nature of his or her conduct.
- 6. "Mentally incapacitated" means that a person is rendered temporarily incapable of appraising or controlling his conduct owing to the influence of a narcotic or intoxicating substance administered to him without his consent, or to any other act committed upon him without his consent.
- 7. "Physically helpless" means that a person is unconscious or for any other reason is physically unable to communicate unwillingness to an act.
- 8. "Forcible compulsion" means to compel by either:
 - a. use of physical force; or
- b. a threat, express or implied, which places a person in fear of immediate death or physical injury to himself, herself or another person, or in fear that he, she or another person will immediately be kidnapped.
- 9. "Foreign object" means any instrument or article which, when inserted in the vagina, urethra, penis, rectum or anus, is capable of causing physical injury.
- 10. "Sexual conduct" means sexual intercourse, oral sexual conduct, anal sexual conduct, aggravated sexual contact, or sexual contact.
- 11. "Aggravated sexual contact" means inserting, other than for a valid medical purpose, a foreign object in the vagina, urethra, penis, rectum or anus of a child, thereby causing physical injury to such child.
- 12. "Health care provider" means any person who is, or is required to be, licensed or registered or holds himself or herself out to be licensed or registered, or provides services as if he or she were licensed or registered in the profession of medicine, chiropractic, dentistry or podiatry under any of the following: article one hundred thirty-one, one hundred thirty-two, one hundred thirty-three, or one hundred forty-one of the education law.
- 13. "Mental health care provider" shall mean a licensed physician, licensed psychologist, registered professional nurse, licensed clinical

social worker or a licensed master social worker under the supervision of a physician, psychologist or licensed clinical social worker.

§ 130.05 Sex offenses; lack of consent

- 1. Whether or not specifically stated, it is an element of every offense defined in this article that the sexual act was committed without consent of the victim.
 - 2. Lack of consent results from:
 - (a) Forcible compulsion; or
 - (b) Incapacity to consent; or
- (c) Where the offense charged is sexual abuse or forcible touching, any circumstances, in addition to forcible compulsion or incapacity to consent, in which the victim does not expressly or impliedly acquiesce in the actor's conduct; or
- (d) Where the offense charged is rape in the third degree as defined in subdivision three of section 130.25, or criminal sexual act in the third degree as defined in subdivision three of section 130.40, in addition to forcible compulsion, circumstances under which, at the time of the act of intercourse, oral sexual conduct or anal sexual conduct, the victim clearly expressed that he or she did not consent to engage in such act, and a reasonable person in the actor's situation would have understood such person's words and acts as an expression of lack of consent to such act under all the circumstances.
- 3. A person is deemed incapable of consent when he or she is:
 - (a) less than seventeen years old; or
 - (b) mentally disabled; or
 - (c) mentally incapacitated; or
 - (d) physically helpless; or
- (e) committed to the care and custody or supervision of the state department of corrections and community supervision or a hospital, as such term is defined in subdivision two of section four hundred of the correction law, and the actor is an employee who knows or reasonably should know that such person is committed to the care and custody or supervision of such department or hospital. For purposes of this paragraph, "employee" means (i) an employee of the state department of corrections and community supervision who, as part of his or her employment,

performs duties: (A) in a state correctional facility in which the victim is confined at the time of the offense consisting of providing custody, medical or mental health services, counseling services, educational programs, vocational training, institutional parole services or direct supervision to inmates; or

- (B) of supervising persons released on community supervision and supervises the victim at the time of the offense or has supervised the victim and the victim is still under community supervision at the time of the offense; or
- (ii) an employee of the office of mental health who, as part of his or her employment, performs duties in a state correctional facility or hospital, as such term is defined in subdivision two of section four hundred of the correction law in which the inmate is confined at the time of the offense, consisting of providing custody, medical or mental health services, or direct supervision to such inmates; or
- (iii) a person, including a volunteer, providing direct services to inmates in a state correctional facility in which the victim is confined at the time of the offense pursuant to a contractual arrangement with the state department of corrections and community supervision or, in the case of a volunteer, a written agreement with such department, provided that the person received written notice concerning the provisions of this paragraph; or
- (f) committed to the care and custody of a local correctional facility, as such term is defined in subdivision two of section forty of the correction law, and the actor is an employee, not married to such person, who knows or reasonably should know that such person is committed to the care and custody of such facility. For purposes of this paragraph, "employee" means an employee of the local correctional facility where the person is committed who performs professional duties consisting of providing custody, medical or mental health services, counseling services, educational services, or vocational training for inmates. For purposes of this paragraph, "employee" shall also mean a person, including a volunteer or a government employee of the state department of corrections and community supervision or a local health, education or probation agency, providing direct services to inmates in the local correctional facility in which the victim is confined at the time of the offense pursuant to a contractual arrangement

- with the local correctional department or, in the case of such a volunteer or government employee, a written agreement with such department, provided that such person received written notice concerning the provisions of this paragraph; or
- (g) committed to or placed with the office of children and family services and in residential care, and the actor is an employee, not married to such person, who knows or reasonably should know that such person is committed to or placed with such office of children and family services and in residential care. For purposes of this paragraph, "employee" means an employee of the office of children and family services or of a residential facility in which such person is committed to or placed at the time of the offense who, as part of his or her employment, performs duties consisting of providing custody, medical or mental health services, counseling services, educational services, vocational training, or direct supervision to persons committed to or placed in a residential facility operated by the office of children and family services; or
- (h) a client or patient and the actor is a health care provider or mental health care provider charged with rape in the third degree as defined in section 130.25, criminal sexual act in the third degree as defined in section 130.40, aggravated sexual abuse in the fourth degree as defined in section 130.65-a, or sexual abuse in the third degree as defined in section 130.55, and the act of sexual conduct occurs during a treatment session, consultation, interview, or examination.

§ 130.10 Sex offenses; limitation; defenses

- 1. In any prosecution under this article in which the victim's lack of consent is based solely upon his or her incapacity to consent because he or she was mentally disabled, mentally incapacitated or physically helpless, it is an affirmative defense that the defendant, at the time he or she engaged in the conduct constituting the offense, did not know of the facts or conditions responsible for such incapacity to consent.
- 2. Conduct performed for a valid medical or mental health care purpose shall not constitute a violation of any section of this article in which incapacity to consent is based on the circumstances set forth in paragraph (h) of subdivision three of section 130.05 of this article.
- 3. In any prosecution for the crime of rape in the third degree as defined in section 130.25, criminal

sexual act in the third degree as defined in section 130.40, aggravated sexual abuse in the fourth degree as defined in section 130.65-a, or sexual abuse in the third degree as defined in section 130.55 in which incapacity to consent is based on the circumstances set forth in paragraph (h) of subdivision three of section 130.05 of this article it shall be an affirmative defense that the client or patient consented to such conduct charged after having been expressly advised by the health care or mental health care provider that such conduct was not performed for a valid medical purpose.

4. In any prosecution under this article in which the victim's lack of consent is based solely on his or her incapacity to consent because he or she was less than seventeen years old, mentally disabled, a client or patient and the actor is a health care provider, or committed to the care and custody or supervision of the state department of corrections and community supervision or a hospital and the actor is an employee, it shall be a defense that the defendant was married to the victim as defined in subdivision four of section 130.00 of this article.

§ 130.16 Sex offenses; corroboration

A person shall not be convicted of any offense defined in this article of which lack of consent is an element but results solely from incapacity to consent because of the victim's mental defect, or mental incapacity, or an attempt to commit the same, solely on the testimony of the victim, unsupported by other evidence tending to:

- (a) Establish that an attempt was made to engage the victim in sexual intercourse, oral sexual conduct, anal sexual conduct, or sexual contact, as the case may be, at the time of the occurrence; and
- (b) Connect the defendant with the commission of the offense or attempted offense.

§ 130.20 Sexual misconduct

A person is guilty of sexual misconduct when:

- 1. He or she engages in sexual intercourse with another person without such person's consent; or
- 2. He or she engages in oral sexual conduct or anal sexual conduct with another person without such person's consent; or
- 3. He or she engages in sexual conduct with an animal or a dead human body.

Sexual misconduct is a class A misdemeanor.

§ 130.25 Rape in the third degree

A person is guilty of rape in the third degree when:

- 1. He or she engages in sexual intercourse with another person who is incapable of consent by reason of some factor other than being less than seventeen years old;
- 2. Being twenty-one years old or more, he or she engages in sexual intercourse with another person less than seventeen years old; or
- 3. He or she engages in sexual intercourse with another person without such person's consent where such lack of consent is by reason of some factor other than incapacity to consent.

Rape in the third degree is a class E felony.

§ 130.30 Rape in the second degree

A person is guilty of rape in the second degree when:

- 1. being eighteen years old or more, he or she engages in sexual intercourse with another person less than fifteen years old; or
- 2. he or she engages in sexual intercourse with another person who is incapable of consent by reason of being mentally disabled or mentally incapacitated.

It shall be an affirmative defense to the crime of rape in the second degree as defined in subdivision one of this section that the defendant was less than four years older than the victim at the time of the act.

Rape in the second degree is a class D felony.

§ 130.35 Rape in the first degree

A person is guilty of rape in the first degree when he or she engages in sexual intercourse with another person:

- 1. By forcible compulsion; or
- 2. Who is incapable of consent by reason of being physically helpless; or
 - 3. Who is less than eleven years old; or
- 4. Who is less than thirteen years old and the actor is eighteen years old or more.

Rape in the first degree is a class B felony.

§ 130.40 Criminal sexual act in the third degree

A person is guilty of criminal sexual act in the

third degree when:

- 1. He or she engages in oral sexual conduct or anal sexual conduct with a person who is incapable of consent by reason of some factor other than being less than seventeen years old;
- 2. Being twenty-one years old or more, he or she engages in oral sexual conduct or anal sexual conduct with a person less than seventeen years old; or
- 3. He or she engages in oral sexual conduct or anal sexual conduct with another person without such person's consent where such lack of consent is by reason of some factor other than incapacity to consent.

Criminal sexual act in the third degree is a class E felony.

§ 130.45 Criminal sexual act in the second degree

A person is guilty of criminal sexual act in the second degree when:

- 1. being eighteen years old or more, he or she engages in oral sexual conduct or anal sexual conduct with another person less than fifteen years old: or
- 2. he or she engages in oral sexual conduct or anal sexual conduct with another person who is incapable of consent by reason of being mentally disabled or mentally incapacitated.

It shall be an affirmative defense to the crime of criminal sexual act in the second degree as defined in subdivision one of this section that the defendant was less than four years older than the victim at the time of the act.

Criminal sexual act in the second degree is a class D felony.

§ 130.50 Criminal sexual act in the first degree

A person is guilty of criminal sexual act in the first degree when he or she engages in oral sexual conduct or anal sexual conduct with another person:

- 1. By forcible compulsion; or
- 2. Who is incapable of consent by reason of being physically helpless; or
 - 3. Who is less than eleven years old; or
- 4. Who is less than thirteen years old and the actor is eighteen years old or more.

Criminal sexual act in the first degree is a class B felony.

§ 130.52 Forcible touching

A person is guilty of forcible touching when such person intentionally, and for no legitimate purpose, forcibly touches the sexual or other intimate parts of another person for the purpose of degrading or abusing such person; or for the purpose of gratifying the actor's sexual desire.

For the purposes of this section, forcible touching includes squeezing, grabbing or pinching.

Forcible touching is a class A misdemeanor.

§ 130.53 Persistent sexual abuse

A person is guilty of persistent sexual abuse when he or she commits the crime of forcible touching, as defined in section 130.52 of this article, sexual abuse in the third degree, as defined in section 130.55 of this article, or sexual abuse in the second degree, as defined in section 130.60 of this article, and, within the previous ten year period, has been convicted two or more times, in separate criminal transactions for which sentence was imposed on separate occasions, of forcible touching, as defined in section 130.52 of this article, sexual abuse in the third degree as defined in section 130.55 of this article, sexual abuse in the second degree, as defined in section 130. 60 of this article, or any offense defined in this article, of which the commission or attempted commission thereof is a felony.

Persistent sexual abuse is a class E felony.

§ 130.55 Sexual abuse in the third degree

A person is guilty of sexual abuse in the third degree when he or she subjects another person to sexual contact without the latter's consent; except that in any prosecution under this section, it is an affirmative defense that (a) such other person's lack of consent was due solely to incapacity to consent by reason of being less than seventeen years old, and (b) such other person was more than fourteen years old, and (c) the defendant was less than five years older than such other person.

Sexual abuse in the third degree is a class B misdemeanor.

§ 130.60 Sexual abuse in the second degree

A person is guilty of sexual abuse in the second degree when he or she subjects another person to sexual contact and when such other person is:

- 1. Incapable of consent by reason of some factor other than being less than seventeen years old; or
 - 2. Less than fourteen years old.

Sexual abuse in the second degree is a class A misdemeanor.

§ 130.65 Sexual abuse in the first degree

A person is guilty of sexual abuse in the first degree when he or she subjects another person to sexual contact:

- 1. By forcible compulsion; or
- 2. When the other person is incapable of consent by reason of being physically helpless; or
- 3. When the other person is less than eleven years old; or
- 4. When the other person is less than thirteen years old and the actor is twenty-one years old or older

Sexual abuse in the first degree is a class D felony.

§ 130.65-a Aggravated sexual abuse in the fourth degree

- 1. A person is guilty of aggravated sexual abuse in the fourth degree when:
- (a) He or she inserts a foreign object in the vagina, urethra, penis, rectum or anus of another person and the other person is incapable of consent by reason of some factor other than being less than seventeen years old; or
- (b) He or she inserts a finger in the vagina, urethra, penis, rectum or anus of another person causing physical injury to such person and such person is incapable of consent by reason of some factor other than being less than seventeen years old.
- 2. Conduct performed for a valid medical purpose does not violate the provisions of this section.

Aggravated sexual abuse in the fourth degree is a class E felony.

§ 130.66 Aggravated sexual abuse in the third degree

- 1. A person is guilty of aggravated sexual abuse in the third degree when he or she inserts a foreign object in the vagina, urethra, penis, rectum or anus of another person:
 - (a) By forcible compulsion; or

- (b) When the other person is incapable of consent by reason of being physically helpless; or
- (c) When the other person is less than eleven years old.
- 2. A person is guilty of aggravated sexual abuse in the third degree when he or she inserts a foreign object in the vagina, urethra, penis, rectum or anus of another person causing physical injury to such person and such person is incapable of consent by reason of being mentally disabled or mentally incapacitated.
- 3. Conduct performed for a valid medical purpose does not violate the provisions of this section.

Aggravated sexual abuse in the third degree is a class D felony.

§ 130.67 Aggravated sexual abuse in the second degree

- 1. A person is guilty of aggravated sexual abuse in the second degree when he or she inserts a finger in the vagina, urethra, penis, rectum or anus of another person causing physical injury to such person:
 - (a) By forcible compulsion; or
- (b) When the other person is incapable of consent by reason of being physically helpless; or
- (c) When the other person is less than eleven years old.
- 2. Conduct performed for a valid medical purpose does not violate the provisions of this section.

Aggravated sexual abuse in the second degree is a class C felony.

§ 130.70 Aggravated sexual abuse in the first degree

- 1. A person is guilty of aggravated sexual abuse in the first degree when he or she inserts a foreign object in the vagina, urethra, penis, rectum or anus of another person causing physical injury to such person:
 - (a) By forcible compulsion; or
- (b) When the other person is incapable of consent by reason of being physically helpless; or
- (c) When the other person is less than eleven years old.
- 2. Conduct performed for a valid medical purpose does not violate the provisions of this section.

Aggravated sexual abuse in the first degree is a class B felony.

§ 130.75 Course of sexual conduct against a child in the first degree...

§ 130.80 Course of sexual conduct against a child in the second degree...

§ 130.85 Female genital mutilation...

§ 130.90 Facilitating a sex offense with a controlled substance

A person is guilty of facilitating a sex offense with a controlled substance when he or she:

- 1. knowingly and unlawfully possesses a controlled substance or any preparation, compound, mixture or substance that requires a prescription to obtain and administers such substance or preparation, compound, mixture or substance that requires a prescription to obtain to another person without such person's consent and with intent to commit against such person conduct constituting a felony defined in this article; and
- 2. commits or attempts to commit such conduct constituting a felony defined in this article.

Facilitating a sex offense with a controlled substance is a class D felony.

§ 130.91 Sexually motivated felony...

§ 130.92 Sentencing...

§ 130.95. Predatory sexual assault

A person is guilty of predatory sexual assault when he or she commits the crime of rape in the first degree, criminal sexual act in the first degree, aggravated sexual abuse in the first degree, or course of sexual conduct against a child in the first degree, as defined in this article, and when:

- 1. In the course of the commission of the crime or the immediate flight therefrom, he or she:
- (a) Causes serious physical injury to the victim of such crime; or
- (b) Uses or threatens the immediate use of a dangerous instrument; or
- 2. He or she has engaged in conduct constituting the crime of rape in the first degree, criminal sexual act in the first degree, aggravated sexual abuse in the first degree, or course of sexual conduct against a child in the first degree, as defined in this article, against one or more additional persons; or

3. He or she has previously been subjected to a conviction for a felony defined in this article, incest as defined in section 255.25 of this chapter or use of a child in a sexual performance as defined in section 263.05 of this chapter.

Predatory sexual assault is a class A-II felony.

§ 130.96. Predatory sexual assault against a child

A person is guilty of predatory sexual assault against a child when, being eighteen years old or more, he or she commits the crime of rape in the first degree, criminal sexual act in the first degree, aggravated sexual abuse in the first degree, or course of sexual conduct against a child in the first degree, as defined in this article, and the victim is less than thirteen years old.

Predatory sexual assault against a child is a class A-II felony.

Article 135. Kidnapping, Coercion and Related Offenses

§ 135.00 Unlawful imprisonment, kidnapping and custodial interference; definitions of terms

The following definitions are applicable to this article:

- 1. "Restrain" means to restrict a person's movements intentionally and unlawfully in such manner as to interfere substantially with his liberty by moving him from one place to another, or by confining him either in the place where the restriction commences or in a place to which he has been moved, without consent and with knowledge that the restriction is unlawful. A person is so moved or confined "without consent" when such is accomplished by (a) physical force, intimidation or deception, or (b) any means whatever, including acquiescence of the victim, if he is a child less than sixteen years old or an incompetent person and the parent, guardian or other person or institution having lawful control or custody of him has not acquiesced in the movement or confinement.
- 2. "Abduct" means to restrain a person with intent to prevent his liberation by either (a) secreting or holding him in a place where he is not likely to be found, or (b) using or threatening to use deadly physical force.
- 3. "Relative" means a parent, ancestor, brother, sister, uncle or aunt.

§ 135.05 Unlawful imprisonment in the second degree

A person is guilty of unlawful imprisonment in the second degree when he restrains another person.

Unlawful imprisonment in the second degree is a class A misdemeanor.

§ 135.10 Unlawful imprisonment in the first degree

A person is guilty of unlawful imprisonment in the first degree when he restrains another person under circumstances which expose the latter to a risk of serious physical injury.

Unlawful imprisonment in the first degree is a class E felony.

§ 135.15 Unlawful imprisonment; defense

In any prosecution for unlawful imprisonment, it is an affirmative defense that (a) the person restrained was a child less than sixteen years old, and (b) the defendant was a relative of such child, and (c) his sole purpose was to assume control of such child.

§ 135.20 Kidnapping in the second degree

A person is guilty of kidnapping in the second degree when he abducts another person.

Kidnapping in the second degree is a class B felony.

§ 135.25 Kidnapping in the first degree

A person is guilty of kidnapping in the first degree when he abducts another person and when:

- 1. His intent is to compel a third person to pay or deliver money or property as ransom, or to engage in other particular conduct, or to refrain from engaging in particular conduct; or
- 2. He restrains the person abducted for a period of more than twelve hours with intent to:
- (a) Inflict physical injury upon him or violate or abuse him sexually; or
- (b) Accomplish or advance the commission of a felony; or
 - (c) Terrorize him or a third person; or
- (d) Interfere with the performance of a governmental or political function; or
- 3. The person abducted dies during the abduction or before he is able to return or to be returned to

safety. Such death shall be presumed, in a case where such person was less than sixteen years old or an incompetent person at the time of the abduction, from evidence that his parents, guardians or other lawful custodians did not see or hear from him following the termination of the abduction and prior to trial and received no reliable information during such period persuasively indicating that he was alive. In all other cases, such death shall be presumed from evidence that a person whom the person abducted would have been extremely likely to visit or communicate with during the specified period were he alive and free to do so did not see or hear from him during such period and received no reliable information during such period persuasively indicating that he was alive.

Kidnapping in the first degree is a class A-I felony.

§ 135.30 Kidnapping; defense

In any prosecution for kidnapping, it is an affirmative defense that (a) the defendant was a relative of the person abducted, and (b) his sole purpose was to assume control of such person.

- § 135.35 Labor trafficking...
- § 135.36 Labor trafficking; accomplice...
- § 135.45 Custodial interference in the second degree...
- § 135.50 Custodial interference in the first degree....
- § 135.55 Substitution of children....
- § 135.60 Coercion in the second degree....
- § 135.65 Coercion in the first degree...
- § 135.70 Coercion; no defense...
- § 135.75 Coercion; defense....

TITLE I. OFFENSES INVOLVING DAMAGE TO AND INTRUSION UPON PROPERTY

Article 140. Burglary and Related Offenses

§ 140.00 Criminal trespass and burglary; definitions of terms

The following definitions are applicable to this article:

- 1. "Premises" includes the term "building," as defined herein, and any real property.
- 2. "Building," in addition to its ordinary meaning, includes any structure, vehicle or watercraft used for overnight lodging of persons, or used by persons for carrying on business therein, or used as an elementary or secondary school, or an inclosed motor truck, or an inclosed motor truck trailer. Where a building consists of two or more units separately secured or occupied, each unit shall be deemed both a separate building in itself and a part of the main building.
- 3. "Dwelling" means a building which is usually occupied by a person lodging therein at night.
- 4. "Night" means the period between thirty minutes after sunset and thirty minutes before sunrise.
- 5. "Enter or remain unlawfully." A person "enters or remains unlawfully" in or upon premises when he is not licensed or privileged to do so. A person who, regardless of his intent, enters or remains in or upon premises which are at the time open to the public does so with license and privilege unless he defies a lawful order not to enter or remain, personally communicated to him by the owner of such premises or other authorized person. A license or privilege to enter or remain in a building which is only partly open to the public is not a license or privilege to enter or remain in that part of the building which is not open to the public. A person who enters or remains upon unimproved and apparently unused land, which is neither fenced nor otherwise enclosed in a manner designed to exclude intruders, does so with license and privilege unless notice against trespass is personally communicated to him by the owner of such land or other authorized person, or unless such notice is given by posting in a conspicuous manner. A person who enters or remains in or about a school building without written permission from someone authorized to issue such permission or without a legitimate reason which includes a relationship involving custody of or responsibility for a pupil or student enrolled in the school or without legitimate business or a purpose relating to the operation of the school does so without license and privilege.

§ 140.05 Trespass

A person is guilty of trespass when he knowingly enters or remains unlawfully in or upon premises.

Trespass is a violation.

§ 140.10 Criminal trespass in the third degree

A person is guilty of criminal trespass in the third degree when he knowingly enters or remains unlawfully in a building or upon real property

- (a) which is fenced or otherwise enclosed in a manner designed to exclude intruders; or
- (b) where the building is utilized as an elementary or secondary school or a children's overnight camp as defined in section one thousand three hundred ninety-two of the public health law or a summer day camp as defined in section one thousand three hundred ninety-two of the public health law in violation of conspicuously posted rules or regulations governing entry and use thereof; or
- (c) located within a city with a population in excess of one million and where the building or real property is utilized as an elementary or secondary school in violation of a personally communicated request to leave the premises from a principal, custodian or other person in charge thereof; or
- (d) located outside of a city with a population in excess of one million and where the building or real property is utilized as an elementary or secondary school in violation of a personally communicated request to leave the premises from a principal, custodian, school board member or trustee, or other person in charge thereof; or
- (e) where the building is used as a public housing project in violation of conspicuously posted rules or regulations governing entry and use thereof; or
- (f) where a building is used as a public housing project in violation of a personally communicated request to leave the premises from a housing police officer or other person in charge thereof; or
- (g) where the property consists of a right-of-way or yard of a railroad or rapid transit railroad which has been designated and conspicuously posted as a no-trespass railroad zone.

§ 140.15 Criminal trespass in the second degree

A person is guilty of criminal trespass in the second degree when:

- 1. he or she knowingly enters or remains unlawfully in a dwelling; or
- 2. being a person required to maintain registration under article six-C of the correction law and

designated a level two or level three offender pursuant to subdivision six of section one hundred sixty-eight-I of the correction law, he or she enters or remains in a public or private elementary, parochial, intermediate, junior high, vocational or high school knowing that the victim of the offense for which such registration is required attends or formerly attended such school. It shall not be an offense subject to prosecution under this subdivision if: the person is a lawfully registered student at such school; the person is a lawful student participant in a school sponsored event; the person is a parent or a legal quardian of a lawfully registered student at such school and enters the school for the purpose of attending their child's or dependent's event or activity; such school is the person's designated polling place and he or she enters such school building for the limited purpose of voting; or if the person enters such school building for the limited purposes authorized by the superintendent or chief administrator of such school.

Criminal trespass in the second degree is a class A misdemeanor.

§ 140.17 Criminal trespass in the first degree

A person is guilty of criminal trespass in the first degree when he knowingly enters or remains unlawfully in a building, and when, in the course of committing such crime, he:

- 1. Possesses, or knows that another participant in the crime possesses, an explosive or a deadly weapon; or
- 2. Possesses a firearm, rifle or shotgun, as those terms are defined in section 265.00, and also possesses or has readily accessible a quantity of ammunition which is capable of being discharged from such firearm, rifle or shotgun; or
- 3. Knows that another participant in the crime possesses a firearm, rifle or shotgun under circumstances described in subdivision two.

Criminal trespass in the first degree is a class D felony.

§ 140.20 Burglary in the third degree

A person is guilty of burglary in the third degree when he knowingly enters or remains unlawfully in a building with intent to commit a crime therein.

Burglary in the third degree is a class D felony.

§ 140.25 Burglary in the second degree

A person is guilty of burglary in the second degree when he knowingly enters or remains unlawfully in a building with intent to commit a crime therein, and when:

- 1. In effecting entry or while in the building or in immediate flight therefrom, he or another participant in the crime:
- (a) Is armed with explosives or a deadly weapon; or
- (b) Causes physical injury to any person who is not a participant in the crime; or
- (c) Uses or threatens the immediate use of a dangerous instrument; or
- (d) Displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm; or
 - 2. The building is a dwelling.

Burglary in the second degree is a class C felony.

§ 140.30 Burglary in the first degree

A person is guilty of burglary in the first degree when he knowingly enters or remains unlawfully in a dwelling with intent to commit a crime therein, and when, in effecting entry or while in the dwelling or in immediate flight therefrom, he or another participant in the crime:

- 1. Is armed with explosives or a deadly weapon; or
- 2. Causes physical injury to any person who is not a participant in the crime; or
- 3. Uses or threatens the immediate use of a dangerous instrument; or
- 4. Displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm; except that in any prosecution under this subdivision, it is an affirmative defense that such pistol, revolver, rifle, shotgun, machine gun or other firearm was not a loaded weapon from which a shot, readily capable of producing death or other serious physical injury, could be discharged. Nothing contained in this subdivision shall constitute a defense to a prosecution for, or preclude a conviction of, burglary in the second degree, burglary in the third degree or any other crime.

Burglary in the first degree is a class B felony.

§ 140.35 Possession of burglar's tools

A person is guilty of possession of burglar's tools when he possesses any tool, instrument or other article adapted, designed or commonly used for committing or facilitating offenses involving forcible entry into premises, or offenses involving larceny by a physical taking, or offenses involving theft of services as defined in subdivisions four, five and six of section 165.15, under circumstances evincing an intent to use or knowledge that some person intends to use the same in the commission of an offense of such character.

Possession of burglar's tools is a class A misdemeanor.

§ 140.40 Unlawful possession of radio devices...

Article 145. Criminal Mischief and Related Offenses

§ 145.00 Criminal mischief in the fourth degree

A person is guilty of criminal mischief in the fourth degree when, having no right to do so nor any reasonable ground to believe that he or she has such right, he or she:

- 1. Intentionally damages property of another person; or
- 2. Intentionally participates in the destruction of an abandoned building as defined in section one thousand nine hundred seventy-one-a of the real property actions and proceedings law; or
- 3. Recklessly damages property of another person in an amount exceeding two hundred fifty dollars...

Criminal mischief in the fourth degree is a class A misdemeanor.

§ 145.05 Criminal mischief in the third degree

A person is guilty of criminal mischief in the third degree when, with intent to damage property of another person, and having no right to do so nor any reasonable ground to believe that he or she has such right, he or she:

1. damages the motor vehicle of another person, by breaking into such vehicle when it is locked with the intent of stealing property, and within the previous ten year period, has been convicted three or more times, in separate criminal transactions for which sentence was imposed on separate occasions, of criminal mischief in the fourth degree as defined in section 145.00, criminal mischief in

the third degree as defined in this section, criminal mischief in the second degree as defined in section 145.10, or criminal mischief in the first degree as defined in section 145.12 of this article; or

2. damages property of another person in an amount exceeding two hundred fifty dollars.

Criminal mischief in the third degree is a class E felony.

§ 145.10 Criminal mischief in the second degree

A person is guilty of criminal mischief in the second degree when with intent to damage property of another person, and having no right to do so nor any reasonable ground to believe that he has such right, he damages property of another person in an amount exceeding one thousand five hundred dollars.

Criminal mischief in the second degree is a class D felony.

§ 145.12 Criminal mischief in the first degree

A person is guilty of criminal mischief in the first degree when with intent to damage property of another person, and having no right to do so nor any reasonable ground to believe that he has such right, he damages property of another person by means of an explosive.

Criminal mischief in the first degree is a class B felony.

- § 145.13 Definitions...
- § 145.14 Criminal tampering in the third degree...
- § 145.15 Criminal tampering in the second degree...
- § 145.20 Criminal tampering in the first degree....
- § 145.22 Cemetery desecration in the second degree...
- § 145.23 Cemetery desecration in the first degree....
- § 145.25 Reckless endangerment of property....
- § 145.26 Aggravated cemetery desecration in the second degree....
- § 145.27 Aggravated cemetery desecration in the first degree....

§ 145.30 Unlawfully posting advertisements...

§ 145.35 Tampering with a consumer product; consumer product defined....

§ 145.40 Tampering with a consumer product in the second degree....

§ 145.45 Tampering with a consumer product in the first degree....

§ 145.50 Penalties for littering on railroad tracks and rights-of-way...

§ 145.60 Making graffiti

- 1. For purposes of this section, the term "graffiti" shall mean the etching, painting, covering, drawing upon or otherwise placing of a mark upon public or private property with intent to damage such property.
- 2. No person shall make graffiti of any type on any building, public or private, or any other property real or personal owned by any person, firm or corporation or any public agency or instrumentality, without the express permission of the owner or operator of said property.

Making graffiti is a class A misdemeanor.

§ 145.65 Possession of graffiti instruments

A person is guilty of possession of graffiti instruments when he possesses any tool, instrument, article, substance, solution or other compound designed or commonly used to etch, paint, cover, draw upon or otherwise place a mark upon a piece of property which that person has no permission or authority to etch, paint, cover, draw upon or otherwise mark, under circumstances evincing an intent to use same in order to damage such property.

Possession of graffiti instruments is a class B misdemeanor.

§ 145.70 Criminal possession of a taximeter accelerating device...

Article 150. Arson

§ 150.00 Arson; definitions

As used in this article, 1. "Building", in addition to its ordinary meaning, includes any structure, vehicle or watercraft used for overnight lodging of persons, or used by persons for carrying on business therein. Where a building consists of two

or more units separately secured or occupied, each unit shall not be deemed a separate building.

2. "Motor vehicle", includes every vehicle operated or driven upon a public highway which is propelled by any power other than muscular power, except (a) electrically-driven invalid chairs being operated or driven by an invalid, (b) vehicles which run only upon rails or tracks, and (c) snowmobiles as defined in article forty-seven of the vehicle and traffic law.

§ 150.01 Arson in the fifth degree

A person is guilty of arson in the fifth degree when he or she intentionally damages property of another without consent of the owner by intentionally starting a fire or causing an explosion.

Arson in the fifth degree is a class A misdemeanor.

§ 150.05 Arson in the fourth degree

- 1. A person is guilty of arson in the fourth degree when he recklessly damages a building or motor vehicle by intentionally starting a fire or causing an explosion.
- 2. In any prosecution under this section, it is an affirmative defense that no person other than the defendant had a possessory or proprietary interest in the building or motor vehicle.

Arson in the fourth degree is a class E felony.

§ 150.10 Arson in the third degree

- 1. A person is guilty of arson in the third degree when he intentionally damages a building or motor vehicle by starting a fire or causing an explosion.
- 2. In any prosecution under this section, it is an affirmative defense that (a) no person other than the defendant had a possessory or proprietary interest in the building or motor vehicle, or if other persons had such interests, all of them consented to the defendant's conduct, and (b) the defendant's sole intent was to destroy or damage the building or motor vehicle for a lawful and proper purpose, and (c) the defendant had no reasonable ground to believe that his conduct might endanger the life or safety of another person or damage another building or motor vehicle.

Arson in the third degree is a class C felony.

§ 150.15 Arson in the second degree

A person is guilty of arson in the second degree when he intentionally damages a building or motor

vehicle by starting a fire, and when (a) another person who is not a participant in the crime is present in such building or motor vehicle at the time, and (b) the defendant knows that fact or the circumstances are such as to render the presence of such a person therein a reasonable possibility.

Arson in the second degree is a class B felony.

§ 150.20 Arson in the first degree

- 1. A person is guilty of arson in the first degree when he intentionally damages a building or motor vehicle by causing an explosion or a fire and when (a) such explosion or fire is caused by an incendiary device propelled, thrown or placed inside or near such building or motor vehicle; or when such explosion or fire is caused by an explosive; or when such explosion or fire either (i) causes serious physical injury to another person other than a participant, or (ii) the explosion or fire was caused with the expectation or receipt of financial advantage or pecuniary profit by the actor; and when (b) another person who is not a participant in the crime is present in such building or motor vehicle at the time; and (c) the defendant knows that fact or the circumstances are such as to render the presence of such person therein a reasonable possibility.
- 2. As used in this section, "incendiary device" means a breakable container designed to explode or produce uncontained combustion upon impact, containing flammable liquid and having a wick or a similar device capable of being ignited.

Arson in the first degree is a class A-I felony.

TITLE J. OFFENSES INVOLVING THEFT

Article 155. Larceny

§ 155.00 Larceny; definitions of terms

The following definitions are applicable to this title:

- 1. "Property" means any money, personal property, real property, computer data, computer program, thing in action, evidence of debt or contract, or any article, substance or thing of value, including any gas, steam, water or electricity, which is provided for a charge or compensation.
- 2. "Obtain" includes, but is not limited to, the bringing about of a transfer or purported transfer of property or of a legal interest therein, whether to the obtainer or another.

- 3. "Deprive." To "deprive" another of property means (a) to withhold it or cause it to be withheld from him permanently or for so extended a period or under such circumstances that the major portion of its economic value or benefit is lost to him, or (b) to dispose of the property in such manner or under such circumstances as to render it unlikely that an owner will recover such property.
- 4. "Appropriate." To "appropriate" property of another to oneself or a third person means (a) to exercise control over it, or to aid a third person to exercise control over it, permanently or for so extended a period or under such circumstances as to acquire the major portion of its economic value or benefit, or (b) to dispose of the property for the benefit of oneself or a third person.
- 5. "Owner." When property is taken, obtained or withheld by one person from another person, an "owner" thereof means any person who has a right to possession thereof superior to that of the taker, obtainer or withholder.

A person who has obtained possession of property by theft or other illegal means shall be deemed to have a right of possession superior to that of a person who takes, obtains or withholds it from him by larcenous means.

A joint or common owner of property shall not be deemed to have a right of possession thereto superior to that of any other joint or common owner thereof.

In the absence of a specific agreement to the contrary, a person in lawful possession of property shall be deemed to have a right of possession superior to that of a person having only a security interest therein, even if legal title lies with the holder of the security interest pursuant to a conditional sale contract or other security agreement....

- 7. "Credit card" means any instrument or article defined as a credit card in section five hundred eleven of the general business law.
- 7-a. "Debit card" means any instrument or article defined as a debit card in section five hundred eleven of the general business law...
- 8. "Service" includes, but is not limited to, labor, professional service, a computer service, transportation service, the supplying of hotel accommodations, restaurant services, entertainment, the supplying of equipment for use, and the supplying of commodities of a public utility

nature such as gas, electricity, steam and water. A ticket or equivalent instrument which evidences a right to receive a service is not in itself service but constitutes property within the meaning of subdivision one.

9. "Cable television service" means any and all services provided by or through the facilities of any cable television system or closed circuit coaxial cable communications system, or any microwave or similar transmission service used in connection with any cable television system or other similar closed circuit coaxial cable communications system.

§ 155.05 Larceny; defined

- 1. A person steals property and commits larceny when, with intent to deprive another of property or to appropriate the same to himself or to a third person, he wrongfully takes, obtains or withholds such property from an owner thereof.
- 2. Larceny includes a wrongful taking, obtaining or withholding of another's property, with the intent prescribed in subdivision one of this section, committed in any of the following ways:
- (a) By conduct heretofore defined or known as common law larceny by trespassory taking, common law larceny by trick, embezzlement, or obtaining property by false pretenses;
 - (b) By acquiring lost property.

A person acquires lost property when he exercises control over property of another which he knows to have been lost or mislaid, or to have been delivered under a mistake as to the identity of the recipient or the nature or amount of the property, without taking reasonable measures to return such property to the owner;

- (c) By committing the crime of issuing a bad check, as defined in section 190.05;
 - (d) By false promise.

A person obtains property by false promise when, pursuant to a scheme to defraud, he obtains property of another by means of a representation, express or implied, that he or a third person will in the future engage in particular conduct, and when he does not intend to engage in such conduct or, as the case may be, does not believe that the third person intends to engage in such conduct.

In any prosecution for larceny based upon a false promise, the defendant's intention or belief that the promise would not be performed may not be established by or inferred from the fact alone that such promise was not performed. Such a finding may be based only upon evidence establishing that the facts and circumstances of the case are wholly consistent with guilty intent or belief and wholly inconsistent with innocent intent or belief, and excluding to a moral certainty every hypothesis except that of the defendant's intention or belief that the promise would not be performed;

(e) By extortion.

A person obtains property by extortion when he compels or induces another person to deliver such property to himself or to a third person by means of instilling in him a fear that, if the property is not so delivered, the actor or another will:

- (i) Cause physical injury to some person in the future; or
 - (ii) Cause damage to property; or
- (iii) Engage in other conduct constituting a crime; or
- (iv) Accuse some person of a crime or cause criminal charges to be instituted against him; or
- (v) Expose a secret or publicize an asserted fact, whether true or false, tending to subject some person to hatred, contempt or ridicule; or
- (vi) Cause a strike, boycott or other collective labor group action injurious to some person's business; except that such a threat shall not be deemed extortion when the property is demanded or received for the benefit of the group in whose interest the actor purports to act; or
- (vii) Testify or provide information or withhold testimony or information with respect to another's legal claim or defense; or
- (viii) Use or abuse his position as a public servant by performing some act within or related to his official duties, or by failing or refusing to perform an official duty, in such manner as to affect some person adversely; or
- (ix) Perform any other act which would not in itself materially benefit the actor but which is calculated to harm another person materially with respect to his health, safety, business, calling, career, financial condition, reputation or personal relationships.

§ 155.10 Larceny; no defense

The crimes of (a) larceny committed by means of extortion and an attempt to commit the same, and (b) bribe receiving by a labor official as defined in section 180.20, and bribe receiving as defined in section 200.05, are not mutually exclusive, and it is no defense to a prosecution for larceny committed by means of extortion or for an attempt to commit the same that, by reason of the same conduct, the defendant also committed one of such specified crimes of bribe receiving.

§ 155.15 Larceny; defenses

- 1. In any prosecution for larceny committed by trespassory taking or embezzlement, it is an affirmative defense that the property was appropriated under a claim of right made in good faith.
- 2. In any prosecution for larceny by extortion committed by instilling in the victim a fear that he or another person would be charged with a crime, it is an affirmative defense that the defendant reasonably believed the threatened charge to be true and that his sole purpose was to compel or induce the victim to take reasonable action to make good the wrong which was the subject of such threatened charge.

§ 155.20 Larceny; value of stolen property

For the purposes of this title, the value of property shall be ascertained as follows:

- 1. Except as otherwise specified in this section, value means the market value of the property at the time and place of the crime, or if such cannot be satisfactorily ascertained, the cost of replacement of the property within a reasonable time after the crime.
- 2. Whether or not they have been issued or delivered, certain written instruments, not including those having a readily ascertainable market value such as some public and corporate bonds and securities, shall be evaluated as follows:
- (a) The value of an instrument constituting an evidence of debt, such as a check, draft or promissory note, shall be deemed the amount due or collectable thereon or thereby, such figure ordinarily being the face amount of the indebtedness less any portion thereof which has been satisfied.
- (b) The value of a ticket or equivalent instrument which evidences a right to receive a transportation, entertainment or other service shall be deemed the

price stated thereon, if any; and if no price is stated thereon the value shall be deemed the price of such ticket or equivalent instrument which the issuer charges the general public.

- (c) The value of any other instrument which creates, releases, discharges or otherwise affects any valuable legal right, privilege or obligation shall be deemed the greatest amount of economic loss which the owner of the instrument might reasonably suffer by virtue of the loss of the instrument.
- 3. Where the property consists of gas, steam, water or electricity, which is provided for charge or compensation, the value shall be the value of the property stolen in any consecutive twelve-month period.
- 4. When the value of property cannot be satisfactorily ascertained pursuant to the standards set forth in subdivisions one and two of this section, its value shall be deemed to be an amount less than two hundred fifty dollars.

§ 155.25 Petit larceny

A person is guilty of petit larceny when he steals property.

Petit larceny is a class A misdemeanor.

§ 155.30 Grand Larceny in the fourth degree

A person is guilty of grand larceny in the fourth degree when he steals property and when:

- 1. The value of the property exceeds one thousand dollars; or
- 2. The property consists of a public record, writing or instrument kept, filed or deposited according to law with or in the keeping of any public office or public servant; or
- 3. The property consists of secret scientific material; or
- 4. The property consists of a credit card or debit card; or
- 5. The property, regardless of its nature and value, is taken from the person of another; or
- 6. The property, regardless of its nature and value, is obtained by extortion; or
- 7. The property consists of one or more firearms, rifles or shotguns, as such terms are defined in section 265.00 of this chapter; or
 - 8. The value of the property exceeds one

hundred dollars and the property consists of a motor vehicle, as defined in section one hundred twenty-five of the vehicle and traffic law, other than a motorcycle, as defined in section one hundred twenty-three of such law; or

- 9. The property consists of a scroll, religious vestment, a vessel, an item comprising a display of religious symbols which forms a representative expression of faith, or other miscellaneous item of property which:
- (a) has a value of at least one hundred dollars; and
- (b) is kept for or used in connection with religious worship in any building, structure or upon the curtilage of such building or structure used as a place of religious worship by a religious corporation, as incorporated under the religious corporations law or the education law.
- 10. The property consists of an access device which the person intends to use unlawfully to obtain telephone service.
- 11. The property consists of anhydrous ammonia or liquified ammonia gas and the actor intends to use, or knows another person intends to use, such anhydrous ammonia or liquified ammonia gas to manufacture methamphetamine.

Grand larceny in the fourth degree is a class E felony.

§ 155.35 Grand larceny in the third degree

A person is guilty of grand larceny in the third degree when he or she steals property and:

- 1. when the value of the property exceeds three thousand dollars, or
- 2. the property is an automated teller machine or the contents of an automated teller machine.

Grand larceny in the third degree is a class D felony.

§ 155.40 Grand larceny in the second degree

A person is guilty of grand larceny in the second degree when he steals property and when:

- 1. The value of the property exceeds fifty thousand dollars; or
- 2. The property, regardless of its nature and value, is obtained by extortion committed by instilling in the victim a fear that the actor or another person will (a) cause physical injury to

some person in the future, or (b) cause damage to property, or (c) use or abuse his position as a public servant by engaging in conduct within or related to his official duties, or by failing or refusing to perform an official duty, in such manner as to affect some person adversely.

Grand larceny in the second degree is a class C felony.

§ 155.42 Grand larceny in the first degree

A person is guilty of grand larceny in the first degree when he steals property and when the value of the property exceeds one million dollars.

Grand larceny in the first degree is a class B felony.

§ 155.43 Aggravated grand larceny of an automated teller machine

A person is guilty of aggravated grand larceny of an automated teller machine when he or she commits the crime of grand larceny in the third degree, as defined in subdivision two of section 155.35 of this article and has been previously convicted of grand larceny in the third degree within the previous five years.

Aggravated grand larceny of an automated teller machine is a class C felony.

§ 155.45 Larceny; pleading and proof

- 1. Where it is an element of the crime charged that property was taken from the person or obtained by extortion, an indictment for larceny must so specify. In all other cases, an indictment, information or complaint for larceny is sufficient if it alleges that the defendant stole property of the nature or value required for the commission of the crime charged without designating the particular way or manner in which such property was stolen or the particular theory of larceny involved.
- 2. Proof that the defendant engaged in any conduct constituting larceny as defined in section 155.05 is sufficient to support any indictment, information or complaint for larceny other than one charging larceny by extortion. An indictment charging larceny by extortion must be supported by proof establishing larceny by extortion.

Article 156. Offenses Involving Computers; Definition of Terms

§ 156.00 Offenses involving computers; definition of terms...

- § 156.05 Unauthorized use of a computer...
- § 156.10 Computer trespass....
- § 156.20 Computer tampering in the fourth degree....
- § 156.25 Computer tampering in the third degree...
- § 156.26 Computer tampering in the second degree....
- § 156.27 Computer tampering in the first degree....
- § 156.29 Unlawful duplication of computer related material in the second degree....
- § 156.30 Unlawful duplication of computer related material in the first degree....
- § 156.35 Criminal possession of computer related material....
- § 156.50 Offenses involving computers; defenses....

Article 158. Welfare Fraud

- § 158.00 Definitions; presumption; limitation....
- § 158.05 Welfare fraud in the fifth degree....
- § 158.10 Welfare fraud in the fourth degree....
- § 158.15 Welfare fraud in the third degree....
- § 158.20 Welfare fraud in the second degree....
- § 158.25 Welfare fraud in the first degree...
- § 158.30 Criminal use of a public benefit card in the second degree....
- § 158.35 Criminal use of a public benefit card in the first degree....
- § 158.40 Criminal possession of public benefit cards in the third degree....
- § 158.45 Criminal possession of public benefit cards in the second degree....
- § 158.50 Criminal possession of public benefit cards in the first degree....

Article 160. Robbery

§ 160.00 Robbery; defined

Robbery is forcible stealing. A person forcibly steals property and commits robbery when, in the course of committing a larceny, he uses or threatens the immediate use of physical force upon another person for the purpose of:

- 1. Preventing or overcoming resistance to the taking of the property or to the retention thereof immediately after the taking; or
- 2. Compelling the owner of such property or another person to deliver up the property or to engage in other conduct which aids in the commission of the larceny.

§ 160.05 Robbery in the third degree

A person is guilty of robbery in the third degree when he forcibly steals property.

Robbery in the third degree is a class D felony.

§ 160.10 Robbery in the second degree

A person is guilty of robbery in the second degree when he forcibly steals property and when:

- 1. He is aided by another person actually present; or
- 2. In the course of the commission of the crime or of immediate flight therefrom, he or another participant in the crime:
- (a) Causes physical injury to any person who is not a participant in the crime; or
- (b) Displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm; or
- 3. The property consists of a motor vehicle, as defined in section one hundred twenty-five of the vehicle and traffic law.

Robbery in the second degree is a class C felony.

§ 160.15 Robbery in the first degree

A person is guilty of robbery in the first degree when he forcibly steals property and when, in the course of the commission of the crime or of immediate flight therefrom, he or another participant in the crime:

- 1. Causes serious physical injury to any person who is not a participant in the crime; or
- 2. Is armed with a deadly weapon; or
- 3. Uses or threatens the immediate use of a dangerous instrument; or
 - 4. Displays what appears to be a pistol, revolver,

rifle, shotgun, machine gun or other firearm; except that in any prosecution under this subdivision, it is an affirmative defense that such pistol, revolver, rifle, shotgun, machine gun or other firearm was not a loaded weapon from which a shot, readily capable of producing death or other serious physical injury, could be discharged. Nothing contained in this subdivision shall constitute a defense to a prosecution for, or preclude a conviction of, robbery in the second degree, robbery in the third degree or any other crime.

Robbery in the first degree is a class B felony.

Article 165. Other Offenses Relating to

§ 165.00 Misapplication of property...

§ 165.05 Unauthorized use of a vehicle in the third degree

A person is guilty of unauthorized use of a vehicle in the third degree when:

- 1. Knowing that he does not have the consent of the owner, he takes, operates, exercises control over, rides in or otherwise uses a vehicle. A person who engages in any such conduct without the consent of the owner is presumed to know that he does not have such consent; or
- 2. Having custody of a vehicle pursuant to an agreement between himself or another and the owner thereof whereby he or another is to perform for compensation a specific service for the owner involving the maintenance, repair or use of such vehicle, he intentionally uses or operates the same, without the consent of the owner, for his own purposes in a manner constituting a gross deviation from the agreed purpose; or
- 3. Having custody of a vehicle pursuant to an agreement with the owner thereof whereby such vehicle is to be returned to the owner at a specified time, he intentionally retains or withholds possession thereof, without the consent of the owner, for so lengthy a period beyond the specified time as to render such retention or possession a gross deviation from the agreement.

For purposes of this section "a gross deviation from the agreement" shall consist of, but not be limited to, circumstances wherein a person who having had custody of a vehicle for a period of fifteen days or less pursuant to a written agreement retains possession of such vehicle for at least seven days beyond the period specified in the

agreement and continues such possession for a period of more than two days after service or refusal of attempted service of a notice in person or by certified mail at an address indicated in the agreement stating (i) the date and time at which the vehicle was to have been returned under the agreement; (ii) that the owner does not consent to the continued withholding or retaining of such vehicle and demands its return; and that continued withholding or retaining of the vehicle may constitute a class A misdemeanor punishable by a fine of up to one thousand dollars or by a sentence to a term of imprisonment for a period of up to one year or by both such fine and imprisonment.

Unauthorized use of a vehicle in the third degree is a class A misdemeanor.

§ 165.06 Unauthorized use of a vehicle in the second degree

A person is guilty of unauthorized use of a vehicle in the second degree when:

He commits the crime of unauthorized use of a vehicle in the third degree as defined in subdivision one of section 165.05 of this article and has been previously convicted of the crime of unauthorized use of a vehicle in the third degree as defined in subdivision one of section 165.05 or second degree within the preceding ten years.

Unauthorized use of a vehicle in the second degree is a class E felony.

§ 165.07 Unlawful use of secret scientific material...

§ 165.08 Unauthorized use of a vehicle in the first degree....

§ 165.09 Auto stripping in the third degree

A person is guilty of auto stripping in the third degree when:

- 1. He or she removes or intentionally destroys or defaces any part of a vehicle, other than an abandoned vehicle, as defined in subdivision one of section one thousand two hundred twenty-four of the vehicle and traffic law, without the permission of the owner; or
- 2. He or she removes or intentionally destroys or defaces any part of an abandoned vehicle, as defined in subdivision one of section one thousand two hundred twenty-four of the vehicle and traffic law, except that it is a defense to such charge that such person was authorized to do so pursuant to

law or by permission of the owner.

Auto stripping in the third degree is a class A misdemeanor.

§ 165.10 Auto stripping in the second degree

A person is guilty of auto stripping in the second degree when:

- 1. He or she commits the offense of auto stripping in the third degree and when he or she has been previously convicted within the last five years of having violated the provisions of section 165.09 or this section; or
- 2. He or she removes or intentionally destroys, defaces, disguises, or alters any part of two or more vehicles, other than abandoned vehicles, as defined in subdivision one of section one thousand two hundred twenty-four of the vehicle and traffic law, without the permission of the owner, and the value of the parts of vehicles removed, destroyed, defaced, disguised, or altered exceeds an aggregate value of one thousand dollars.

Auto stripping in the second degree is a class E felony.

§ 165.11 Auto stripping in the first degree

A person is guilty of auto stripping in the first degree when he or she removes or intentionally destroys, defaces, disguises, or alters any part of three or more vehicles, other than abandoned vehicles, as defined in subdivision one of section one thousand two hundred twenty-four of the vehicle and traffic law, without the permission of the owner, and the value of the parts of vehicles removed, destroyed, defaced, disguised, or altered exceeds an aggregate value of three thousand dollars.

Auto stripping in the first degree is a class D felony.

§ 165.15 Theft of services

A person is guilty of theft of services when:

- 1. He obtains or attempts to obtain a service, or induces or attempts to induce the supplier of a rendered service to agree to payment therefor on a credit basis, by the use of a credit card or debit card which he knows to be stolen.
- 2. With intent to avoid payment for restaurant services rendered, or for services rendered to him as a transient guest at a hotel, motel, inn, tourist cabin, rooming house or comparable

- establishment, he avoids or attempts to avoid such payment by unjustifiable failure or refusal to pay, by stealth, or by any misrepresentation of fact which he knows to be false. A person who fails or refuses to pay for such services is presumed to have intended to avoid payment therefor; or
- 3. With intent to obtain railroad, subway, bus, air, taxi or any other public transportation service without payment of the lawful charge therefor, or to avoid payment of the lawful charge for such transportation service which has been rendered to him, he obtains or attempts to obtain such service or avoids or attempts to avoid payment therefor by force, intimidation, stealth, deception or mechanical tampering, or by unjustifiable failure or refusal to pay; or
- 4. With intent to avoid payment by himself or another person of the lawful charge for any telecommunications service, including, without limitation, cable television service, or any gas, steam, sewer, water, electrical, telegraph or telephone service which is provided for a charge or compensation, he obtains or attempts to obtain such service for himself or another person or avoids or attempts to avoid payment therefor by himself or another person by means of (a) tampering or making connection with the equipment of the supplier, whether by mechanical, electrical, acoustical or other means, or (b) offering for sale or otherwise making available, to anyone other than the provider of a telecommunications service for such service provider's own use in the provision of its service, any telecommunications decoder or descrambler, a principal function of which defeats a mechanism of electronic signal encryption, jamming or individually addressed switching imposed by the provider of any such telecommunications service to restrict the delivery of such service, or (c) any misrepresentation of fact which he knows to be false, or (d) any other artifice, trick, deception, code or device. For the purposes of this subdivision the telecommunications decoder or descrambler described in paragraph (b) above or the device described in paragraph (d) above shall not include any non-decoding and non-descrambling channel frequency converter or any television receiver-type accepted by the federal communications commission. In any prosecution under this subdivision, proof that telecommunications equipment, including, without limitation, any cable television converter, descrambler, or related equipment, has been tampered with or otherwise

intentionally prevented from performing its functions of control of service delivery without the consent of the supplier of the service, or that telecommunications equipment, including, without limitation, any cable television converter, descrambler, receiver, or related equipment, has been connected to the equipment of the supplier of the service without the consent of the supplier of the service, shall be presumptive evidence that the resident to whom the service which is at the time being furnished by or through such equipment has, with intent to avoid payment by himself or another person for a prospective or already rendered service, created or caused to be created with reference to such equipment, the condition so existing. A person who tampers with such a device or equipment without the consent of the supplier of the service is presumed to do so with intent to avoid, or to enable another to avoid, payment for the service involved. In any prosecution under this subdivision, proof that any telecommunications decoder or descrambler, a principal function of which defeats a mechanism of electronic signal encryption, jamming or individually addressed switching imposed by the provider of any such telecommunications service to restrict the delivery of such service, has been offered for sale or otherwise made available by anyone other than the supplier of such service shall be presumptive evidence that the person offering such equipment for sale or otherwise making it available has, with intent to avoid payment by himself or another person of the lawful charge for such service, obtained or attempted to obtain such service for himself or another person or avoided or attempted to avoid payment therefor by himself or another person; or

5. With intent to avoid payment by himself or another person of the lawful charge for any telephone service which is provided for a charge or compensation he (a) sells, offers for sale or otherwise makes available, without consent, an existing, canceled or revoked access device; or (b) uses, without consent, an existing, canceled or revoked access device; or (c) knowingly obtains any telecommunications service with fraudulent intent by use of an unauthorized, false, or fictitious name, identification, telephone number, or access device. For purposes of this subdivision access device means any telephone calling card number, credit card number, account number, mobile identification number, electronic serial number or personal identification number that can be used to

obtain telephone service.

- 6. With intent to avoid payment by himself or another person for a prospective or already rendered service the charge or compensation for which is measured by a meter or other mechanical device, he tampers with such device or with other equipment related thereto, or in any manner attempts to prevent the meter or device from performing its measuring function, without the consent of the supplier of the service. In any prosecution under this subdivision, proof that a meter or related equipment has been tampered with or otherwise intentionally prevented from performing its measuring function without the consent of the supplier of the service shall be presumptive evidence that the person to whom the service which is at the time being furnished by or through such meter or related equipment has, with intent to avoid payment by himself or another person for a prospective or already rendered service, created or caused to be created with reference to such meter or related equipment, the condition so existing. A person who tampers with such a device or equipment without the consent of the supplier of the service is presumed to do so with intent to avoid, or to enable another to avoid, payment for the service involved; or
- 7. He knowingly accepts or receives the use and benefit of service, including gas, steam or electricity service, which should pass through a meter but has been diverted therefrom, or which has been prevented from being correctly registered by a meter provided therefor, or which has been diverted from the pipes, wires or conductors of the supplier thereof. In any prosecution under this subdivision proof that service has been intentionally diverted from passing through a meter, or has been intentionally prevented from being correctly registered by a meter provided therefor, or has been intentionally diverted from the pipes, wires or conductors of the supplier thereof, shall be presumptive evidence that the person who accepts or receives the use and benefit of such service has done so with knowledge of the condition so existing; or
- 8. With intent to obtain, without the consent of the supplier thereof, gas, electricity, water, steam or telephone service, he tampers with any equipment designed to supply or to prevent the supply of such service either to the community in general or to particular premises; or
 - 9. With intent to avoid payment of the lawful

charge for admission to any theatre or concert hall, or with intent to avoid payment of the lawful charge for admission to or use of a chair lift, gondola, rope-tow or similar mechanical device utilized in assisting skiers in transportation to a point of ski arrival or departure, he obtains or attempts to obtain such admission without payment of the lawful charge therefor.

- 10. Obtaining or having control over labor in the employ of another person, or of business, commercial or industrial equipment or facilities of another person, knowing that he is not entitled to the use thereof, and with intent to derive a commercial or other substantial benefit for himself or a third person, he uses or diverts to the use of himself or a third person such labor, equipment or facilities.
- 11. With intent to avoid payment by himself, herself, or another person of the lawful charge for use of any computer, computer service, or computer network which is provided for a charge or compensation he or she uses, causes to be used, accesses, or attempts to use or access a computer, computer service, or computer network and avoids or attempts to avoid payment therefor. In any prosecution under this subdivision proof that a person overcame or attempted to overcome any device or coding system a function of which is to prevent the unauthorized use of said computer or computer service shall be presumptive evidence of an intent to avoid payment for the computer or computer service.

Theft of services is a class A misdemeanor, provided, however, that theft of cable television service as defined by the provisions of paragraphs (a), (c) and (d) of subdivision four of this section, and having a value not in excess of one hundred dollars by a person who has not been previously convicted of theft of services under subdivision four of this section is a violation, that theft of services under subdivision nine of this section by a person who has not been previously convicted of theft of services under subdivision nine of this section is a violation and provided further, however, that theft of services of any telephone service under paragraph (a) or (b) of subdivision five of this section having a value in excess of one thousand dollars or by a person who has been previously convicted within five years of theft of services under paragraph (a) of subdivision five of this section is a class E felony.

§ 165.16 Unauthorized sale of certain transportation services....

§ 165.17 Unlawful use of credit card, debit card or public benefit card...

§ 165.20 Fraudulently obtaining a signature...

§ 165.25 Jostling

A person is guilty of jostling when, in a public place, he intentionally and unnecessarily:

- 1. Places his hand in the proximity of a person's pocket or handbag; or
- 2. Jostles or crowds another person at a time when a third person's hand is in the proximity of such person's pocket or handbag.

Jostling is a class A misdemeanor.

§ 165.30 Fraudulent accosting

- 1. A person is guilty of fraudulent accosting when he accosts a person in a public place with intent to defraud him of money or other property by means of a trick, swindle or confidence game.
- 2. A person who, either at the time he accosts another in a public place or at some subsequent time or at some other place, makes statements to him or engages in conduct with respect to him of a kind commonly made or performed in the perpetration of a known type of confidence game, is presumed to intend to defraud such person of money or other property.

Fraudulent accosting is a class A misdemeanor.

§ 165.35 Fortune telling...

§ 165.40 Criminal possession of stolen property in the fifth degree

A person is guilty of criminal possession of stolen property in the fifth degree when he knowingly possesses stolen property, with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner thereof.

Criminal possession of stolen property in the fifth degree is a class A misdemeanor.

§ 165.45 Criminal possession of stolen property in the fourth degree

A person is guilty of criminal possession of stolen property in the fourth degree when he knowingly possesses stolen property, with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner thereof, and when:

- 1. The value of the property exceeds one thousand dollars; or
- 2. The property consists of a credit card, debit card or public benefit card; or
- 3. He is a collateral loan broker or is in the business of buying, selling or otherwise dealing in property; or
- 4. The property consists of one or more firearms, rifles and shotguns, as such terms are defined in section 265.00 of this chapter; or
- 5. The value of the property exceeds one hundred dollars and the property consists of a motor vehicle, as defined in section one hundred twenty-five of the vehicle and traffic law, other than a motorcycle, as defined in section one hundred twenty-three of such law; or
- 6. The property consists of a scroll, religious vestment, vessel or other item of property having a value of at least one hundred dollars kept for or used in connection with religious worship in any building or structure used as a place of religious worship by a religious corporation, as incorporated under the religious corporations law or the education law.
- 7. The property consists of anhydrous ammonia or liquified ammonia gas and the actor intends to use, or knows another person intends to use, such anhydrous ammonia or liquified ammonia gas to manufacture methamphetamine.

Criminal possession of stolen property in the fourth degree is a class E felony.

§ 165.50 Criminal possession of stolen property in the third degree

A person is guilty of criminal possession of stolen property in the third degree when he knowingly possesses stolen property, with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner thereof, and when the value of the property exceeds three thousand dollars.

Criminal possession of stolen property in the third degree is a class D felony.

§ 165.52 Criminal possession of stolen property in the second degree

A person is guilty of criminal possession of stolen property in the second degree when he knowingly possesses stolen property, with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner thereof, and when the value of the property exceeds fifty thousand dollars.

Criminal possession of stolen property in the second degree is a class C felony.

§ 165.54 Criminal possession of stolen property in the first degree

A person is guilty of criminal possession of stolen property in the first degree when he knowingly possesses stolen property, with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner, and when the value of the property exceeds one million dollars.

Criminal possession of stolen property in the first degree is a class B felony.

§ 165.55 Criminal possession of stolen property; presumptions

- 1. A person who knowingly possesses stolen property is presumed to possess it with intent to benefit himself or a person other than an owner thereof or to impede the recovery by an owner thereof.
- 2. A collateral loan broker or a person in the business of buying, selling or otherwise dealing in property who possesses stolen property is presumed to know that such property was stolen if he obtained it without having ascertained by reasonable inquiry that the person from whom he obtained it had a legal right to possess it.
- 3. A person who possesses two or more stolen credit cards, debit cards or public benefit cards is presumed to know that such credit cards, debit cards or public benefit cards were stolen.
- 4. A person who possesses three or more tickets or equivalent instrument for air transportation service, which tickets or instruments were stolen by reason of having been obtained from the issuer or agent thereof by the use of one or more stolen or forged credit cards, is presumed to known that such tickets or instruments were stolen.

§ 165.60 Criminal possession of stolen property; no defense

In any prosecution for criminal possession of stolen property, it is no defense that:

1. The person who stole the property has not been convicted, apprehended or identified; or

- 2. The defendant stole or participated in the larceny of the property; or
- 3. The larceny of the property did not occur in this state.
- § 165.65 Criminal possession of stolen property; corroboration...
- § 165.70 Definitions...
- § 165.71 Trademark counterfeiting in the third degree...
- § 165.72 Trademark counterfeiting in the second degree....
- § 165.73 Trademark counterfeiting in the first degree...

§ 165.74 Seizure and destruction of goods bearing counterfeit trademarks....

Any goods manufactured, sold, offered for sale, distributed or produced in violation of this article may be seized by any police officer. The magistrate must, within forty-eight hours after arraignment of the defendant, determine whether probable cause exists to believe that the goods had been manufactured, sold, offered for sale, distributed or produced in violation of this article, and upon a finding that probable cause exists to believe that the goods had been manufactured, sold, offered for sale, distributed, or produced in violation of this article, the court shall authorize such articles to be retained as evidence pending the trial of the defendant. Upon conviction of the defendant, the articles in respect whereof the defendant stands convicted shall be destroyed. Destruction shall not include auction, sale or distribution of the items in their original form.

Title K. Offenses Involving Fraud

Article 170. Forgery and Related Offenses

- § 170.00 Forgery; definitions of terms...
- § 170.05 Forgery in the third degree...
- § 170.10 Forgery in the second degree...
- § 170.15 Forgery in the first degree...
- § 170.20 Criminal possession of a forged instrument in the third degree...
- § 170.25 Criminal possession of a forged instrument in the second degree...

- § 170.27 Criminal possession of a forged instrument in the second degree; presumption...
- § 170.30 Criminal possession of a forged instrument in the first degree...
- § 170.35 Criminal possession of a forged instrument; no defense...
- § 170.40 Criminal possession of forgery devices....
- § 170.45 Criminal simulation....
- § 170.47 Criminal possession of an anti-security item...
- § 170.50 Unlawfully using slugs; definitions of terms....
- § 170.55 Unlawfully using slugs in the second degree...
- § 170.60 Unlawfully using slugs in the first degree...
- § 170.65 Forgery of a vehicle identification number...
- § 170.70 Illegal possession of a vehicle identification number...
- § 170.71 Illegal possession of a vehicle identification number; presumptions...
- § 170.75 Fraudulent making of an electronic access device in the second degree...

Article 175. Offenses Involving False Written Statements

- § 175.00 Definitions of terms...
- § 175.05 Falsifying business records in the second degree
- § 175.10 Falsifying business records in the first degree...
- § 175.15 Falsifying business records; defense...
- § 175.20 Tampering with public records in the second degree...
- § 175.25 Tampering with public records in the first degree...
- § 175.30 Offering a false instrument for filing in the second degree....

- § 175.35 Offering a false instrument for filing in the first degree...
- § 175.40 Issuing a false certificate....
- § 175.45 Issuing a false financial statement....

Article 176. Insurance Fraud

- § 176.00 Insurance fraud; definition of terms....
- § 176.05 Insurance fraud; defined...
- § 176.10 Insurance fraud in the fifth degree.....
- § 176.15 Insurance fraud in the fourth degree...
- § 176.20 Insurance fraud in the third degree
- § 176.25 Insurance fraud in the second degree....
- § 176.30 Insurance fraud in the first degree...
- § 176.35 Aggravated insurance fraud...
- § 176.40 Fraudulent life settlement act; defined...
- § 176.45 Life settlement fraud in the fifth degree...
- § 176.50 Life settlement fraud in the fourth degree...
- § 176.55 Life settlement fraud in the third degree...
- § 176.60 Life settlement fraud in the second degree...
- § 176.65 Life settlement fraud in the first degree...
- § 176.70 Aggravated life settlement fraud...

Article 177. Health Care Fraud

- § 177.00 Definitions...
- § 177.05 Health care fraud in the fifth degree....
- § 177.10 Health care fraud in the fourth degree....
- § 177.15 Health care fraud in the third degree....
- § 177.20 Health care fraud in the second degree....
- § 177.25 Health care fraud in the first degree

§ 177.30 Health care fraud; affirmative defense....

Article 178. Criminal Diversion of Prescription Medications and Prescriptions

- § 178.00 Criminal diversion of prescription medications and prescriptions; definitions....
- § 178.05 Criminal diversion of prescription medications and prescriptions; limitation...
- § 178.10 Criminal diversion of prescription medications and prescriptions in the fourth degree....
- § 178.15 Criminal diversion of prescription medications and prescriptions in the third degree....
- § 178.20 Criminal diversion of prescription medications and prescriptions in the second degree....
- § 178.25 Criminal diversion of prescription medications and prescriptions in the first degree....

Article 180. Bribery Not Involving Public Servants, and Related Offenses

§ 180.00 Commercial bribing in the second degree...

Article 180. Bribery Not Involving Public Servants, and Related Offenses

- § 180.03 Commercial bribing in the first degree...
- § 180.05 Commercial bribe receiving in the second degree....
- § 180.08 Commercial bribe receiving in the first degree....
- § 180.10 Bribery of labor official; definition of term...
- § 180.15 Bribing a labor official....
- § 180.20 Bribing a labor official; defense....
- § 180.25 Bribe receiving by a labor official....
- § 180.30 Bribe receiving by a labor official; no defense

- § 180.35 Sports bribery and tampering; definitions of terms
- § 180.40 Sports bribing
- § 180.45 Sports bribe receiving
- § 180.50 Tampering with a sports contest in the second degree
- § 180.51 Tampering with a sports contest in the first degree
- § 180.52 Impairing the integrity of a pari-mutuel betting system in the second degree
- § 180.53 Impairing the integrity of a pari-mutuel betting system in the first degree
- § 180.54 Rent gouging; definition of term
- § 180.55 Rent gouging in the third degree
- § 180.56 Rent gouging in the second degree
- § 180.57 Rent gouging in the first degree

Article 185. Frauds on Creditors

- § 185.00 Fraud in insolvency
- § 185.05 Fraud involving a security interest
- § 185.10 Fraudulent disposition of mortgaged property
- § 185.15 Fraudulent disposition of property subject to a conditional sale contract

Article 187. Residential Mortgage Fraud

- § 187.00 Definitions
- § 187.01 Limitation on prosecution
- § 187.05 Residential mortgage fraud in the fifth degree
- § 187.10 Residential mortgage fraud in the fourth degree
- § 187.15 Residential mortgage fraud in the third degree
- § 187.20 Residential mortgage fraud in the second degree
- § 187.25 Residential mortgage fraud in the first degree

Article 190. Other Frauds

- § 190.00 Issuing a bad check; definitions of terms
- § 190.05 Issuing a bad check
- § 190.10 Issuing a bad check; presumptions
- § 190.15 Issuing a bad check; defenses
- § 190.20 False advertising
- § 190.23 False personation
- § 190.25 Criminal impersonation in the second degree
- § 190.26 Criminal impersonation in the first degree
- § 190.27 Criminal sale of a police uniform
- § 190.30 Unlawfully concealing a will
- § 190.35 Misconduct by corporate official
- § 190.40 Criminal usury in the second degree
- § 190.42 Criminal usury in the first degree
- § 190.45 Possession of usurious loan records
- § 190.50 Unlawful collection practices
- § 190.55 Making a false statement of credit terms
- § 190.60 Scheme to defraud in the second degree
- § 190.65 Scheme to defraud in the first degree
- § 190.70 Scheme to defraud the state by unlawfully selling prescriptions
- § 190.72 Unauthorized radio transmission
- § 190.75 Criminal use of an access device in the second degree
- § 190.76 Criminal use of an access device in the first degree
- § 190.77 Offenses involving theft of identity; definitions
- § 190.78 Identity theft in the third degree
- § 190.79 Identity theft in the second degree
- § 190.80 Identity theft in the first degree
- § 190.80-a Aggravated identity theft
- § 190.81 Unlawful possession of personal identification information in the third degree

- § 190.82 Unlawful possession of personal identification information in the second degree
- § 190.83 Unlawful possession of personal identification information in the first degree
- § 190.84 Defenses
- § 190.85 Unlawful possession of a skimmer device in the second degree
- § 190.86 Unlawful possession of a skimmer device in the first degree

Title L. Offenses Against Public Administration

Article 195. Official Misconduct and Obstruction of Public Servants Generally

- § 195.00 Official misconduct
- § 195.05 Obstructing governmental administration in the second degree
- § 195.06 Killing or injuring a police animal
- § 195.07 Obstructing governmental administration in the first degree
- § 195.08 Obstructing governmental administration by means of a self-defense spray device
- § 195.10 Refusing to aid a peace or a police officer
- § 195.11 Harming an animal trained to aid a person with a disability in the second degree
- § 195.12 Harming an animal trained to aid a person with a disability in the first degree
- § 195.15 Obstructing firefighting operations
- § 195.16 Obstructing emergency medical services
- § 195.17 Obstruction of governmental duties by means of a bomb, destructive device, explosive, or hazardous substance
- § 195.20 Defrauding the government

Article 200. Bribery Involving Public Servants and Related Offenses

§ 200.00 Bribery in the third degree

- § 200.03 Bribery in the second degree
- § 200.04 Bribery in the first degree
- § 200.05 Bribery; defense
- § 200.10 Bribe receiving in the third degree
- § 200.11 Bribe receiving in the second degree
- § 200.12 Bribe receiving in the first degree
- § 200.15 Bribe receiving; no defense
- § 200.20 Rewarding official misconduct in the second degree
- § 200.22 Rewarding official misconduct in the first degree
- § 200.25 Receiving reward for official misconduct in the second degree
- § 200.27 Receiving reward for official misconduct in the first degree
- § 200.30 Giving unlawful gratuities
- § 200.35 Receiving unlawful gratuities
- § 200.40 Bribe giving and bribe receiving for public office; definition of term
- § 200.45 Bribe giving for public office
- § 200.50 Bribe receiving for public office
- § 200.55 Impairing the integrity of a government licensing examination

Article 205. Escape and Other Offenses Relating to Custody

- § 205.00 Escape and other offenses relating to custody; definitions of terms
- § 205.05 Escape in the third degree
- § 205.10 Escape in the second degree
- § 205.15 Escape in the first degree
- § 205.16 Absconding from temporary release in the second degree
- § 205.17 Absconding from temporary release in the first degree
- § 205.18 Absconding from a furlough program
- § 205.19 Absconding from a community treatment facility

- § 205.20 Promoting prison contraband in the second degree
- § 205.25 Promoting prison contraband in the first degree
- § 205.30 Resisting arrest
- § 205.50 Hindering prosecution; definition of term
- § 205.55 Hindering prosecution in the third degree
- § 205.60 Hindering prosecution in the second degree
- § 205.65 Hindering prosecution in the first degree

Article 210. Perjury and Related Offenses

- § 210.00 Perjury and related offenses; definitions of terms
- § 210.05 Perjury in the third degree
- § 210.10 Perjury in the second degree
- § 210.15 Perjury in the first degree
- § 210.20 Perjury; pleading and proof where inconsistent statements involved
- § 210.25 Perjury; defense
- § 210.30 Perjury; no defense
- § 210.35 Making an apparently sworn false statement in the second degree
- § 210.40 Making an apparently sworn false statement in the first degree
- § 210.45 Making a punishable false written statement
- § 210.50 Perjury and related offenses; requirement of corroboration

Article 215. Other Offenses Relating to Judicial and Other Proceedings

- § 215.00 Bribing a witness
- § 215.05 Bribe receiving by a witness
- § 215.10 Tampering with a witness in the fourth degree

- § 215.11 Tampering with a witness in the third degree
- § 215.12 Tampering with a witness in the second degree
- § 215.13 Tampering with a witness in the first degree
- § 215.14 Employer unlawfully penalizing witness or victim
- § 215.15 Intimidating a victim or witness in the third degree
- § 215.16 Intimidating a victim or witness in the second degree
- § 215.17 Intimidating a victim or witness in the first degree
- § 215.19 Bribing a juror
- § 215.20 Bribe receiving by a juror
- § 215.22 Providing a juror with a gratuity
- § 215.23 Tampering with a juror in the second degree
- § 215.25 Tampering with a juror in the first degree
- § 215.28 Misconduct by a juror in the second degree
- § 215.30 Misconduct by a juror in the first degree
- § 215.35 Tampering with physical evidence; definitions of terms
- § 215.40 Tampering with physical evidence
- § 215.45 Compounding a crime
- § 215.50 Criminal contempt in the second degree
- § 215.51 Criminal contempt in the first degree
- § 215.52 Aggravated criminal contempt
- § 215.54 Criminal contempt; prosecution and punishment
- § 215.55 Bail jumping in the third degree
- § 215.56 Bail jumping in the second degree
- § 215.57 Bail jumping in the first degree
- § 215.58 Failing to respond to an appearance ticket

§ 215.59 Bail jumping and failing to respond to an appearance ticket; defense

§ 215.60 Criminal contempt of the legislature

Article 215. Other Offenses Relating to Judicial and Other Proceedings

§ 215.65 Criminal contempt of a temporary state commission

§ 215.66 Criminal contempt of the state commission on judicial conduct

§ 215.70 Unlawful grand jury disclosure

§ 215.75 Unlawful disclosure of an indictment

§ 215.80 Unlawful disposition of assets subject to forfeiture

TITLE M. OFFENSES AGAINST PUBLIC HEALTH AND MORALS

Article 220. Controlled Substances Offenses

§ 220.00 Controlled substances; definitions

- 1. "Sell" means to sell, exchange, give or dispose of to another, or to offer or agree to do the same.
- 2. "Unlawfully" means in violation of article thirty-three of the public health law.
- 3. "Ounce" means an avoirdupois ounce as applied to solids or semisolids, and a fluid ounce as applied to liquids.
 - 4. "Pound" means an avoirdupois pound.
- 5. "Controlled substance" means any substance listed in schedule I, II, III, IV or V of section thirty-three hundred six of the public health law other than marihuana, but including concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of such law.
- 6. "Marihuana" means "marihuana" or "concentrated cannabis" as those terms are defined in section thirty-three hundred two of the public health law.
- 7. "Narcotic drug" means any controlled substance listed in schedule I(b), I(c), II(b) or II(c) other than methadone.
- 8. [Eff. until Feb. 23, 2013. See, also, subd. 8 below.] "Narcotic preparation" means any controlled substance listed in schedule III(d) or

III(e).

- 8. [Eff. Feb. 23, 2013. See, also, subd. 8 above.] "Narcotic preparation" means any controlled substance listed in schedule II(b-1), III(d) or III(e).
- 9. "Hallucinogen" means any controlled substance listed in schedule I(d) (5), (18), (19), (20), (21) and (22).
- 10. "Hallucinogenic substance" means any controlled substance listed in schedule I(d) other than concentrated cannabis, lysergic acid diethylamide, or an hallucinogen.
- 11. "Stimulant" means any controlled substance listed in schedule I(f), II(d).
- 12. "Dangerous depressant" means any controlled substance listed in schedule I(e)(2), (3), II(e), III(c)(3) or IV(c)(2), (31), (32), (40).
- 13. "Depressant" means any controlled substance listed in schedule IV(c) except (c)(2), (31), (32), (40).
- 14. "School grounds" means (a) in or on or within any building, structure, athletic playing field, playground or land contained within the real property boundary line of a public or private elementary, parochial, intermediate, junior high, vocational, or high school, or (b) any area accessible to the public located within one thousand feet of the real property boundary line comprising any such school or any parked automobile or other parked vehicle located within one thousand feet of the real property boundary line comprising any such school. For the purposes of this section an "area accessible to the public" shall mean sidewalks, streets, parking lots, parks, playgrounds, stores and restaurants.
- 15. "Prescription for a controlled substance" means a direction or authorization, by means of an official New York state prescription form, a written prescription form or an oral prescription, which will permit a person to lawfully obtain a controlled substance from any person authorized to dispense controlled substances.
- 16. For the purposes of sections 220.70, 220.71, 220.72, 220.73, 220.74, 220.75 and 220.76 of this article:
- (a) "Precursor" means ephedrine, pseudoephedrine, or any salt, isomer or salt of an isomer of such substances.
 - (b) "Chemical reagent" means a chemical

reagent that can be used in the manufacture, production or preparation of methamphetamine.

- (c) "Solvent" means a solvent that can be used in the manufacture, production or preparation of methamphetamine.
- (d) "Laboratory equipment" means any items, components or materials that can be used in the manufacture, preparation or production of methamphetamine.
- (e) "Hazardous or dangerous material" means any substance, or combination of substances, that results from or is used in the manufacture, preparation or production of methamphetamine which, because of its quantity, concentration, or physical or chemical characteristics, poses a substantial risk to human health or safety, or a substantial danger to the environment.
- 17. "School bus" means every motor vehicle owned by a public or governmental agency or private school and operated for the transportation of pupils, teachers and other persons acting in a supervisory capacity, to or from school or school activities or privately owned and operated for compensation for the transportation of pupils, children of pupils, teachers and other persons acting in a supervisory capacity to or from school or school activities.
- 18. "Controlled substance organization" means four or more persons sharing a common purpose to engage in conduct that constitutes or advances the commission of a felony under this article.
- 19. "Director" means a person who is the principal administrator, organizer, or leader of a controlled substance organization or one of several principal administrators, organizers, or leaders of a controlled substance organization.
- 20. "Profiteer" means a person who: (a) is a director of a controlled substance organization; (b) is a member of a controlled substance organization and has managerial responsibility over one or more other members of that organization; or (c) arranges, devises or plans one or more transactions constituting a felony under this article so as to obtain profits or expected profits. A person is not a profiteer if he or she is acting only as an employee; or if he or she is acting as an accommodation to a friend or relative; or if he or she is acting only under the direction and control of others and exercises no substantial, independent role in arranging or directing the transactions in

question.

§ 220.03 Criminal possession of a controlled substance in the seventh degree

A person is guilty of criminal possession of a controlled substance in the seventh degree when he or she knowingly and unlawfully possesses a controlled substance; provided, however, that it shall not be a violation of this section when a person possesses a residual amount of a controlled substance and that residual amount is in or on a hypodermic syringe or hypodermic needle obtained and possessed pursuant to section thirty-three hundred eighty-one of the public health law; nor shall it be a violation of this section when a person's unlawful possession of a controlled substance is discovered as a result of seeking immediate health care as defined in paragraph (b) of subdivision three of section 220.78 of the penal law, for either another person or him or herself because such person is experiencing a drug or alcohol overdose or other life threatening medical emergency as defined in paragraph (a) of subdivision three of section 220.78 of the penal

Criminal possession of a controlled substance in the seventh degree is a class A misdemeanor.

§ 220.06 Criminal possession of a controlled substance in the fifth degree

A person is guilty of criminal possession of a controlled substance in the fifth degree when he knowingly and unlawfully possesses:

- 1. a controlled substance with intent to sell it; or
- 2. one or more preparations, compounds, mixtures or substances containing a narcotic preparation and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 3. phencyclidine and said phencyclidine weighs fifty milligrams or more; or
- 4. one or more preparations, compounds, mixtures or substances containing concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law and said preparations, compounds, mixtures or substances are of an aggregate weight of one-fourth ounce or more; or
- 5. cocaine and said cocaine weighs five hundred milligrams or more.

- 6. ketamine and said ketamine weighs more than one thousand milligrams; or
- 7. ketamine and has previously been convicted of possession or the attempt to commit possession of ketamine in any amount; or
- 8. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of twenty-eight grams or more.

Criminal possession of a controlled substance in the fifth degree is a class D felony.

§ 220.09 Criminal possession of a controlled substance in the fourth degree

A person is guilty of criminal possession of a controlled substance in the fourth degree when he knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 2. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 3. one or more preparations, compounds, mixtures or substances containing a narcotic preparation and said preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or
- 4. a stimulant and said stimulant weighs one gram or more; or
- 5. lysergic acid diethylamide and said lysergic acid diethylamide weighs one milligram or more; or
- 6. a hallucinogen and said hallucinogen weighs twenty-five milligrams or more; or
- 7. a hallucinogenic substance and said hallucinogenic substance weighs one gram or more; or
- 8. a dangerous depressant and such dangerous depressant weighs ten ounces or more; or

- 9. a depressant and such depressant weighs two pounds or more; or
- 10. one or more preparations, compounds, mixtures or substances containing concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law and said preparations, compounds, mixtures or substances are of an aggregate weight of one ounce or more; or
- 11. phencyclidine and said phencyclidine weighs two hundred fifty milligrams or more; or
- 12. methadone and said methadone weighs three hundred sixty milligrams or more; or
- 13. phencyclidine and said phencyclidine weighs fifty milligrams or more with intent to sell it and has previously been convicted of an offense defined in this article or the attempt or conspiracy to commit any such offense; or
- 14. ketamine and said ketamine weighs four thousand milligrams or more; or
- 15. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of two hundred grams or more.

Criminal possession of a controlled substance in the fourth degree is a class C felony.

§ 220.16 Criminal possession of a controlled substance in the third degree

A person is guilty of criminal possession of a controlled substance in the third degree when he knowingly and unlawfully possesses:

- 1. a narcotic drug with intent to sell it; or
- 2. a stimulant, hallucinogen, hallucinogenic substance, or lysergic acid diethylamide, with intent to sell it and has previously been convicted of an offense defined in article two hundred twenty or the attempt or conspiracy to commit any such offense; or
- 3. a stimulant with intent to sell it and said stimulant weighs one gram or more; or
- 4. lysergic acid diethylamide with intent to sell it and said lysergic acid diethylamide weighs one milligram or more; or

- 5. a hallucinogen with intent to sell it and said hallucinogen weighs twenty-five milligrams or more: or
- 6. a hallucinogenic substance with intent to sell it and said hallucinogenic substance weighs one gram or more; or
- 7. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers with intent to sell it and said preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 8. a stimulant and said stimulant weighs five grams or more; or
- 9. lysergic acid diethylamide and said lysergic acid diethylamide weighs five milligrams or more; or
- 10. a hallucinogen and said hallucinogen weighs one hundred twenty-five milligrams or more; or
- 11. a hallucinogenic substance and said hallucinogenic substance weighs five grams or more; or
- 12. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 13. phencyclidine and said phencyclidine weighs one thousand two hundred fifty milligrams or more.

Criminal possession of a controlled substance in the third degree is a class B felony.

§ 220.18 Criminal possession of a controlled substance in the second degree

A person is guilty of criminal possession of a controlled substance in the second degree when he or she knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of four ounces or more; or
- 2. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and said preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or

- 3. a stimulant and said stimulant weighs ten grams or more; or
- 4. lysergic acid diethylamide and said lysergic acid diethylamide weighs twenty-five milligrams or more; or
- 5. a hallucinogen and said hallucinogen weighs six hundred twenty-five milligrams or more; or
- 6. a hallucinogenic substance and said hallucinogenic substance weighs twenty-five grams or more; or
- 7. methadone and said methadone weighs two thousand eight hundred eighty milligrams or more.

Criminal possession of a controlled substance in the second degree is a class A-II felony.

§ 220.21 Criminal possession of a controlled substance in the first degree

A person is guilty of criminal possession of a controlled substance in the first degree when he or she knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of eight ounces or more; or
- 2. methadone and said methadone weighs five thousand seven hundred sixty milligrams or more.

Criminal possession of a controlled substance in the first degree is a class A-I felony.

§ 220.25 Criminal possession of a controlled substance; presumption

- 1. The presence of a controlled substance in an automobile, other than a public omnibus, is presumptive evidence of knowing possession thereof by each and every person in the automobile at the time such controlled substance was found; except that such presumption does not apply (a) to a duly licensed operator of an automobile who is at the time operating it for hire in the lawful and proper pursuit of his trade, or (b) to any person in the automobile if one of them, having obtained the controlled substance and not being under duress, is authorized to possess it and such controlled substance is in the same container as when he received possession thereof, or (c) when the controlled substance is concealed upon the person of one of the occupants.
 - 2. The presence of a narcotic drug, narcotic

preparation, marihuana or phencyclidine in open view in a room, other than a public place, under circumstances evincing an intent to unlawfully mix, compound, package or otherwise prepare for sale such controlled substance is presumptive evidence of knowing possession thereof by each and every person in close proximity to such controlled substance at the time such controlled substance was found; except that such presumption does not apply to any such persons if (a) one of them, having obtained such controlled substance and not being under duress, is authorized to possess it and such controlled substance is in the same container as when he received possession thereof, or (b) one of them has such controlled substance upon his person.

§ 220.28. Use of a child to commit a controlled substance offense

- 1. A person is guilty of use of a child to commit a controlled substance offense when, being eighteen years old or more, he or she commits a felony sale or felony attempted sale of a controlled substance in violation of this article and, as part of that criminal transaction, knowingly uses a child to effectuate such felony sale or felony attempted sale of such controlled substance.
- 2. For purposes of this section, "uses a child to effectuate the felony sale or felony attempted sale of such controlled substance" means conduct by which the actor: (a) conceals such controlled substance on or about the body or person of such child for the purpose of effectuating the criminal sale or attempted sale of such controlled substance to a third person; or (b) directs, forces or otherwise requires such child to sell or attempt to sell or offer direct assistance to the defendant in selling or attempting to sell such controlled substance to a third person.

For purposes of this section, "child" means a person less than sixteen years of age.

Use of a child to commit a controlled substance offense is a class E felony.

§ 220.31 Criminal sale of a controlled substance in the fifth degree

A person is guilty of criminal sale of a controlled substance in the fifth degree when he knowingly and unlawfully sells a controlled substance.

Criminal sale of a controlled substance in the fifth degree is a class D felony.

§ 220.34 Criminal sale of a controlled substance in the fourth degree

A person is guilty of criminal sale of a controlled substance in the fourth degree when he knowingly and unlawfully sells:

- 1. a narcotic preparation; or
- 2. a dangerous depressant or a depressant and the dangerous depressant weighs ten ounces or more, or the depressant weighs two pounds or more: or
- 3. concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law; or
- 4. phencyclidine and the phencyclidine weighs fifty milligrams or more; or
 - 5. methadone; or
- 6. any amount of phencyclidine and has previously been convicted of an offense defined in this article or the attempt or conspiracy to commit any such offense; or
- 6-a. ketamine and said ketamine weighs four thousand milligrams or more.
- 7. a controlled substance in violation of section 220.31 of this article, when such sale takes place upon school grounds or on a school bus; or
- 8. a controlled substance in violation of section 220.31 of this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds. As used in this subdivision, the phrase "the grounds of a child day care or educational facility" shall have the same meaning as provided for in subdivision five of section 220.44 of this article. For the purposes of this subdivision, a rebuttable presumption shall be established that a person has knowledge that they are within the grounds of a child day care or educational facility when notice is conspicuously posted of the presence or proximity of such facility; or
- 9. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of twenty-eight grams or more.

Criminal sale of a controlled substance in the fourth degree is a class C felony.

§ 220.39 Criminal sale of a controlled substance in the third degree

A person is guilty of criminal sale of a controlled substance in the third degree when he knowingly and unlawfully sells:

- 1. a narcotic drug; or
- 2. a stimulant, hallucinogen, hallucinogenic substance, or lysergic acid diethylamide and has previously been convicted of an offense defined in article two hundred twenty or the attempt or conspiracy to commit any such offense; or
- 3. a stimulant and the stimulant weighs one gram or more; or
- 4. lysergic acid diethylamide and the lysergic acid diethylamide weighs one milligram or more; or
- 5. a hallucinogen and the hallucinogen weighs twenty-five milligrams or more; or
- 6. a hallucinogenic substance and the hallucinogenic substance weighs one gram or more; or
- 7. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and the preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 8. phencyclidine and the phencyclidine weighs two hundred fifty milligrams or more; or
- 9. a narcotic preparation to a person less than twenty-one years old.

Criminal sale of a controlled substance in the third degree is a class B felony.

§ 220.41 Criminal sale of a controlled substance in the second degree

A person is guilty of criminal sale of a controlled substance in the second degree when he knowingly and unlawfully sells:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and the preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 2. one or more preparations, compounds, mixtures or substances containing

methamphetamine, its salts, isomers or salts of isomers and the preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or

- 3. a stimulant and the stimulant weighs five grams or more; or
- 4. lysergic acid diethylamide and the lysergic acid diethylamide weighs five milligrams or more; or
- 5. a hallucinogen and the hallucinogen weighs one hundred twenty-five milligrams or more; or
- 6. a hallucinogenic substance and the hallucinogenic substance weighs five grams or more; or
- 7. methadone and the methadone weighs three hundred sixty milligrams or more.

Criminal sale of a controlled substance in the second degree is a class A-II felony.

§ 220.43 Criminal sale of a controlled substance in the first degree

A person is guilty of criminal sale of a controlled substance in the first degree when he knowingly and unlawfully sells:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and the preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or
- 2. methadone and the methadone weighs two thousand eight hundred eighty milligrams or more.

Criminal sale of a controlled substance in the first degree is a class A-I felony.

§ 220.44 Criminal sale of a controlled substance in or near school grounds

A person is guilty of criminal sale of a controlled substance in or near school grounds when he knowingly and unlawfully sells:

- 1. a controlled substance in violation of any one of subdivisions one through six-a of section 220.34 of this article, when such sale takes place upon school grounds or on a school bus; or
- 2. a controlled substance in violation of any one of subdivisions one through eight of section 220.39 of this article, when such sale takes place upon school grounds or on a school bus; or
- 3. a controlled substance in violation of any one of subdivisions one through six of section 220.34 of

this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds; or

- 4. a controlled substance in violation of any one of subdivisions one through eight of section 220.39 of this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds.
- 5. For purposes of subdivisions three and four of this section, "the grounds of a child day care or educational facility" means (a) in or on or within any building, structure, athletic playing field, a playground or land contained within the real property boundary line of a public or private child day care center as such term is defined in paragraph (c) of subdivision one of section three hundred ninety of the social services law, or nursery, pre-kindergarten or kindergarten, or (b) any area accessible to the public located within one thousand feet of the real property boundary line comprising any such facility or any parked automobile or other parked vehicle located within one thousand feet of the real property boundary line comprising any such facility. For the purposes of this section an "area accessible to the public" shall mean sidewalks, streets, parking lots, parks, playgrounds, stores and restaurants.
- 6. For the purposes of this section, a rebuttable presumption shall be established that a person has knowledge that they are within the grounds of a child day care or educational facility when notice is conspicuously posted of the presence or proximity of such facility.

Criminal sale of a controlled substance in or near school grounds is a class B felony.

§ 220.45 Criminally possessing a hypodermic instrument

A person is guilty of criminally possessing a hypodermic instrument when he or she knowingly and unlawfully possesses or sells a hypodermic syringe or hypodermic needle. It shall not be a violation of this section when a person obtains and possesses a hypodermic syringe or hypodermic needle pursuant to section thirty-three hundred eighty-one of the public health law.

Criminally possessing a hypodermic instrument is

a class A misdemeanor.

§ 220.46 Criminal injection of a narcotic drug

A person is guilty of criminal injection of a narcotic drug when he knowingly and unlawfully possesses a narcotic drug and he intentionally injects by means of a hypodermic syringe or hypodermic needle all or any portion of that drug into the body of another person with the latter's consent.

Criminal injection of a narcotic drug is a class E felony.

§ 220.48 Criminal sale of a controlled substance to a child

A person is guilty of criminal sale of a controlled substance to a child when, being over twenty-one years old, he or she knowingly and unlawfully sells a controlled substance in violation of section 220.34 or 220.39 of this article to a person less than seventeen years old.

Criminal sale of a controlled substance to a child is a class B felony.

§ 220.50 Criminally using drug paraphernalia in the second degree

A person is guilty of criminally using drug paraphernalia in the second degree when he knowingly possesses or sells:

- 1. Diluents, dilutants or adulterants, including but not limited to, any of the following: quinine hydrochloride, mannitol, mannite, lactose or dextrose, adapted for the dilution of narcotic drugs or stimulants under circumstances evincing an intent to use, or under circumstances evincing knowledge that some person intends to use, the same for purposes of unlawfully mixing, compounding, or otherwise preparing any narcotic drug or stimulant; or
- 2. Gelatine capsules, glassine envelopes, vials, capsules or any other material suitable for the packaging of individual quantities of narcotic drugs or stimulants under circumstances evincing an intent to use, or under circumstances evincing knowledge that some person intends to use, the same for the purpose of unlawfully manufacturing, packaging or dispensing of any narcotic drug or stimulant; or
- 3. Scales and balances used or designed for the purpose of weighing or measuring controlled substances, under circumstances evincing an

intent to use, or under circumstances evincing knowledge that some person intends to use, the same for purpose of unlawfully manufacturing, packaging or dispensing of any narcotic drug or stimulant.

Criminally using drug paraphernalia in the second degree is a class A misdemeanor.

§ 220.55 Criminally using drug paraphernalia in the first degree

A person is guilty of criminally using drug paraphernalia in the first degree when he commits the crime of criminally using drug paraphernalia in the second degree and he has previously been convicted of criminally using drug paraphernalia in the second degree.

Criminally using drug paraphernalia in the first degree is a class D felony.

§ 220.60 Criminal possession of precursors of controlled substances

A person is guilty of criminal possession of precursors of controlled substances when, with intent to manufacture a controlled substance unlawfully, he possesses at the same time:

- (a) carbamide (urea) and propanedioc and malonic acid or its derivatives; or
- (b) ergot or an ergot derivative and diethylamine or dimethylformamide or diethylamide; or
- (c) phenylacetone (1-phenyl-2 propanone) and hydroxylamine or ammonia or formamide or benzaldehyde or nitroethane or methylamine.
 - (d) pentazocine and methyliodide; or
- (e) phenylacetonitrile and dichlorodiethyl methylamine or dichlorodiethyl benzylamine; or
- (f) diephenylacetonitrile [FN1] and dimethylaminoisopropyl chloride; or
- (g) piperidine and cyclohexanone and bromobenzene and lithium or magnesium; or
- (h) 2, 5-dimethoxy benzaldehyde and nitroethane and a reducing agent.

Criminal prossession [FN1] of precursors of controlled substances is a class E felony.

[FN1] So in original.

§ 220.65 Criminal sale of a prescription for a controlled substance

A person is guilty of criminal sale of a prescription

for a controlled substance when, being a practitioner, as that term is defined in section thirty-three hundred two of the public health law, he knowingly and unlawfully sells a prescription for a controlled substance. For the purposes of this section, a person sells a prescription for a controlled substance unlawfully when he does so other than in good faith in the course of his professional practice.

Criminal sale of a prescription is a class C felony.

§ 220.70 Criminal possession of methamphetamine manufacturing material in the second degree

A person is guilty of criminal possession of methamphetamine manufacturing material in the second degree when he or she possesses a precursor, a chemical reagent or a solvent with the intent to use or knowing another intends to use such precursor, chemical reagent, or solvent to unlawfully produce, prepare or manufacture methamphetamine.

Criminal possession of methamphetamine manufacturing material in the second degree is a class A misdemeanor.

§ 220.71 Criminal possession of methamphetamine manufacturing material in the first degree

A person is guilty of criminal possession of methamphetamine manufacturing material in the first degree when he or she commits the offense of criminal possession of methamphetamine manufacturing material in the second degree, as defined in section 220.70 of this article, and has previously been convicted within the preceding five years of criminal possession of methamphetamine manufacturing material in the second degree, as defined in section 220.70 of this article, or a violation of this section.

Criminal possession of methamphetamine manufacturing material in the first degree is a class E felony.

§ 220.72 Criminal possession of precursors of methamphetamine

A person is guilty of criminal possession of precursors of methamphetamine when he or she possesses at the same time a precursor and a solvent or chemical reagent, with intent to use or knowing that another intends to use each such precursor, solvent or chemical reagent to unlawfully

manufacture methamphetamine.

Criminal possession of precursors of methamphetamine is a class E felony.

§ 220.73 Unlawful manufacture of methamphetamine in the third degree

A person is guilty of unlawful manufacture of methamphetamine in the third degree when he or she possesses at the same time and location, with intent to use, or knowing that another intends to use each such product to unlawfully manufacture, prepare or produce methamphetamine:

- 1. Two or more items of laboratory equipment and two or more precursors, chemical reagents or solvents in any combination; or
- 2. One item of laboratory equipment and three or more precursors, chemical reagents or solvents in any combination; or
 - 3. A precursor:
- (a) mixed together with a chemical reagent or solvent; or
- (b) with two or more chemical reagents and/or solvents mixed together.

Unlawful manufacture of methamphetamine in the third degree is a class D felony.

§ 220.74 Unlawful manufacture of methamphetamine in the second degree

A person is guilty of unlawful manufacture of methamphetamine in the second degree when he or she:

- 1. Commits the offense of unlawful manufacture of methamphetamine in the third degree as defined in section 220.73 of this article in the presence of another person under the age of sixteen, provided, however, that the actor is at least five years older than such other person under the age of sixteen; or
- 2. Commits the crime of unlawful manufacture of methamphetamine in the third degree as defined in section 220.73 of this article and has previously been convicted within the preceding five years of the offense of criminal possession of precursors of methamphetamine as defined in section 220.72 of this article, criminal possession of methamphetamine manufacturing material in the first degree as defined in section 220.71 of this article, unlawful disposal of methamphetamine laboratory material as defined in section 220.76 of this article, unlawful manufacture of

methamphetamine in the third degree as defined in section 220.73 of this article, unlawful manufacture of methamphetamine in the second degree as defined in this section, or unlawful manufacture of methamphetamine in the first degree as defined in section 220. 75 of this article.

Unlawful manufacture of methamphetamine in the second degree is a class C felony.

§ 220.75 Unlawful manufacture of methamphetamine in the first degree

A person is guilty of unlawful manufacture of methamphetamine in the first degree when such person commits the crime of unlawful manufacture of methamphetamine in the second degree, as defined in subdivision one of section 220.74 of this article, after having previously been convicted within the preceding five years of unlawful manufacture of methamphetamine in the third degree, as defined in section 220.73, unlawful manufacture of methamphetamine in the second degree, as defined in section 220.74 of this article, or unlawful manufacture of methamphetamine in the first degree, as defined in this section.

Unlawful manufacturer of methamphetamine in the first degree is a class B felony.

§ 220.76 Unlawful disposal of methamphetamine laboratory material

A person is guilty of unlawful disposal of methamphetamine laboratory material when, knowing that such actions are in furtherance of a methamphetamine operation, he or she knowingly disposes of, or possesses with intent to dispose of, hazardous or dangerous material under circumstances that create a substantial risk to human health or safety or a substantial danger to the environment.

Unlawful disposal of methamphetamine laboratory material is a class E felony.

§ 220.77 Operating as a major trafficker

A person is guilty of operating as a major trafficker when:

1. Such person acts as a director of a controlled substance organization during any period of twelve months or less, during which period such controlled substance organization sells one or more controlled substances, and the proceeds collected or due from such sale or sales have a total aggregate value of seventy-five thousand dollars or more; or

- 2. As a profiteer, such person knowingly and unlawfully sells, on one or more occasions within six months or less, a narcotic drug, and the proceeds collected or due from such sale or sales have a total aggregate value of seventy-five thousand dollars or more.
- 3. As a profiteer, such person knowingly and unlawfully possesses, on one or more occasions within six months or less, a narcotic drug with intent to sell the same, and such narcotic drugs have a total aggregate value of seventy-five thousand dollars or more.

Operating as a major trafficker is a class A-I felony.

§ 220.78 Witness or victim of drug or alcohol overdose

- 1. A person who, in good faith, seeks health care for someone who is experiencing a drug or alcohol overdose or other life threatening medical emergency shall not be charged or prosecuted for a controlled substance offense under article two hundred twenty or a marihuana offense under article two hundred twenty-one of this title, other than an offense involving sale for consideration or other benefit or gain, or charged or prosecuted for possession of alcohol by a person under age twenty-one years under section sixty-five-c of the alcoholic beverage control law, or for possession of drug paraphernalia under article thirty-nine of the general business law, with respect to any controlled substance, marihuana, alcohol or paraphernalia that was obtained as a result of such seeking or receiving of health care.
- 2. A person who is experiencing a drug or alcohol overdose or other life threatening medical emergency and, in good faith, seeks health care for himself or herself or is the subject of such a good faith request for health care, shall not be charged or prosecuted for a controlled substance offense under this article or a marihuana offense under article two hundred twenty-one of this title, other than an offense involving sale for consideration or other benefit or gain, or charged or prosecuted for possession of alcohol by a person under age twenty-one years under section sixty-five-c of the alcoholic beverage control law, or for possession of drug paraphernalia under article thirty-nine of the general business law, with respect to any substance, marihuana, alcohol or paraphernalia that was obtained as a result of such seeking or receiving of health care.

- 3. Definitions. As used in this section the following terms shall have the following meanings:
- (a) "Drug or alcohol overdose" or "overdose" means an acute condition including, but not limited to, physical illness, coma, mania, hysteria or death, which is the result of consumption or use of a controlled substance or alcohol and relates to an adverse reaction to or the quantity of the controlled substance or alcohol or a substance with which the controlled substance or alcohol was combined; provided that a patient's condition shall be deemed to be a drug or alcohol overdose if a prudent layperson, possessing an average knowledge of medicine and health, could reasonably believe that the condition is in fact a drug or alcohol overdose and (except as to death) requires health care.
- (b) "Health care" means the professional services provided to a person experiencing a drug or alcohol overdose by a health care professional licensed, registered or certified under title eight of the education law or article thirty of the public health law who, acting within his or her lawful scope of practice, may provide diagnosis, treatment or emergency services for a person experiencing a drug or alcohol overdose.
- 4. It shall be an affirmative defense to a criminal sale controlled substance offense under this article or a criminal sale of marihuana offense under article two hundred twenty-one of this title, not covered by subdivision one or two of this section, with respect to any controlled substance or marihuana which was obtained as a result of such seeking or receiving of health care, that:
- (a) the defendant, in good faith, seeks health care for someone or for him or herself who is experiencing a drug or alcohol overdose or other life threatening medical emergency; and
- (b) the defendant has no prior conviction for the commission or attempted commission of a class A-I, A-II or B felony under this article.
- 5. Nothing in this section shall be construed to bar the admissibility of any evidence in connection with the investigation and prosecution of a crime with regard to another defendant who does not independently qualify for the bar to prosecution or for the affirmative defense; nor with regard to other crimes committed by a person who otherwise qualifies under this section; nor shall anything in this section be construed to bar any seizure pursuant to law, including but not limited to pursuant to section thirty-three hundred

eighty-seven of the public health law.

6. The bar to prosecution described in subdivisions one and two of this section shall not apply to the prosecution of a class A-I felony under this article, and the affirmative defense described in subdivision four of this section shall not apply to the prosecution of a class A-I or A-II felony under this article.

Article 221. Offenses Involving Marihuana

§ 221.00 Marihuana; definitions

Unless the context in which they are used clearly otherwise requires, the terms occurring in this article shall have the same meaning ascribed to them in article two hundred twenty of this chapter.

§ 221.05 Unlawful possession of marihuana

A person is guilty of unlawful possession of marihuana when he knowingly and unlawfully possesses marihuana.

Unlawful possession of marihuana is a violation punishable only by a fine of not more than one hundred dollars. However, where the defendant has previously been convicted of an offense defined in this article or article 220 of this chapter, committed within the three years immediately preceding such violation, it shall be punishable (a) only by a fine of not more than two hundred dollars, if the defendant was previously convicted of one such offense committed during such period, and (b) by a fine of not more than two hundred fifty dollars or a term of imprisonment not in excess of fifteen days or both, if the defendant was previously convicted of two such offenses committed during such period.

§ 221.10 Criminal possession of marihuana in the fifth degree

A person is guilty of criminal possession of marihuana in the fifth degree when he knowingly and unlawfully possesses:

- 1. marihuana in a public place, as defined in section 240.00 of this chapter, and such marihuana is burning or open to public view; or
- 2. one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than twenty-five grams.

Criminal possession of marihuana in the fifth degree is a class B misdemeanor.

§ 221.15 Criminal possession of marihuana in the fourth degree

A person is guilty of criminal possession of marihuana in the fourth degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than two ounces.

Criminal possession of marihuana in the fourth degree is a class A misdemeanor.

§ 221.20 Criminal possession of marihuana in the third degree

A person is guilty of criminal possession of marihuana in the third degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than eight ounces.

Criminal possession of marihuana in the third degree is a class E felony.

§ 221.25 Criminal possession of marihuana in the second degree

A person is guilty of criminal possession of marihuana in the second degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than sixteen ounces.

Criminal possession of marihuana in the second degree is a class D felony.

§ 221.30 Criminal possession of marihuana in the first degree

A person is guilty of criminal possession of marihuana in the first degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than ten pounds.

Criminal possession of marihuana in the first degree is a class C felony.

§ 221.35 Criminal sale of marihuana in the fifth degree

A person is guilty of criminal sale of marihuana in the fifth degree when he knowingly and unlawfully sells, without consideration, one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of two grams or less; or one cigarette containing marihuana.

Criminal sale of marihuana in the fifth degree is a class B misdemeanor.

§ 221.40 Criminal sale of marihuana in the fourth degree

A person is guilty of criminal sale of marihuana in the fourth degree when he knowingly and unlawfully sells marihuana except as provided in section 221.35 of this article.

Criminal sale of marihuana in the fourth degree is a class A misdemeanor.

§ 221.45 Criminal sale of marihuana in the third degree

A person is guilty of criminal sale of marihuana in the third degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than twenty-five grams.

Criminal sale of marihuana in the third degree is a class E felony.

§ 221.50 Criminal sale of marihuana in the second degree

A person is guilty of criminal sale of marihuana in the second degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than four ounces, or knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana to a person less than eighteen years of age.

Criminal sale of marihuana in the second degree is a class D felony.

§ 221.55 Criminal sale of marihuana in the first degree

A person is guilty of criminal sale of marihuana in the first degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than sixteen ounces.

Criminal sale of marihuana in the first degree is a class C felony.

Article 225. Gambling Offenses

- § 225.00 Gambling offenses; definitions of terms
- § 225.05 Promoting gambling in the second degree
- § 225.10 Promoting gambling in the first degree
- § 225.15 Possession of gambling records in the second degree
- § 225.20 Possession of gambling records in the first degree
- § 225.25 Possession of gambling records; defense
- § 225.30 Possession of a gambling device
- § 225.32 Possession of a gambling device; defenses
- § 225.35 Gambling offenses; presumptions
- § 225.40 Lottery offenses; no defense

Article 230. Prostitution Offenses

- § 230.00 Prostitution
- § 230.02 Patronizing a prostitute; definitions
- § 230.03 Prostitution in a school zone
- § 230.04 Patronizing a prostitute in the third degree
- § 230.05 Patronizing a prostitute in the second degree
- § 230.06 Patronizing a prostitute in the first degree
- § 230.07 Patronizing a prostitute; defense
- § 230.10 Prostitution and patronizing a prostitute; no defense

- § 230.15 Promoting prostitution; definitions of terms
- § 230.19 Promoting prostitution in a school zone
- § 230.20 Promoting prostitution in the fourth degree
- § 230.25 Promoting prostitution in the third degree
- § 230.30 Promoting prostitution in the second degree
- § 230.32 Promoting prostitution in the first degree
- § 230.33. Compelling prostitution
- § 230.34 Sex trafficking
- § 230.35. Promoting or compelling prostitution; accomplice
- § 230.36 Sex trafficking; accomplice
- § 230.40 Permitting prostitution

Article 235. Obscenity and Related Offenses

- § 235.00 Obscenity; definitions of terms
- § 235.05 Obscenity in the third degree
- § 235.06 Obscenity in the second degree
- § 235.07 Obscenity in the first degree
- § 235.10 Obscenity; presumptions
- § 235.15 Obscenity or disseminating indecent material to minors in the second degree; defense
- § 235.20 Disseminating indecent material to minors; definitions of terms
- § 235.21 Disseminating indecent material to minors in the second degree
- § 235.22 Disseminating indecent material to minors in the first degree
- § 235.23 Disseminating indecent material to minors; presumption and defenses
- § 235.24 Disseminating indecent material to minors: limitations

TITLE M. OFFENSES AGAINST PUBLIC HEALTH AND MORALS

Article 220. Controlled Substances Offenses

§ 220.00 Controlled substances; definitions

- 1. "Sell" means to sell, exchange, give or dispose of to another, or to offer or agree to do the same.
- 2. "Unlawfully" means in violation of article thirty-three of the public health law.
- 3. "Ounce" means an avoirdupois ounce as applied to solids or semisolids, and a fluid ounce as applied to liquids.
 - 4. "Pound" means an avoirdupois pound.
- 5. "Controlled substance" means any substance listed in schedule I, II, III, IV or V of section thirty-three hundred six of the public health law other than marihuana, but including concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of such law.
- 6. "Marihuana" means "marihuana" or "concentrated cannabis" as those terms are defined in section thirty-three hundred two of the public health law.
- 7. "Narcotic drug" means any controlled substance listed in schedule I(b), I(c), II(b) or II(c) other than methadone.
- 8. [Eff. until Feb. 23, 2013. See, also, subd. 8 below.] "Narcotic preparation" means any controlled substance listed in schedule III(d) or III(e).
- 8. [Eff. Feb. 23, 2013. See, also, subd. 8 above.] "Narcotic preparation" means any controlled substance listed in schedule II(b-1), III(d) or III(e).
- 9. "Hallucinogen" means any controlled substance listed in schedule I(d) (5), (18), (19), (20), (21) and (22).
- 10. "Hallucinogenic substance" means any controlled substance listed in schedule I(d) other than concentrated cannabis, lysergic acid diethylamide, or an hallucinogen.
- 11. "Stimulant" means any controlled substance listed in schedule I(f), II(d).
- 12. "Dangerous depressant" means any controlled substance listed in schedule I(e)(2), (3), II(e), III(c)(3) or IV(c)(2), (31), (32), (40).

- 13. "Depressant" means any controlled substance listed in schedule IV(c) except (c)(2), (31), (32), (40).
- 14. "School grounds" means (a) in or on or within any building, structure, athletic playing field, playground or land contained within the real property boundary line of a public or private elementary, parochial, intermediate, junior high, vocational, or high school, or (b) any area accessible to the public located within one thousand feet of the real property boundary line comprising any such school or any parked automobile or other parked vehicle located within one thousand feet of the real property boundary line comprising any such school. For the purposes of this section an "area accessible to the public" shall mean sidewalks, streets, parking lots, parks, playgrounds, stores and restaurants.
- 15. "Prescription for a controlled substance" means a direction or authorization, by means of an official New York state prescription form, a written prescription form or an oral prescription, which will permit a person to lawfully obtain a controlled substance from any person authorized to dispense controlled substances.
- 16. For the purposes of sections 220.70, 220.71, 220.72, 220.73, 220.74, 220.75 and 220.76 of this article:
- (a) "Precursor" means ephedrine, pseudoephedrine, or any salt, isomer or salt of an isomer of such substances.
- (b) "Chemical reagent" means a chemical reagent that can be used in the manufacture, production or preparation of methamphetamine.
- (c) "Solvent" means a solvent that can be used in the manufacture, production or preparation of methamphetamine.
- (d) "Laboratory equipment" means any items, components or materials that can be used in the manufacture, preparation or production of methamphetamine.
- (e) "Hazardous or dangerous material" means any substance, or combination of substances, that results from or is used in the manufacture, preparation or production of methamphetamine which, because of its quantity, concentration, or physical or chemical characteristics, poses a substantial risk to human health or safety, or a substantial danger to the environment.

- 17. "School bus" means every motor vehicle owned by a public or governmental agency or private school and operated for the transportation of pupils, teachers and other persons acting in a supervisory capacity, to or from school or school activities or privately owned and operated for compensation for the transportation of pupils, children of pupils, teachers and other persons acting in a supervisory capacity to or from school or school activities.
- 18. "Controlled substance organization" means four or more persons sharing a common purpose to engage in conduct that constitutes or advances the commission of a felony under this article.
- 19. "Director" means a person who is the principal administrator, organizer, or leader of a controlled substance organization or one of several principal administrators, organizers, or leaders of a controlled substance organization.
- 20. "Profiteer" means a person who: (a) is a director of a controlled substance organization; (b) is a member of a controlled substance organization and has managerial responsibility over one or more other members of that organization; or (c) arranges, devises or plans one or more transactions constituting a felony under this article so as to obtain profits or expected profits. A person is not a profiteer if he or she is acting only as an employee; or if he or she is acting as an accommodation to a friend or relative; or if he or she is acting only under the direction and control of others and exercises no substantial, independent role in arranging or directing the transactions in question.

§ 220.03 Criminal possession of a controlled substance in the seventh degree

A person is guilty of criminal possession of a controlled substance in the seventh degree when he or she knowingly and unlawfully possesses a controlled substance; provided, however, that it shall not be a violation of this section when a person possesses a residual amount of a controlled substance and that residual amount is in or on a hypodermic syringe or hypodermic needle obtained and possessed pursuant to section thirty-three hundred eighty-one of the public health law; nor shall it be a violation of this section when a person's unlawful possession of a controlled substance is discovered as a result of seeking immediate health care as defined in paragraph (b) of subdivision three of section 220.78 of the penal

law, for either another person or him or herself because such person is experiencing a drug or alcohol overdose or other life threatening medical emergency as defined in paragraph (a) of subdivision three of section 220.78 of the penal law.

Criminal possession of a controlled substance in the seventh degree is a class A misdemeanor.

§ 220.06 Criminal possession of a controlled substance in the fifth degree

A person is guilty of criminal possession of a controlled substance in the fifth degree when he knowingly and unlawfully possesses:

- 1. a controlled substance with intent to sell it; or
- 2. one or more preparations, compounds, mixtures or substances containing a narcotic preparation and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 3. phencyclidine and said phencyclidine weighs fifty milligrams or more; or
- 4. one or more preparations, compounds, mixtures or substances containing concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law and said preparations, compounds, mixtures or substances are of an aggregate weight of one-fourth ounce or more; or
- 5. cocaine and said cocaine weighs five hundred milligrams or more.
- 6. ketamine and said ketamine weighs more than one thousand milligrams; or
- 7. ketamine and has previously been convicted of possession or the attempt to commit possession of ketamine in any amount; or
- 8. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of twenty-eight grams or more.

Criminal possession of a controlled substance in the fifth degree is a class D felony.

§ 220.09 Criminal possession of a controlled substance in the fourth degree

A person is guilty of criminal possession of a controlled substance in the fourth degree when he knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 2. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 3. one or more preparations, compounds, mixtures or substances containing a narcotic preparation and said preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or
- 4. a stimulant and said stimulant weighs one gram or more; or
- 5. lysergic acid diethylamide and said lysergic acid diethylamide weighs one milligram or more; or
- 6. a hallucinogen and said hallucinogen weighs twenty-five milligrams or more; or
- 7. a hallucinogenic substance and said hallucinogenic substance weighs one gram or more; or
- 8. a dangerous depressant and such dangerous depressant weighs ten ounces or more; or
- 9. a depressant and such depressant weighs two pounds or more; or
- 10. one or more preparations, compounds, mixtures or substances containing concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law and said preparations, compounds, mixtures or substances are of an aggregate weight of one ounce or more; or
- 11. phencyclidine and said phencyclidine weighs two hundred fifty milligrams or more; or
- 12. methadone and said methadone weighs three hundred sixty milligrams or more; or
- 13. phencyclidine and said phencyclidine weighs fifty milligrams or more with intent to sell it and has previously been convicted of an offense defined in this article or the attempt or conspiracy to commit

any such offense; or

- 14. ketamine and said ketamine weighs four thousand milligrams or more; or
- 15. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of two hundred grams or more.

Criminal possession of a controlled substance in the fourth degree is a class C felony.

§ 220.16 Criminal possession of a controlled substance in the third degree

A person is guilty of criminal possession of a controlled substance in the third degree when he knowingly and unlawfully possesses:

- 1. a narcotic drug with intent to sell it; or
- 2. a stimulant, hallucinogen, hallucinogenic substance, or lysergic acid diethylamide, with intent to sell it and has previously been convicted of an offense defined in article two hundred twenty or the attempt or conspiracy to commit any such offense; or
- 3. a stimulant with intent to sell it and said stimulant weighs one gram or more; or
- 4. lysergic acid diethylamide with intent to sell it and said lysergic acid diethylamide weighs one milligram or more; or
- 5. a hallucinogen with intent to sell it and said hallucinogen weighs twenty-five milligrams or more; or
- 6. a hallucinogenic substance with intent to sell it and said hallucinogenic substance weighs one gram or more; or
- 7. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers with intent to sell it and said preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 8. a stimulant and said stimulant weighs five grams or more; or
- 9. lysergic acid diethylamide and said lysergic acid diethylamide weighs five milligrams or more; or

- 10. a hallucinogen and said hallucinogen weighs one hundred twenty-five milligrams or more; or
- 11. a hallucinogenic substance and said hallucinogenic substance weighs five grams or more; or
- 12. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 13. phencyclidine and said phencyclidine weighs one thousand two hundred fifty milligrams or more.

Criminal possession of a controlled substance in the third degree is a class B felony.

§ 220.18 Criminal possession of a controlled substance in the second degree

A person is guilty of criminal possession of a controlled substance in the second degree when he or she knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of four ounces or more; or
- 2. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and said preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or
- 3. a stimulant and said stimulant weighs ten grams or more; or
- 4. lysergic acid diethylamide and said lysergic acid diethylamide weighs twenty-five milligrams or more; or
- 5. a hallucinogen and said hallucinogen weighs six hundred twenty-five milligrams or more; or
- 6. a hallucinogenic substance and said hallucinogenic substance weighs twenty-five grams or more; or
- 7. methadone and said methadone weighs two thousand eight hundred eighty milligrams or more.

Criminal possession of a controlled substance in the second degree is a class A-II felony.

§ 220.21 Criminal possession of a controlled substance in the first degree

A person is guilty of criminal possession of a controlled substance in the first degree when he or she knowingly and unlawfully possesses:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and said preparations, compounds, mixtures or substances are of an aggregate weight of eight ounces or more; or
- 2. methadone and said methadone weighs five thousand seven hundred sixty milligrams or more.

Criminal possession of a controlled substance in the first degree is a class A-I felony.

§ 220.25 Criminal possession of a controlled substance; presumption

- 1. The presence of a controlled substance in an automobile, other than a public omnibus, is presumptive evidence of knowing possession thereof by each and every person in the automobile at the time such controlled substance was found; except that such presumption does not apply (a) to a duly licensed operator of an automobile who is at the time operating it for hire in the lawful and proper pursuit of his trade, or (b) to any person in the automobile if one of them, having obtained the controlled substance and not being under duress, is authorized to possess it and such controlled substance is in the same container as when he received possession thereof, or (c) when the controlled substance is concealed upon the person of one of the occupants.
- 2. The presence of a narcotic drug, narcotic preparation, marihuana or phencyclidine in open view in a room, other than a public place, under circumstances evincing an intent to unlawfully mix, compound, package or otherwise prepare for sale such controlled substance is presumptive evidence of knowing possession thereof by each and every person in close proximity to such controlled substance at the time such controlled substance was found; except that such presumption does not apply to any such persons if (a) one of them, having obtained such controlled substance and not being under duress, is authorized to possess it and such controlled substance is in the same container as when he received possession thereof, or (b) one of them has such controlled substance upon his person.

§ 220.28. Use of a child to commit a controlled substance offense

1. A person is guilty of use of a child to commit a

controlled substance offense when, being eighteen years old or more, he or she commits a felony sale or felony attempted sale of a controlled substance in violation of this article and, as part of that criminal transaction, knowingly uses a child to effectuate such felony sale or felony attempted sale of such controlled substance.

2. For purposes of this section, "uses a child to effectuate the felony sale or felony attempted sale of such controlled substance" means conduct by which the actor: (a) conceals such controlled substance on or about the body or person of such child for the purpose of effectuating the criminal sale or attempted sale of such controlled substance to a third person; or (b) directs, forces or otherwise requires such child to sell or attempt to sell or offer direct assistance to the defendant in selling or attempting to sell such controlled substance to a third person.

For purposes of this section, "child" means a person less than sixteen years of age.

Use of a child to commit a controlled substance offense is a class E felony.

§ 220.31 Criminal sale of a controlled substance in the fifth degree

A person is guilty of criminal sale of a controlled substance in the fifth degree when he knowingly and unlawfully sells a controlled substance.

Criminal sale of a controlled substance in the fifth degree is a class D felony.

§ 220.34 Criminal sale of a controlled substance in the fourth degree

A person is guilty of criminal sale of a controlled substance in the fourth degree when he knowingly and unlawfully sells:

- 1. a narcotic preparation; or
- 2. a dangerous depressant or a depressant and the dangerous depressant weighs ten ounces or more, or the depressant weighs two pounds or more; or
- 3. concentrated cannabis as defined in paragraph (a) of subdivision four of section thirty-three hundred two of the public health law; or
- 4. phencyclidine and the phencyclidine weighs fifty milligrams or more; or
 - 5. methadone; or
 - 6. any amount of phencyclidine and has

previously been convicted of an offense defined in this article or the attempt or conspiracy to commit any such offense; or

- 6-a. ketamine and said ketamine weighs four thousand milligrams or more.
- 7. a controlled substance in violation of section 220.31 of this article, when such sale takes place upon school grounds or on a school bus; or
- 8. a controlled substance in violation of section 220.31 of this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds. As used in this subdivision, the phrase "the grounds of a child day care or educational facility" shall have the same meaning as provided for in subdivision five of section 220.44 of this article. For the purposes of this subdivision, a rebuttable presumption shall be established that a person has knowledge that they are within the grounds of a child day care or educational facility when notice is conspicuously posted of the presence or proximity of such facility; or
- 9. one or more preparations, compounds, mixtures or substances containing gamma hydroxybutyric acid, as defined in paragraph four of subdivision (e) of schedule I of section thirty-three hundred six of the public health law, and said preparations, compounds, mixtures or substances are of an aggregate weight of twenty-eight grams or more.

Criminal sale of a controlled substance in the fourth degree is a class C felony.

§ 220.39 Criminal sale of a controlled substance in the third degree

A person is guilty of criminal sale of a controlled substance in the third degree when he knowingly and unlawfully sells:

- 1. a narcotic drug; or
- 2. a stimulant, hallucinogen, hallucinogenic substance, or lysergic acid diethylamide and has previously been convicted of an offense defined in article two hundred twenty or the attempt or conspiracy to commit any such offense; or
- 3. a stimulant and the stimulant weighs one gram or more; or
- 4. lysergic acid diethylamide and the lysergic acid diethylamide weighs one milligram or more; or

- 5. a hallucinogen and the hallucinogen weighs twenty-five milligrams or more; or
- 6. a hallucinogenic substance and the hallucinogenic substance weighs one gram or more; or
- 7. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and the preparations, compounds, mixtures or substances are of an aggregate weight of one-eighth ounce or more; or
- 8. phencyclidine and the phencyclidine weighs two hundred fifty milligrams or more; or
- 9. a narcotic preparation to a person less than twenty-one years old.

Criminal sale of a controlled substance in the third degree is a class B felony.

§ 220.41 Criminal sale of a controlled substance in the second degree

A person is guilty of criminal sale of a controlled substance in the second degree when he knowingly and unlawfully sells:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and the preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 2. one or more preparations, compounds, mixtures or substances containing methamphetamine, its salts, isomers or salts of isomers and the preparations, compounds, mixtures or substances are of an aggregate weight of one-half ounce or more; or
- 3. a stimulant and the stimulant weighs five grams or more; or
- 4. lysergic acid diethylamide and the lysergic acid diethylamide weighs five milligrams or more; or
- 5. a hallucinogen and the hallucinogen weighs one hundred twenty-five milligrams or more; or
- 6. a hallucinogenic substance and the hallucinogenic substance weighs five grams or more; or
- 7. methadone and the methadone weighs three hundred sixty milligrams or more.

Criminal sale of a controlled substance in the second degree is a class A-II felony.

§ 220.43 Criminal sale of a controlled substance in the first degree

A person is guilty of criminal sale of a controlled substance in the first degree when he knowingly and unlawfully sells:

- 1. one or more preparations, compounds, mixtures or substances containing a narcotic drug and the preparations, compounds, mixtures or substances are of an aggregate weight of two ounces or more; or
- 2. methadone and the methadone weighs two thousand eight hundred eighty milligrams or more.

Criminal sale of a controlled substance in the first degree is a class A-I felony.

§ 220.44 Criminal sale of a controlled substance in or near school grounds

A person is guilty of criminal sale of a controlled substance in or near school grounds when he knowingly and unlawfully sells:

- 1. a controlled substance in violation of any one of subdivisions one through six-a of section 220.34 of this article, when such sale takes place upon school grounds or on a school bus; or
- 2. a controlled substance in violation of any one of subdivisions one through eight of section 220.39 of this article, when such sale takes place upon school grounds or on a school bus; or
- 3. a controlled substance in violation of any one of subdivisions one through six of section 220.34 of this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds; or
- 4. a controlled substance in violation of any one of subdivisions one through eight of section 220.39 of this article, when such sale takes place upon the grounds of a child day care or educational facility under circumstances evincing knowledge by the defendant that such sale is taking place upon such grounds.
- 5. For purposes of subdivisions three and four of this section, "the grounds of a child day care or educational facility" means (a) in or on or within any building, structure, athletic playing field, a playground or land contained within the real property boundary line of a public or private child day care center as such term is defined in

paragraph (c) of subdivision one of section three hundred ninety of the social services law, or nursery, pre-kindergarten or kindergarten, or (b) any area accessible to the public located within one thousand feet of the real property boundary line comprising any such facility or any parked automobile or other parked vehicle located within one thousand feet of the real property boundary line comprising any such facility. For the purposes of this section an "area accessible to the public" shall mean sidewalks, streets, parking lots, parks, playgrounds, stores and restaurants.

6. For the purposes of this section, a rebuttable presumption shall be established that a person has knowledge that they are within the grounds of a child day care or educational facility when notice is conspicuously posted of the presence or proximity of such facility.

Criminal sale of a controlled substance in or near school grounds is a class B felony.

§ 220.45 Criminally possessing a hypodermic instrument

A person is guilty of criminally possessing a hypodermic instrument when he or she knowingly and unlawfully possesses or sells a hypodermic syringe or hypodermic needle. It shall not be a violation of this section when a person obtains and possesses a hypodermic syringe or hypodermic needle pursuant to section thirty-three hundred eighty-one of the public health law.

Criminally possessing a hypodermic instrument is a class A misdemeanor.

§ 220.46 Criminal injection of a narcotic drug

A person is guilty of criminal injection of a narcotic drug when he knowingly and unlawfully possesses a narcotic drug and he intentionally injects by means of a hypodermic syringe or hypodermic needle all or any portion of that drug into the body of another person with the latter's consent.

Criminal injection of a narcotic drug is a class E felony.

§ 220.48 Criminal sale of a controlled substance to a child

A person is guilty of criminal sale of a controlled substance to a child when, being over twenty-one years old, he or she knowingly and unlawfully sells a controlled substance in violation of section 220.34 or 220.39 of this article to a person less

than seventeen years old.

Criminal sale of a controlled substance to a child is a class B felony.

§ 220.50 Criminally using drug paraphernalia in the second degree

A person is guilty of criminally using drug paraphernalia in the second degree when he knowingly possesses or sells:

- 1. Diluents, dilutants or adulterants, including but not limited to, any of the following: quinine hydrochloride, mannitol, mannite, lactose or dextrose, adapted for the dilution of narcotic drugs or stimulants under circumstances evincing an intent to use, or under circumstances evincing knowledge that some person intends to use, the same for purposes of unlawfully mixing, compounding, or otherwise preparing any narcotic drug or stimulant; or
- 2. Gelatine capsules, glassine envelopes, vials, capsules or any other material suitable for the packaging of individual quantities of narcotic drugs or stimulants under circumstances evincing an intent to use, or under circumstances evincing knowledge that some person intends to use, the same for the purpose of unlawfully manufacturing, packaging or dispensing of any narcotic drug or stimulant; or
- 3. Scales and balances used or designed for the purpose of weighing or measuring controlled substances, under circumstances evincing an intent to use, or under circumstances evincing knowledge that some person intends to use, the same for purpose of unlawfully manufacturing, packaging or dispensing of any narcotic drug or stimulant.

Criminally using drug paraphernalia in the second degree is a class A misdemeanor.

§ 220.55 Criminally using drug paraphernalia in the first degree

A person is guilty of criminally using drug paraphernalia in the first degree when he commits the crime of criminally using drug paraphernalia in the second degree and he has previously been convicted of criminally using drug paraphernalia in the second degree.

Criminally using drug paraphernalia in the first degree is a class D felony.

§ 220.60 Criminal possession of precursors of controlled substances

A person is guilty of criminal possession of precursors of controlled substances when, with intent to manufacture a controlled substance unlawfully, he possesses at the same time:

- (a) carbamide (urea) and propanedioc and malonic acid or its derivatives: or
- (b) ergot or an ergot derivative and diethylamine or dimethylformamide or diethylamide; or
- (c) phenylacetone (1-phenyl-2 propanone) and hydroxylamine or ammonia or formamide or benzaldehyde or nitroethane or methylamine.
 - (d) pentazocine and methyliodide; or
- (e) phenylacetonitrile and dichlorodiethyl methylamine or dichlorodiethyl benzylamine; or
- (f) diephenylacetonitrile [FN1] and dimethylaminoisopropyl chloride; or
- (g) piperidine and cyclohexanone and bromobenzene and lithium or magnesium; or
- (h) 2, 5-dimethoxy benzaldehyde and nitroethane and a reducing agent.

Criminal prossession [FN1] of precursors of controlled substances is a class E felony.

[FN1] So in original.

§ 220.65 Criminal sale of a prescription for a controlled substance

A person is guilty of criminal sale of a prescription for a controlled substance when, being a practitioner, as that term is defined in section thirty-three hundred two of the public health law, he knowingly and unlawfully sells a prescription for a controlled substance. For the purposes of this section, a person sells a prescription for a controlled substance unlawfully when he does so other than in good faith in the course of his professional practice.

Criminal sale of a prescription is a class C felony.

§ 220.70 Criminal possession of methamphetamine manufacturing material in the second degree

§ 220.71 Criminal possession of methamphetamine manufacturing material in the first degree

§ 220.72 Criminal possession of precursors of methamphetamine

- § 220.73 Unlawful manufacture of methamphetamine in the third degree
- § 220.74 Unlawful manufacture of methamphetamine in the second degree
- § 220.75 Unlawful manufacture of methamphetamine in the first degree
- § 220.76 Unlawful disposal of methamphetamine laboratory material
- § 220.77 Operating as a major trafficker
- § 220.78 Witness or victim of drug or alcohol overdose

Article 221. Offenses Involving Marihuana

§ 221.00 Marihuana; definitions

Unless the context in which they are used clearly otherwise requires, the terms occurring in this article shall have the same meaning ascribed to them in article two hundred twenty of this chapter.

§ 221.05 Unlawful possession of marihuana

A person is guilty of unlawful possession of marihuana when he knowingly and unlawfully possesses marihuana.

Unlawful possession of marihuana is a violation punishable only by a fine of not more than one hundred dollars. However, where the defendant has previously been convicted of an offense defined in this article or article 220 of this chapter, committed within the three years immediately preceding such violation, it shall be punishable (a) only by a fine of not more than two hundred dollars, if the defendant was previously convicted of one such offense committed during such period, and (b) by a fine of not more than two hundred fifty dollars or a term of imprisonment not in excess of fifteen days or both, if the defendant was previously convicted of two such offenses committed during such period.

§ 221.10 Criminal possession of marihuana in the fifth degree

A person is guilty of criminal possession of marihuana in the fifth degree when he knowingly and unlawfully possesses:

- 1. marihuana in a public place, as defined in section 240.00 of this chapter, and such marihuana is burning or open to public view; or
- 2. one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than twenty-five grams.

Criminal possession of marihuana in the fifth degree is a class B misdemeanor.

§ 221.15 Criminal possession of marihuana in the fourth degree

A person is guilty of criminal possession of marihuana in the fourth degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than two ounces.

Criminal possession of marihuana in the fourth degree is a class A misdemeanor.

§ 221.20 Criminal possession of marihuana in the third degree

A person is guilty of criminal possession of marihuana in the third degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than eight ounces.

Criminal possession of marihuana in the third degree is a class E felony.

§ 221.25 Criminal possession of marihuana in the second degree

A person is guilty of criminal possession of marihuana in the second degree when he knowingly and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than sixteen ounces.

Criminal possession of marihuana in the second degree is a class D felony.

§ 221.30 Criminal possession of marihuana in the first degree

A person is guilty of criminal possession of marihuana in the first degree when he knowingly

and unlawfully possesses one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than ten pounds.

Criminal possession of marihuana in the first degree is a class C felony.

§ 221.35 Criminal sale of marihuana in the fifth degree

A person is guilty of criminal sale of marihuana in the fifth degree when he knowingly and unlawfully sells, without consideration, one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of two grams or less; or one cigarette containing marihuana.

Criminal sale of marihuana in the fifth degree is a class B misdemeanor.

§ 221.40 Criminal sale of marihuana in the fourth degree

A person is guilty of criminal sale of marihuana in the fourth degree when he knowingly and unlawfully sells marihuana except as provided in section 221.35 of this article.

Criminal sale of marihuana in the fourth degree is a class A misdemeanor.

§ 221.45 Criminal sale of marihuana in the third degree

A person is guilty of criminal sale of marihuana in the third degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than twenty-five grams.

Criminal sale of marihuana in the third degree is a class E felony.

§ 221.50 Criminal sale of marihuana in the second degree

A person is guilty of criminal sale of marihuana in the second degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than four ounces, or knowingly and unlawfully sells one or more preparations,

compounds, mixtures or substances containing marihuana to a person less than eighteen years of age.

Criminal sale of marihuana in the second degree is a class D felony.

§ 221.55 Criminal sale of marihuana in the first degree

A person is guilty of criminal sale of marihuana in the first degree when he knowingly and unlawfully sells one or more preparations, compounds, mixtures or substances containing marihuana and the preparations, compounds, mixtures or substances are of an aggregate weight of more than sixteen ounces.

Criminal sale of marihuana in the first degree is a class C felony.

Article 225. Gambling Offenses

- § 225.00 Gambling offenses; definitions of terms
- § 225.05 Promoting gambling in the second degree
- § 225.10 Promoting gambling in the first degree
- § 225.15 Possession of gambling records in the second degree
- § 225.20 Possession of gambling records in the first degree
- § 225.25 Possession of gambling records; defense
- § 225.30 Possession of a gambling device
- § 225.32 Possession of a gambling device; defenses
- § 225.35 Gambling offenses; presumptions
- § 225.40 Lottery offenses; no defense

Article 230. Prostitution Offenses

- § 230.00 Prostitution
- § 230.02 Patronizing a prostitute; definitions
- § 230.03 Prostitution in a school zone
- § 230.04 Patronizing a prostitute in the third degree
- § 230.05 Patronizing a prostitute in the second degree

§ 230.06 Patronizing a prostitute in the first degree

§ 230.07 Patronizing a prostitute; defense

§ 230.10 Prostitution and patronizing a prostitute; no defense

§ 230.15 Promoting prostitution; definitions of terms

§ 230.19 Promoting prostitution in a school zone

§ 230.20 Promoting prostitution in the fourth degree

§ 230.25 Promoting prostitution in the third degree

§ 230.30 Promoting prostitution in the second degree

§ 230.32 Promoting prostitution in the first degree

§ 230.33. Compelling prostitution

§ 230.34 Sex trafficking

§ 230.35. Promoting or compelling prostitution; accomplice

§ 230.36 Sex trafficking; accomplice

§ 230.40 Permitting prostitution

Article 235. Obscenity and Related Offenses

§ 235.00 Obscenity; definitions of terms

§ 235.05 Obscenity in the third degree

§ 235.06 Obscenity in the second degree

§ 235.07 Obscenity in the first degree

§ 235.10 Obscenity; presumptions

§ 235.15 Obscenity or disseminating indecent material to minors in the second degree; defense

§ 235.20 Disseminating indecent material to minors; definitions of terms

§ 235.21 Disseminating indecent material to minors in the second degree

§ 235.22 Disseminating indecent material to minors in the first degree

§ 235.23 Disseminating indecent material to minors; presumption and defenses

§ 235.24 Disseminating indecent material to minors; limitations

TITLE N. OFFENSES AGAINST PUBLIC ORDER, PUBLIC SENSIBILITIES AND THE RIGHT TO PRIVACY

Article 240. Offenses Against Public Order

§ 240.00 Offenses against public order; definitions of terms

The following definitions are applicable to this article:

- 1. "Public place" means a place to which the public or a substantial group of persons has access, and includes, but is not limited to, highways, transportation facilities, schools, places of amusement, parks, playgrounds, and hallways, lobbies and other portions of apartment houses and hotels not constituting rooms or apartments designed for actual residence.
- 2. "Transportation facility" means any conveyance, premises or place used for or in connection with public passenger transportation, whether by air, railroad, motor vehicle or any other method. It includes aircraft, watercraft, railroad cars, buses, school buses as defined in section one hundred forty-two of the vehicle and traffic law, and air, boat, railroad and bus terminals and stations and all appurtenances thereto.
- 3. "School grounds" means in or on or within any building, structure, school bus as defined in section one hundred forty-two of the vehicle and traffic law, athletic playing field, playground or land contained within the real property boundary line of a public or private elementary, parochial, intermediate, junior high, vocational or high school.
- 4. "Hazardous substance" shall mean any physical, chemical, microbiological or radiological substance or matter which, because of its quantity, concentration, or physical, chemical or infectious characteristics, may cause or significantly contribute to an increase in mortality or an increase in serious irreversible or incapacitating reversible illness, or pose a substantial present or potential hazard to human health.

- 5. "Age" means sixty years old or more.
- 6. "Disability" means a physical or mental impairment that substantially limits a major life activity.
- § 240.05 Riot in the second degree
- § 240.06 Riot in the first degree
- § 240.08 Inciting to riot
- § 240.10 Unlawful assembly
- § 240.15 Criminal anarchy
- § 240.20 Disorderly conduct

A person is guilty of disorderly conduct when, with intent to cause public inconvenience, annoyance or alarm, or recklessly creating a risk thereof:

- 1. He engages in fighting or in violent, tumultuous or threatening behavior; or
 - 2. He makes unreasonable noise; or
- 3. In a public place, he uses abusive or obscene language, or makes an obscene gesture; or
- 4. Without lawful authority, he disturbs any lawful assembly or meeting of persons; or
 - 5. He obstructs vehicular or pedestrian traffic; or
- 6. He congregates with other persons in a public place and refuses to comply with a lawful order of the police to disperse; or
- 7. He creates a hazardous or physically offensive condition by any act which serves no legitimate purpose.

Disorderly conduct is a violation.

§ 240.21 Disruption or disturbance of a religious service, funeral, burial or memorial service

A person is guilty of disruption or disturbance of a religious service, funeral, burial or memorial service when he or she makes unreasonable noise or disturbance while at a lawfully assembled religious service, funeral, burial or memorial service, or within three hundred feet thereof, with intent to cause annoyance or alarm or recklessly creating a risk thereof.

Disruption or disturbance of a religious service, funeral, burial or memorial service is a class A misdemeanor.

§ 240.25 Harassment in the first degree

A person is guilty of harassment in the first degree when he or she intentionally and repeatedly harasses another person by following such person in or about a public place or places or by engaging in a course of conduct or by repeatedly committing acts which places such person in reasonable fear of physical injury. This section shall not apply to activities regulated by the national labor relations act, [FN1] as amended, the railway labor act, [FN2] as amended, or the federal employment labor management act, [FN3] as amended.

Harassment in the first degree is a class B misdemeanor.

[FN1] 29 USC § 151 et seq.

[FN2] 45 USC § 151 et seq.

[FN3] So in original. Probably refers to the Federal Service Labor-Management Relations Act, 5 USC § 7101 et seq.

§ 240.26 Harassment in the second degree

A person is guilty of harassment in the second degree when, with intent to harass, annoy or alarm another person:

- 1. He or she strikes, shoves, kicks or otherwise subjects such other person to physical contact, or attempts or threatens to do the same; or
- 2. He or she follows a person in or about a public place or places; or
- 3. He or she engages in a course of conduct or repeatedly commits acts which alarm or seriously annoy such other person and which serve no legitimate purpose.

Subdivisions two and three of this section shall not apply to activities regulated by the national labor relations act, [FN1] as amended, the railway labor act, [FN2] as amended, or the federal employment labor management act, [FN3] as amended.

Harassment in the second degree is a violation.

[FN1] 29 USCA § 151 et seq.

[FN2] 45 USCA § 151 et seq.

[FN3] So in original. Probably refers to the Federal Service Labor-Management Relations Act, 5 USCA § 7101 et seq.

§ 240.30 Aggravated harassment in the second degree

§ 240.31 Aggravated harassment in the first degree

§ 240.32 Aggravated harassment of an employee by an inmate

§ 240.35 Loitering

A person is guilty of loitering when he:

- 1. Repealed by L.2010, c. 232, § 1, eff. July 30, 2010.
- 2. Loiters or remains in a public place for the purpose of gambling with cards, dice or other gambling paraphernalia; or
- 3. Repealed by L.2010, c. 232, § 1, eff. July 30, 2010.
- 4. Being masked or in any manner disguised by unusual or unnatural attire or facial alteration, loiters, remains or congregates in a public place with other persons so masked or disguised, or knowingly permits or aids persons so masked or disguised to congregate in a public place; except that such conduct is not unlawful when it occurs in connection with a masquerade party or like entertainment if, when such entertainment is held in a city which has promulgated regulations in connection with such affairs, permission is first obtained from the police or other appropriate authorities; or
- 5. Loiters or remains in or about school grounds, a college or university building or grounds or a children's overnight camp as defined in section one thousand three hundred ninety-two of the public health law or a summer day camp as defined in section one thousand three hundred ninety-two of the public health law, or loiters, remains in or enters a school bus as defined in section one hundred forty-two of the vehicle and traffic law, not having any reason or relationship involving custody of or responsibility for a pupil or student, or any other specific, legitimate reason for being there, and not having written permission from anyone authorized to grant the same or loiters or remains in or about such children's overnight camp or summer day camp in violation of conspicuously posted rules or regulations governing entry and use thereof; or
- 6. Loiters or remains in any transportation facility, unless specifically authorized to do so, for the purpose of soliciting or engaging in any business, trade or commercial transactions involving the sale of merchandise or services, or for the purpose of

entertaining persons by singing, dancing or playing any musical instrument; or

7. Repealed by L.2010, c. 232, § 1, eff. July 30, 2010.

Loitering is a violation.

§ 240.36 Loitering in the first degree

A person is guilty of loitering in the first degree when he loiters or remains in any place with one or more persons for the purpose of unlawfully using or possessing a controlled substance, as defined in section 220.00 of this chapter.

Loitering in the first degree is a class B misdemeanor.

§ 240.37 Loitering for the purpose of engaging in a prostitution offense

- 1. For the purposes of this section, "public place" means any street, sidewalk, bridge, alley or alleyway, plaza, park, driveway, parking lot or transportation facility or the doorways and entrance ways to any building which fronts on any of the aforesaid places, or a motor vehicle in or on any such place.
- 2. Any person who remains or wanders about in a public place and repeatedly beckons to, or repeatedly stops, or repeatedly attempts to stop, or repeatedly attempts to engage passers-by in conversation, or repeatedly stops or attempts to stop motor vehicles, or repeatedly interferes with the free passage of other persons, for the purpose of prostitution, or of patronizing a prostitute as those terms are defined in article two hundred thirty of the penal law, shall be guilty of a violation and is guilty of a class B misdemeanor if such person has previously been convicted of a violation of this section or of sections 230.00 or 230.05 of the penal law.
- 3. Any person who remains or wanders about in a public place and repeatedly beckons to, or repeatedly stops, or repeatedly attempts to stop, or repeatedly attempts to engage passers-by in conversation, or repeatedly stops or attempts to stop motor vehicles, or repeatedly interferes with the free passage of other persons, for the purpose of promoting prostitution as defined in article two hundred thirty of the penal law is guilty of a class A misdemeanor.

§ 240.40 Appearance in public under the influence of narcotics or a drug other than alcohol

- § 240.45 Criminal nuisance in the second degree
- § 240.46 Criminal nuisance in the first degree
- § 240.48 Disseminating a false registered sex offender notice
- § 240.50 Falsely reporting an incident in the third degree
- § 240.55 Falsely reporting an incident in the second degree
- § 240.60 Falsely reporting an incident in the first degree
- § 240.61 Placing a false bomb or hazardous substance in the second degree
- § 240.62 Placing a false bomb or hazardous substance in the first degree
- § 240.63 Placing a false bomb or hazardous substance in a sports stadium or arena, mass transportation facility or enclosed shopping mall
- § 240.65 Unlawful prevention of public access to records
- § 240.70 Criminal interference with health care services or religious worship in the second degree
- § 240.71 Criminal interference with health care services or religious worship in the first degree
- § 240.72 Aggravated interference with health care services in the second degree
- § 240.73 Aggravated interference with health care services in the first degree

Article 241. Harassment of Rent Regulated Tenants

§ 241.00 Harassment of a rent regulated tenant; definition of terms

Article 242. Offenses Against Service Animals and Handlers

- § 242.00 Definitions
- § 242.05 Interference, harassment or intimidation of a service animal

- § 242.10 Harming a service animal in the second degree
- § 242.15 Harming a service animal in the first degree

Article 245. Offenses Against Public Sensibilities

- § 245.00 Public lewdness
- § 245.01 Exposure of a person
- § 245.02 Promoting the exposure of a person
- § 245.05 Offensive exhibition
- § 245.10 Public display of offensive sexual material; definitions of terms
- § 245.11 Public display of offensive sexual material

Article 250. Offenses Against the Right to Privacy

- § 250.00 Eavesdropping; definitions of terms
- § 250.05 Eavesdropping
- § 250.10 Possession of eavesdropping devices
- § 250.15 Failure to report wiretapping
- § 250.20 Divulging an eavesdropping warrant
- § 250.25 Tampering with private communications
- § 250.30 Unlawfully obtaining communications information
- § 250.35 Failing to report criminal communications
- § 250.40 Unlawful surveillance; definitions
- § 250.45 Unlawful surveillance in the second degree
- § 250.50 Unlawful surveillance in the first degree
- § 250.55 Dissemination of an unlawful surveillance image in the second degree
- § 250.60 Dissemination of an unlawful surveillance image in the first degree
- § 250.65 Additional provisions
- 1. The provisions of sections 250.45, 250.50, 250.55 and 250.60 of this article do not apply with

respect to any: (a) law enforcement personnel engaged in the conduct of their authorized duties; (b) security system wherein a written notice is conspicuously posted on the premises stating that a video surveillance system has been installed for the purpose of security; or (c) video surveillance devices installed in such a manner that their presence is clearly and immediately obvious.

2. With respect to sections 250.55 and 250.60 of this article, the provisions of subdivision two of section 235.15 and subdivisions one and two of section 235.24 of this chapter shall apply.

Title O. Offenses Against Marriage, the Family, and the Welfare of Children and Incompetents

Article 255. Offenses Affecting the Marital Relationship

- § 255.00 Unlawfully solemnizing a marriage
- § 255.05 Unlawfully issuing a dissolution decree
- § 255.10 Unlawfully procuring a marriage license
- § 255.15 Bigamy
- § 255.17 Adultery
- § 255.20 Unlawfully procuring a marriage license, bigamy, adultery: defense
- § 255.25 Incest in the third degree
- § 255.26. Incest in the second degree
- § 255.27. Incest in the first degree
- § 255.30 Adultery and incest; corroboration

Article 260. Offenses Relating to Children, Disabled Persons and Vulnerable Elderly Persons

- § 260.00 Abandonment of a child
- § 260.05 Non-support of a child in the second degree
- § 260.06 Non-support of a child in the first degree
- § 260.10 Endangering the welfare of a child

A person is guilty of endangering the welfare of a child when:

- 1. He or she knowingly acts in a manner likely to be injurious to the physical, mental or moral welfare of a child less than seventeen years old or directs or authorizes such child to engage in an occupation involving a substantial risk of danger to his or her life or health; or
- 2. Being a parent, guardian or other person legally charged with the care or custody of a child less than eighteen years old, he or she fails or refuses to exercise reasonable diligence in the control of such child to prevent him or her from becoming an "abused child," a "neglected child," a "juvenile delinquent" or a "person in need of supervision," as those terms are defined in articles ten, three and seven of the family court act.
- 3. A person is not guilty of the provisions of this section when he or she engages in the conduct described in subdivision one of section 260.00 of this article: (a) with the intent to wholly abandon the child by relinquishing responsibility for and right to the care and custody of such child; (b) with the intent that the child be safe from physical injury and cared for in an appropriate manner; (c) the child is left with an appropriate person, or in a suitable location and the person who leaves the child promptly notifies an appropriate person of the child's location; and (d) the child is not more than thirty days old.

Endangering the welfare of a child is a class A misdemeanor.

§ 260.11 Endangering the welfare of a child; corroboration

A person shall not be convicted of endangering the welfare of a child, or of an attempt to commit the same, upon the testimony of a victim who is incapable of consent because of mental defect or mental incapacity as to conduct that constitutes an offense or an attempt to commit an offense referred to in section 130.16, without additional evidence sufficient pursuant to section 130.16 to sustain a conviction of an offense referred to in section 130.16, or of an attempt to commit the same.

§ 260.15 Endangering the welfare of a child; defense

In any prosecution for endangering the welfare of a child, pursuant to section 260.10 of this article, based upon an alleged failure or refusal to provide proper medical care or treatment to an ill child, it is an affirmative defense that the defendant (a) is a parent, guardian or other person legally charged with the care or custody of such child; and (b) is a member or adherent of an organized church or religious group the tenets of which prescribe prayer as the principal treatment for illness; and (c) treated or caused such ill child to be treated in accordance with such tenets.

§ 260.20 Unlawfully dealing with a child in the first degree

§ 260.21 Unlawfully dealing with a child in the second degree

§ 260.25 Endangering the welfare of an incompetent or physically disabled person

§ 260.31 Vulnerable elderly persons; definitions

§ 260.31 Misrepresentation by a child day care provider

§ 260.32. Endangering the welfare of a vulnerable elderly person, or an incompetent or physically disabled person in the second degree

§ 260.34. Endangering the welfare of a vulnerable elderly person, or an incompetent or physically disabled person in the first degree

Article 263. Sexual Performance by a Child

§ 263.00 Definitions

§ 263.05 Use of a child in a sexual performance

§ 263.10 Promoting an obscene sexual performance by a child

§ 263.11 Possessing an obscene sexual performance by a child

§ 263.15 Promoting a sexual performance by a child

§ 263.16 Possessing a sexual performance by a child

§ 263.20 Sexual performance by a child; affirmative defenses

§ 263.25 Proof of age of child

§ 263.30 Facilitating a sexual performance by a child with a controlled substance or alcohol.

TITLE P. OFFENSES AGAINST PUBLIC SAFETY

Article 265. Firearms and Other Dangerous Weapons

§ 265.00 Definitions

As used in this article and in article four hundred, the following terms shall mean and include:

- 1. "Machine-gun" means a weapon of any description, irrespective of size, by whatever name known, loaded or unloaded, from which a number of shots or bullets may be rapidly or automatically discharged from a magazine with one continuous pull of the trigger and includes a sub-machine gun.
- 2. "Firearm silencer" means any instrument, attachment, weapon or appliance for causing the firing of any gun, revolver, pistol or other firearms to be silent, or intended to lessen or muffle the noise of the firing of any gun, revolver, pistol or other firearms.
- 3. "Firearm" means (a) any pistol or revolver; or (b) a shotgun having one or more barrels less than eighteen inches in length; or (c) a rifle having one or more barrels less than sixteen inches in length; or (d) any weapon made from a shotgun or rifle whether by alteration, modification, or otherwise if such weapon as altered, modified, or otherwise has an overall length of less than twenty-six inches; or (e) an assault weapon. For the purpose of this subdivision the length of the barrel on a shotgun or rifle shall be determined by measuring the distance between the muzzle and the face of the bolt. breech, or breechlock when closed and when the shotgun or rifle is cocked; the overall length of a weapon made from a shotgun or rifle is the distance between the extreme ends of the weapon measured along a line parallel to the center line of the bore. Firearm does not include an antique firearm.
- 4. "Switchblade knife" means any knife which has a blade which opens automatically by hand pressure applied to a button, spring or other device in the handle of the knife.
- 5. "Gravity knife" means any knife which has a blade which is released from the handle or sheath thereof by the force of gravity or the application of centrifugal force which, when released, is locked in place by means of a button, spring, lever or other device.

- 5-a. "Pilum ballistic knife" means any knife which has a blade which can be projected from the handle by hand pressure applied to a button, lever, spring or other device in the handle of the knife.
- 5-b. "Metal knuckle knife" means a weapon that, when closed, cannot function as a set of plastic knuckles or metal knuckles, nor as a knife and when open, can function as both a set of plastic knuckles or metal knuckles as well as a knife.
- 5-c. "Automatic knife" includes a stiletto, a switchblade knife, a gravity knife, a cane sword, a pilum ballistic knife, and a metal knuckle knife.
- 6. "Dispose of" means to dispose of, give, give away, lease-loan, keep for sale, offer, offer for sale, sell, transfer and otherwise dispose of.
- 7. "Deface" means to remove, deface, cover, alter or destroy the manufacturer's serial number or any other distinguishing number or identification mark.
- 8. "Gunsmith" means any person, firm, partnership, corporation or company who engages in the business of repairing, altering, assembling, manufacturing, cleaning, polishing, engraving or trueing, or who performs any mechanical operation on, any firearm, large capacity ammunition feeding device or machine-qun.
- 9. "Dealer in firearms" means any person, firm, partnership, corporation or company who engages in the business of purchasing, selling, keeping for sale, loaning, leasing, or in any manner disposing of, any assault weapon, large capacity ammunition feeding device, pistol or revolver.
- 10. "Licensing officer" means in the city of New York the police commissioner of that city; in the county of Nassau the commissioner of police of that county; in the county of Suffolk the sheriff of that county except in the towns of Babylon, Brookhaven, Huntington, Islip and Smithtown, the commissioner of police of that county; for the purposes of section 400.01 of this chapter the superintendent of state police; and elsewhere in the state a judge or justice of a court of record having his office in the county of issuance.
- 11. "Rifle" means a weapon designed or redesigned, made or remade, and intended to be fired from the shoulder and designed or redesigned and made or remade to use the energy of the explosive in a fixed metallic cartridge to fire only a single projectile through a rifled bore for each single pull of the trigger.

- 12. "Shotgun" means a weapon designed or redesigned, made or remade, and intended to be fired from the shoulder and designed or redesigned and made or remade to use the energy of the explosive in a fixed shotgun shell to fire through a smooth bore either a number of ball shot or a single projectile for each single pull of the trigger.
- 13. "Cane Sword" means a cane or swagger stick having concealed within it a blade that may be used as a sword or stilletto.
- 14. [See also subd. 14 below] "Chuka stick" means any device designed primarily as a weapon, consisting of two or more lengths of a rigid material joined together by a thong, rope or chain in such a manner as to allow free movement of a portion of the device while held in the hand and capable of being rotated in such a manner as to inflict serious injury upon a person by striking or choking. These devices are also known as nunchakus and centrifugal force sticks.
- 14. [See also subd. 14 above] "Antique firearm" means:

Any unloaded muzzle loading pistol or revolver with a matchlock, flintlock, percussion cap, or similar type of ignition system, or a pistol or revolver which uses fixed cartridges which are no longer available in the ordinary channels of commercial trade.

- 15. "Loaded firearm" means any firearm loaded with ammunition or any firearm which is possessed by one who, at the same time, possesses a quantity of ammunition which may be used to discharge such firearm.
- 15-a. "Electronic dart gun" means any device designed primarily as a weapon, the purpose of which is to momentarily stun, knock out or paralyze a person by passing an electrical shock to such person by means of a dart or projectile.
- 15-b. "Kung Fu star" means a disc-like object with sharpened points on the circumference thereof and is designed for use primarily as a weapon to be thrown.
- 15-c. "Electronic stun gun" means any device designed primarily as a weapon, the purpose of which is to stun, cause mental disorientation, knock out or paralyze a person by passing a high voltage electrical shock to such person.
- 16. "Certified not suitable to possess a self-defense spray device, a rifle or shotgun"

- means that the director or physician in charge of any hospital or institution for mental illness, public or private, has certified to the superintendent of state police or to any organized police department of a county, city, town or village of this state, that a person who has been judicially adjudicated incompetent, or who has been confined to such institution for mental illness pursuant to judicial authority, is not suitable to possess a self-defense spray device, as defined in section 265.20 of this article, or a rifle or shotgun.
- 17. "Serious offense" means (a) any of the following offenses defined in the former penal law as in force and effect immediately prior to September first, nineteen hundred sixty-seven: illegally using, carrying or possessing a pistol or other dangerous weapon; making or possessing burglar's instruments; buying or receiving stolen property; unlawful entry of a building; aiding escape from prison; that kind of disorderly conduct defined in subdivisions six and eight of section seven hundred twenty-two of such former penal law; violations of sections four hundred eighty-three, four hundred eighty-three-b, four hundred eighty-four-h and article one hundred six of such former penal law; that kind of criminal sexual act or rape which was designated as a misdemeanor; violation of section seventeen hundred forty-seven-d and seventeen hundred forty-seven-e of such former penal law; any violation of any provision of article thirty-three of the public health law relating to narcotic drugs which was defined as a misdemeanor by section seventeen hundred fifty-one-a of such former penal law, and any violation of any provision of article thirty-three-A of the public health law relating to depressant and stimulant drugs which was defined as a misdemeanor by section seventeen hundred forty-seven-b of such former penal law.
- (b) [As amended by L.1999, c. 635, § 11. See, also, par. (b) below.] any of the following offenses defined in the penal law: illegally using, carrying or possessing a pistol or other dangerous weapon; possession of burglar's tools; criminal possession of stolen property in the third degree; escape in the third degree; jostling; fraudulent accosting; endangering the welfare of a child; the offenses defined in article two hundred thirty-five; issuing abortional articles; permitting prostitution; promoting prostitution in the third degree; stalking in the fourth degree; stalking in the third degree; the offenses defined in article one hundred thirty; the offenses defined in article two hundred twenty.

- (b) [As amended by L.1999, c. 635, § 15. See, also, par. (b) above.] any of the following offenses defined in the penal law: illegally using, carrying or possessing a pistol or other dangerous weapon; possession of burglar's tools; criminal possession of stolen property in the third degree; escape in the third degree; jostling; fraudulent accosting; endangering the welfare of a child; the offenses defined in article two hundred thirty-five; issuing abortional articles; permitting prostitution; promoting prostitution in the third degree; stalking in the third degree; stalking in the fourth degree; the offenses defined in article one hundred thirty; the offenses defined in article two hundred twenty.
- 18. "Armor piercing ammunition" means any ammunition capable of being used in pistols or revolvers containing a projectile or projectile core, or a projectile or projectile core for use in such ammunition, that is constructed entirely (excluding the presence of traces of other substances) from one or a combination of any of the following: tungsten alloys, steel, iron, brass, bronze, beryllium copper, or uranium.
- 19. "Duly authorized instructor" means (a) a duly commissioned officer of the United States army, navy, marine corps or coast guard, or of the national guard of the state of New York; or (b) a duly qualified adult citizen of the United States who has been granted a certificate as an instructor in small arms practice issued by the United States army, navy or marine corps, or by the adjutant general of this state, or by the national rifle association of America, a not-for-profit corporation duly organized under the laws of this state; or (c) by a person duly qualified and designated by the department of environmental conservation under paragraph d of subdivision six of section 11-0713 of the environmental conservation law as its agent in the giving of instruction and the making of certifications of qualification in responsible hunting practices.
- 20. "Disguised gun" means any weapon or device capable of being concealed on the person from which a shot can be discharged through the energy of an explosive and is designed and intended to appear to be something other than a gun.
- 21. "Semiautomatic" means any repeating rifle, shotgun or pistol, regardless of barrel or overall length, which utilizes a portion of the energy of a firing cartridge or shell to extract the fired cartridge case or spent shell and chamber the next round, and which requires a separate pull of the trigger to

fire each cartridge or shell.

- 22. "Assault weapon" means (a) a semiautomatic rifle that has an ability to accept a detachable magazine and has at least two of the following characteristics:
 - (i) a folding or telescoping stock;
- (ii) a pistol grip that protrudes conspicuously beneath the action of the weapon;
 - (iii) a bayonet mount;
- (iv) a flash suppressor or threaded barrel designed to accommodate a flash suppressor;
 - (v) a grenade launcher; or
- (b) a semiautomatic shotgun that has at least two of the following characteristics:
 - (i) a folding or telescoping stock;
- (ii) a pistol grip that protrudes conspicuously beneath the action of the weapon;
- (iii) a fixed magazine capacity in excess of five rounds;
 - (iv) an ability to accept a detachable magazine; or
- (c) a semiautomatic pistol that has an ability to accept a detachable magazine and has at least two of the following characteristics:
- (i) an ammunition magazine that attaches to the pistol outside of the pistol grip;
- (ii) a threaded barrel capable of accepting a barrel extender, flash suppressor, forward handgrip, or silencer;
- (iii) a shroud that is attached to, or partially or completely encircles, the barrel and that permits the shooter to hold the firearm with the nontrigger hand without being burned;
- (iv) a manufactured weight of fifty ounces or more when the pistol is unloaded;
- (v) a semiautomatic version of an automatic rifle, shotgun or firearm; or
- (d) any of the weapons, or functioning frames or receivers of such weapons, or copies or duplicates of such weapons, in any caliber, known as:
- (i) Norinco, Mitchell, and Poly Technologies Avtomat Kalashnikovs (all models);
- (ii) Action Arms Israeli Military Industries UZI and Galil;

- (iii) Beretta Ar70 (SC-70);
- (iv) Colt AR-15;
- (v) Fabrique National FN/FAL, FN/LAR, and FNC;
- (vi) SWD M-10, M-11, M-11/9, and M-12;
- (vii) Steyr AUG;
- (viii) INTRATEC TEC-9, TEC-DC9 and TEC-22;
- (ix) revolving cylinder shotguns, such as (or similar to) the Street Sweeper and Striker 12;
- (e) provided, however, that such term does not include: (i) any rifle, shotgun or pistol that (A) is manually operated by bolt, pump, lever or slide action; (B) has been rendered permanently inoperable; or (C) is an antique firearm as defined in 18 U.S.C. 921(a)(16);
- (ii) a semiautomatic rifle that cannot accept a detachable magazine that holds more than five rounds of ammunition;
- (iii) a semiautomatic shotgun that cannot hold more than five rounds of ammunition in a fixed or detachable magazine;
- (iv) a rifle, shotgun or pistol, or a replica or a duplicate thereof, specified in Appendix A to section 922 of 18 U.S.C. as such weapon was manufactured on October first, nineteen hundred ninety-three. The mere fact that a weapon is not listed in Appendix A shall not be construed to mean that such weapon is an assault weapon; or
- (v) a semiautomatic rifle, a semiautomatic shotgun or a semiautomatic pistol or any of the weapons defined in paragraph (d) of this subdivision lawfully possessed prior to September fourteenth, nineteen hundred ninety-four.
- 23. "Large capacity ammunition feeding device" means a magazine, belt, drum, feed strip, or similar device, manufactured after September thirteenth, nineteen hundred ninety-four, that has a capacity of, or that can be readily restored or converted to accept, more than ten rounds of ammunition; provided, however, that such term does not include an attached tubular device designed to accept, and capable of operating only with, .22 caliber rimfire ammunition.

§ 265.01 Criminal possession of a weapon in the fourth degree

A person is guilty of criminal possession of a weapon in the fourth degree when:

- (1) He or she possesses any firearm, electronic dart gun, electronic stun gun, gravity knife, switchblade knife, pilum ballistic knife, metal knuckle knife, cane sword, billy, blackjack, bludgeon, plastic knuckles, metal knuckles, chuka stick, sand bag, sandclub, wrist-brace type slingshot or slungshot, shirken or "Kung Fu star"; or
- (2) He possesses any dagger, dangerous knife, dirk, razor, stiletto, imitation pistol, or any other dangerous or deadly instrument or weapon with intent to use the same unlawfully against another; or
- (3) He or she knowingly has in his or her possession a rifle, shotgun or firearm in or upon a building or grounds, used for educational purposes, of any school, college or university, except the forestry lands, wherever located, owned and maintained by the State University of New York college of environmental science and forestry, or upon a school bus as defined in section one hundred forty-two of the vehicle and traffic law, without the written authorization of such educational institution; or
- (4) He possesses a rifle, shotgun, antique firearm, black powder rifle, black powder shotgun, or any muzzle-loading firearm, and has been convicted of a felony or serious offense; or
- (5) He possesses any dangerous or deadly weapon and is not a citizen of the United States; or
- (6) He is a person who has been certified not suitable to possess a rifle or shotgun, as defined in subdivision sixteen of section 265.00, and refuses to yield possession of such rifle or shotgun upon the demand of a police officer. Whenever a person is certified not suitable to possess a rifle or shotgun, a member of the police department to which such certification is made, or of the state police, shall forthwith seize any rifle or shotgun possessed by such person. A rifle or shotgun seized as herein provided shall not be destroyed, but shall be delivered to the headquarters of such police department, or state police, and there retained until the aforesaid certificate has been rescinded by the director or physician in charge, or other disposition of such rifle or shotgun has been ordered or authorized by a court of competent jurisdiction.
- (7) He knowingly possesses a bullet containing an explosive substance designed to detonate upon impact.

(8) He possesses any armor piercing ammunition with intent to use the same unlawfully against another.

Criminal possession of a weapon in the fourth degree is a class A misdemeanor.

§ 265.02 Criminal possession of a weapon in the third degree

A person is guilty of criminal possession of a weapon in the third degree when:

- (1) Such person commits the crime of criminal possession of a weapon in the fourth degree as defined in subdivision one, two, three or five of section 265.01, and has been previously convicted of any crime; or
- (2) Such person possesses any explosive or incendiary bomb, bombshell, firearm silencer, machine-gun or any other firearm or weapon simulating a machine-gun and which is adaptable for such use; or
- (3) Such person knowingly possesses a machine-gun, firearm, rifle or shotgun which has been defaced for the purpose of concealment or prevention of the detection of a crime or misrepresenting the identity of such machine-gun, firearm, rifle or shotgun; or
- (4) Repealed by L.2006, c. 742, § 1, eff. Nov. 1, 2006.
- (5) (i) Such person possesses three or more firearms; or (ii) such person possesses a firearm and has been previously convicted of a felony or a class A misdemeanor defined in this chapter within the five years immediately preceding the commission of the offense and such possession did not take place in the person's home or place of business: or
- (6) Such person knowingly possesses any disguised gun; or
- (7) Such person possesses an assault weapon; or
- (8) Such person possesses a large capacity ammunition feeding device.

Criminal possession of a weapon in the third degree is a class D felony.

§ 265.03 Criminal possession of a weapon in the second degree

A person is guilty of criminal possession of a weapon in the second degree when:

- (1) with intent to use the same unlawfully against another, such person:
 - (a) possesses a machine-gun; or
 - (b) possesses a loaded firearm; or
 - (c) possesses a disguised gun; or
- (2) such person possesses five or more firearms; or
- (3) such person possesses any loaded firearm. Such possession shall not, except as provided in subdivision one or seven of section 265.02 of this article, constitute a violation of this subdivision if such possession takes place in such person's home or place of business.

Criminal possession of a weapon in the second degree is a class C felony.

§ 265.04 Criminal possession of a weapon in the first degree

A person is guilty of criminal possession of a weapon in the first degree when such person:

- (1) possesses any explosive substance with intent to use the same unlawfully against the person or property of another; or
 - (2) possesses ten or more firearms.

Criminal possession of a weapon in the first degree is a class B felony.

§ 265.05 Unlawful possession of weapons by persons under sixteen

It shall be unlawful for any person under the age of sixteen to possess any air-gun, spring-gun or other instrument or weapon in which the propelling force is a spring or air, or any gun or any instrument or weapon in or upon which any loaded or blank cartridges may be used, or any loaded or blank cartridges or ammunition therefor, or any dangerous knife; provided that the possession of rifle or shotgun or ammunition therefor by the holder of a hunting license or permit issued pursuant to article eleven of the environmental conservation law and used in accordance with said law shall not be governed by this section.

A person who violates the provisions of this section shall be adjudged a juvenile delinquent.

§ 265.06 Unlawful possession of a weapon upon school grounds

It shall be unlawful for any person age sixteen or older to knowingly possess any air-gun, spring-gun

or other instrument or weapon in which the propelling force is a spring, air, piston or CO2 cartridge in or upon a building or grounds, used for educational purposes, of any school, college or university, without the written authorization of such educational institution. Unlawful possession of a weapon upon school grounds is a violation.

§ 265.08 Criminal use of a firearm in the second degree

A person is guilty of criminal use of a firearm in the second degree when he commits any class C violent felony offense as defined in paragraph (b) of subdivision one of section 70.02 and he either:

- (1) possesses a deadly weapon, if the weapon is a loaded weapon from which a shot, readily capable of producing death or other serious injury may be discharged; or
- (2) displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm.

Criminal use of a firearm in the second degree is a class C felony.

§ 265.09 Criminal use of a firearm in the first degree

- (1) A person is guilty of criminal use of a firearm in the first degree when he commits any class B violent felony offense as defined in paragraph (a) of subdivision one of section 70.02 and he either:
- (a) possesses a deadly weapon, if the weapon is a loaded weapon from which a shot, readily capable of producing death or other serious injury may be discharged; or
- (b) displays what appears to be a pistol, revolver, rifle, shotgun, machine gun or other firearm.

Criminal use of a firearm in the first degree is a class B felony.

(2) Sentencing. Notwithstanding any other provision of law to the contrary, when a person is convicted of criminal use of a firearm in the first degree as defined in subdivision one of this section, the court shall impose an additional consecutive sentence of five years to the minimum term of an indeterminate sentence imposed on the underlying class B violent felony offense where the person convicted of such crime displays a loaded weapon from which a shot, readily capable of producing death or other serious injury may be discharged, in furtherance of the commission of such crime, provided, however, that such additional

sentence shall not be imposed if the court, having regard to the nature and circumstances of the crime and to the history and character of the defendant, finds on the record that such additional consecutive sentence would be unduly harsh and that not imposing such sentence would be consistent with the public safety and would not deprecate the seriousness of the crime. Notwithstanding any other provision of law to the contrary, the aggregate of the five year consecutive term imposed pursuant to this subdivision and the minimum term of the indeterminate sentence imposed on the underlying class B violent felony shall constitute the new aggregate minimum term of imprisonment, and a person subject to such term shall be required to serve the entire aggregate minimum term and shall not be eligible for release on parole or conditional release during such term. This subdivision shall not apply where the defendant's criminal liability for displaying a loaded weapon from which a shot, readily capable of producing death or other serious injury may be discharged, in furtherance of the commission of crime is based on the conduct of another pursuant to section 20.00 of the penal law.

§ 265.10 Manufacture, transport, disposition and defacement of weapons and dangerous instruments and appliances

- 1. Any person who manufactures or causes to be manufactured any machine-gun, assault weapon, large capacity ammunition feeding device or disguised gun is guilty of a class D felony. Any person who manufactures or causes to be manufactured any switchblade knife, gravity knife, pilum ballistic knife, metal knuckle knife, billy, blackjack, bludgeon, plastic knuckles, metal knuckles, Kung Fu star, chuka stick, sandbag, sandclub or slungshot is guilty of a class A misdemeanor.
- 2. Any person who transports or ships any machine-gun, firearm silencer, assault weapon or large capacity ammunition feeding device or disguised gun, or who transports or ships as merchandise five or more firearms, is guilty of a class D felony. Any person who transports or ships as merchandise any firearm, other than an assault weapon, switchblade knife, gravity knife, pilum ballistic knife, billy, blackjack, bludgeon, plastic knuckles, metal knuckles, Kung Fu star, chuka stick, sandbag or slungshot is guilty of a class A misdemeanor.
 - 3. Any person who disposes of any machine-gun,

- assault weapon, large capacity ammunition feeding device or firearm silencer is guilty of a class D felony. Any person who knowingly buys, receives, disposes of, or conceals a machine-gun, firearm, large capacity ammunition feeding device, rifle or shotgun which has been defaced for the purpose of concealment or prevention of the detection of a crime or misrepresenting the identity of such machine-gun, firearm, large capacity ammunition feeding device, rifle or shotgun is guilty of a class D felony.
- 4. Any person who disposes of any of the weapons, instruments or appliances specified in subdivision one of section 265.01, except a firearm, is guilty of a class A misdemeanor, and he is guilty of a class D felony if he has previously been convicted of any crime.
- 5. Any person who disposes of any of the weapons, instruments, appliances or substances specified in section 265.05 to any other person under the age of sixteen years is guilty of a class A misdemeanor.
- 6. Any person who wilfully defaces any machine-gun, large capacity ammunition feeding device or firearm is guilty of a class D felony.
- 7. Any person, other than a wholesale dealer, or gunsmith or dealer in firearms duly licensed pursuant to section 400.00, lawfully in possession of a firearm, who disposes of the same without first notifying in writing the licensing officer in the city of New York and counties of Nassau and Suffolk and elsewhere in the state the executive department, division of state police, Albany, is guilty of a class A misdemeanor.

§ 265.11 Criminal sale of a firearm in the third degree

A person is guilty of criminal sale of a firearm in the third degree when such person is not authorized pursuant to law to possess a firearm and such person unlawfully either:

- (1) sells, exchanges, gives or disposes of a firearm or large capacity ammunition feeding device to another person; or
 - (2) possesses a firearm with the intent to sell it.

Criminal sale of a firearm in the third degree is a class D felony.

§ 265.12 Criminal sale of a firearm in the second degree

A person is guilty of criminal sale of a firearm in the second degree when such person:

- (1) unlawfully sells, exchanges, gives or disposes of to another five or more firearms; or
- (2) unlawfully sells, exchanges, gives or disposes of to another person or persons a total of five or more firearms in a period of not more than one year.

Criminal sale of a firearm in the second degree is a class C felony.

§ 265.13 Criminal sale of a firearm in the first degree

A person is guilty of criminal sale of a firearm in the first degree when such person:

- (1) unlawfully sells, exchanges, gives or disposes of to another ten or more firearms; or
- (2) unlawfully sells, exchanges, gives or disposes of to another person or persons a total of ten or more firearms in a period of not more than one year.

Criminal sale of a firearm in the first degree is a class B felony.

§ 265.14 Criminal sale of a firearm with the aid of a minor

A person over the age of eighteen years of age is guilty of criminal sale of a weapon with the aid of a minor when a person under sixteen years of age knowingly and unlawfully sells, exchanges, gives or disposes of a firearm in violation of this article, and such person over the age of eighteen years of age, acting with the mental culpability required for the commission thereof, solicits, requests, commands, importunes or intentionally aids such person under sixteen years of age to engage in such conduct.

Criminal sale of a firearm with the aid of a minor is a class C felony.

§ 265.15 Presumptions of possession, unlawful intent and defacement

- 1. The presence in any room, dwelling, structure or vehicle of any machine-gun is presumptive evidence of its unlawful possession by all persons occupying the place where such machine-gun is found.
- 2. The presence in any stolen vehicle of any weapon, instrument, appliance or substance specified in sections 265.01, 265.02, 265.03, 265.04 and 265.05 is presumptive evidence of its

possession by all persons occupying such vehicle at the time such weapon, instrument, appliance or substance is found.

- 3. The presence in an automobile, other than a stolen one or a public omnibus, of any firearm, large capacity ammunition feeding device, defaced firearm, defaced rifle or shotgun, defaced large capacity ammunition feeding device, firearm silencer, explosive or incendiary bomb, bombshell, gravity knife, switchblade knife, pilum ballistic knife, metal knuckle knife, dagger, dirk, stiletto, billy, blackjack, plastic knuckles, metal knuckles, chuka stick, sandbag, sandclub or slungshot is presumptive evidence of its possession by all persons occupying such automobile at the time such weapon, instrument or appliance is found, except under the following circumstances: (a) if such weapon, instrument or appliance is found upon the person of one of the occupants therein; (b) if such weapon, instrument or appliance is found in an automobile which is being operated for hire by a duly licensed driver in the due, lawful and proper pursuit of his or her trade, then such presumption shall not apply to the driver; or (c) if the weapon so found is a pistol or revolver and one of the occupants, not present under duress, has in his or her possession a valid license to have and carry concealed the same.
- 4. The possession by any person of the substance as specified in section 265.04 is presumptive evidence of possessing such substance with intent to use the same unlawfully against the person or property of another if such person is not licensed or otherwise authorized to possess such substance. The possession by any person of any dagger, dirk, stiletto, dangerous knife or any other weapon, instrument, appliance or substance designed, made or adapted for use primarily as a weapon, is presumptive evidence of intent to use the same unlawfully against another.
- 5. The possession by any person of a defaced machine-gun, firearm, rifle or shotgun is presumptive evidence that such person defaced the same.
- 6. The possession of five or more firearms by any person is presumptive evidence that such person possessed the firearms with the intent to sell same.

§ 265.16 Criminal sale of a firearm to a minor

A person is guilty of criminal sale of a firearm to a minor when he is not authorized pursuant to law to possess a firearm and he unlawfully sells, exchanges, gives or disposes of a firearm to another person who is or reasonably appears to be less than nineteen years of age who is not licensed pursuant to law to possess a firearm.

Criminal sale of a firearm to a minor is a class C felony.

§ 265.17 Criminal purchase of a weapon

A person is guilty of criminal purchase of a weapon when:

- 1. Knowing that he or she is prohibited by law from possessing a firearm, rifle or shotgun because of a prior conviction or because of some other disability which would render him or her ineligible to lawfully possess a firearm, rifle or shotgun in this state, such person attempts to purchase a firearm, rifle or shotgun from another person; or
- 2. Knowing that it would be unlawful for another person to possess a firearm, rifle or shotgun, he or she purchases a firearm, rifle or shotgun for, on behalf of, or for the use of such other person.

Criminal purchase of a weapon is a class A misdemeanor.

§ 265.20 Exemptions

- a. Sections 265.01, 265.02, 265.03, 265.04, 265.05, 265.10, 265.11, 265.12, 265.13, 265.15 and 270.05 shall not apply to:
- 1. Possession of any of the weapons, instruments, appliances or substances specified in sections 265.01, 265.02, 265.03, 265.04, 265.05 and 270.05 by the following:
- (a) Persons in the military service of the state of New York when duly authorized by regulations issued by the adjutant general to possess the same.
- (b) Police officers as defined in subdivision thirty-four of section 1.20 of the criminal procedure law
- (c) Peace officers as defined by section 2.10 of the criminal procedure law.
- (d) Persons in the military or other service of the United States, in pursuit of official duty or when duly authorized by federal law, regulation or order to possess the same.
- (e) Persons employed in fulfilling defense contracts with the government of the United States or agencies thereof when possession of the same

is necessary for manufacture, transport, installation and testing under the requirements of such contract.

- (f) A person voluntarily surrendering such weapon, instrument, appliance or substance, provided that such surrender shall be made to the superintendent of the division of state police or a member thereof designated by such superintendent, or to the sheriff of the county in which such person resides, or in the county of Nassau or in the towns of Babylon, Brookhaven, Huntington, Islip and Smithtown in the county of Suffolk to the commissioner of police or a member of the police department thereof designated by such commissioner, or if such person resides in a city, town other than one named in this subparagraph, or village to the police commissioner or head of the police force or department thereof or to a member of the force or department designated by such commissioner or head; and provided, further, that the same shall be surrendered by such person in accordance with such terms and conditions as may be established by such superintendent, sheriff, police force or department. Nothing in this paragraph shall be construed as granting immunity from prosecution for any crime or offense except that of unlawful possession of such weapons, instruments, appliances or substances surrendered as herein provided. A person who possesses any such weapon, instrument, appliance or substance as an executor or administrator or any other lawful possessor of such property of a decedent may continue to possess such property for a period not over fifteen days. If such property is not lawfully disposed of within such period the possessor shall deliver it to an appropriate official described in this paragraph or such property may be delivered to the superintendent of state police. Such officer shall hold it and shall thereafter deliver it on the written request of such executor, administrator or other lawful possessor of such property to a named person, provided such named person is licensed to or is otherwise lawfully permitted to possess the same. If no request to deliver the property is received by such official within one year of the delivery of such property, such official shall dispose of it in accordance with the provisions of section 400.05 of this chapter.
- 2. Possession of a machine-gun, large capacity ammunition feeding device, firearm, switchblade knife, gravity knife, pilum ballistic knife, billy or blackjack by a warden, superintendent,

headkeeper or deputy of a state prison, penitentiary, workhouse, county jail or other institution for the detention of persons convicted or accused of crime or detained as witnesses in criminal cases, in pursuit of official duty or when duly authorized by regulation or order to possess the same.

- 3. Possession of a pistol or revolver by a person to whom a license therefor has been issued as provided under section 400.00 or 400.01 of this chapter; provided, that such a license shall not preclude a conviction for the offense defined in subdivision three of section 265.01 of this article.
- 4. Possession of a rifle, shotgun or longbow for use while hunting, trapping or fishing, by a person, not a citizen of the United States, carrying a valid license issued pursuant to section 11-0713 of the environmental conservation law.
- 5. Possession of a rifle or shotgun by a person other than a person who has been convicted of a class A-I felony or a violent felony offense, as defined in subdivision one of section 70.02 of this chapter, who has been convicted as specified in subdivision four of section 265.01 to whom a certificate of good conduct has been issued pursuant to section seven hundred three-b of the correction law.
- 6. Possession of a switchblade or gravity knife for use while hunting, trapping or fishing by a person carrying a valid license issued to him pursuant to section 11-0713 of the environmental conservation law.
- 7. Possession, at an indoor or outdoor shooting range for the purpose of loading and firing, of a rifle or shotgun, the propelling force of which is gunpowder by a person under sixteen years of age but not under twelve, under the immediate supervision, guidance and instruction of (a) a duly commissioned officer of the United States army, navy, air force, marine corps or coast guard, or of the national guard of the state of New York; or (b) a duly qualified adult citizen of the United States who has been granted a certificate as an instructor in small arms practice issued by the United States army, navy, air force or marine corps, or by the adjutant general of this state, or by the national rifle association of America, a not-for-profit corporation duly organized under the laws of this state; or (c) a parent, guardian, or a person over the age of eighteen designated in writing by such parent or guardian who shall have a certificate of

- qualification in responsible hunting, including safety, ethics, and landowner relations-hunter relations, issued or honored by the department of environmental conservation; or (d) an agent of the department of environmental conservation appointed to conduct courses in responsible hunting practices pursuant to article eleven of the environmental conservation law.
- 7-a. Possession and use, at an indoor or outdoor pistol range located in or on premises owned or occupied by a duly incorporated organization organized for conservation purposes or to foster proficiency in small arms or at a target pistol shooting competition under the auspices of or approved by the national rifle association for the purpose of loading and firing the same, by a person duly licensed to possess a pistol or revolver pursuant to section 400.00 or 400.01 of this chapter of a pistol or revolver duly so licensed to another person who is present at the time.
- 7-b. Possession and use, at an indoor or outdoor pistol range located in or on premises owned or occupied by a duly incorporated organization organized for conservation purposes or to foster proficiency in small arms or at a target pistol shooting competition under the auspices of or approved by the national rifle association for the purpose of loading and firing the same, by a person who has applied for a license to possess a pistol or revolver and pre-license possession of same pursuant to section 400.00 or 400.01 of this chapter, who has not been previously denied a license, been previously convicted of a felony or serious offense, and who does not appear to be, or pose a threat to be, a danger to himself or to others, and who has been approved for possession and use herein in accordance with section 400.00 or 400.01 of this chapter; provided however, that such possession shall be of a pistol or revolver duly licensed to and shall be used under the supervision, guidance and instruction of, a person specified in paragraph seven of this subdivision and provided further that such possession and use be within the jurisdiction of the licensing officer with whom the person has made application therefor or within the jurisdiction of the superintendent of state police in the case of a retired sworn member of the division of state police who has made an application pursuant to section 400.01 of this chapter.
- 7-c. Possession for the purpose of loading and firing, of a rifle, pistol or shotgun, the propelling

force of which may be either air, compressed gas or springs, by a person under sixteen years of age but not under twelve, under the immediate supervision, guidance and instruction of (a) a duly commissioned officer of the United States army, navy, marine corps or coast guard, or of the national guard of the state of New York; or (b) a duly qualified adult citizen of the United States who has been granted a certificate as an instructor in small arms practice issued by the United States army, navy or marine corps, or by the adjutant general of this state, or by the national rifle association of America, a not-for-profit corporation duly organized under the laws of this state; or (c) a parent, guardian, or a person over the age of eighteen designated in writing by such parent or guardian who shall have a certificate of qualification in responsible hunting, including safety, ethics, and landowner relations-hunter relations, issued or honored by the department of environmental conservation.

7-d. Possession, at an indoor or outdoor shooting range for the purpose of loading and firing, of a rifle, pistol or shotgun, the propelling force of which may be either air, compressed gas or springs, by a person under twelve years of age, under the immediate supervision, guidance and instruction of (a) a duly commissioned officer of the United States army, navy, marine corps or coast guard, or of the national guard of the state of New York; or (b) a duly qualified adult citizen of the United States who has been granted a certificate as an instructor in small arms practice issued by the United States army, navy or marine corps, or by the adjutant general of this state, or by the national rifle association of America, a not-for-profit corporation duly organized under the laws of this state; or (c) a parent, guardian, or a person over the age of eighteen designated in writing by such parent or guardian who shall have a certificate of qualification in responsible hunting, including safety, ethics, and landowner relations-hunter relations, issued or honored by the department of environmental conservation.

7-e. Possession and use of a pistol or revolver, at an indoor or outdoor pistol range located in or on premises owned or occupied by a duly incorporated organization organized for conservation purposes or to foster proficiency in small arms or at a target pistol shooting competition under the auspices of or approved by an association or organization described in paragraph 7-a of this subdivision for the purpose of loading and firing the same by a

person at least fourteen years of age but under the age of twenty-one who has not been previously convicted of a felony or serious offense, and who does not appear to be, or pose a threat to be, a danger to himself or to others; provided however, that such possession shall be of a pistol or revolver duly licensed to and shall be used under the immediate supervision, guidance and instruction of, a person specified in paragraph seven of this subdivision.

- 8. The manufacturer of machine-guns, firearm silencers, assault weapons, large capacity ammunition feeding devices, disguised guns, pilum ballistic knives, switchblade or gravity knives, billies or blackjacks as merchandise, or as a transferee recipient of the same for repair, lawful distribution or research and development, and the disposal and shipment thereof direct to a regularly constituted or appointed state or municipal police department, sheriff, policeman or other peace officer, or to a state prison, penitentiary, workhouse, county jail or other institution for the detention of persons convicted or accused of crime or held as witnesses in criminal cases, or to the military service of this state or of the United States; or for the repair and return of the same to the lawful possessor or for research and development.
- 9. The regular and ordinary transport of firearms as merchandise, provided that the person transporting such firearms, where he knows or has reasonable means of ascertaining what he is transporting, notifies in writing the police commissioner, police chief or other law enforcement officer performing such functions at the place of delivery, of the name and address of the consignee and the place of delivery, and withholds delivery to the consignee for such reasonable period of time designated in writing by such police commissioner, police chief or other law enforcement officer as such official may deem necessary for investigation as to whether the consignee may lawfully receive and possess such firearms.
- 9-a. a. Except as provided in subdivision b hereof, the regular and ordinary transport of pistols or revolvers by a manufacturer of firearms to whom a license as a dealer in firearms has been issued pursuant to section 400.00 of this chapter, or by an agent or employee of such manufacturer of firearms who is otherwise duly licensed to carry a pistol or revolver and who is duly authorized in writing by such manufacturer of firearms to

transport pistols or revolvers on the date or dates specified, directly between places where the manufacturer of firearms regularly conducts business provided such pistols or revolvers are transported unloaded, in a locked opaque container. For purposes of this subdivision, places where the manufacturer of firearms regularly conducts business includes, but is not limited to places where the manufacturer of firearms regularly or customarily conducts development or design of pistols or revolvers, or regularly or customarily conducts tests on pistols or revolvers, or regularly or customarily participates in the exposition of firearms to the public.

- b. The transportation of such pistols or revolvers into, out of or within the city of New York may be done only with the consent of the police commissioner of the city of New York. To obtain such consent, the manufacturer must notify the police commissioner in writing of the name and address of the transporting manufacturer, or agent or employee of the manufacturer who is authorized in writing by such manufacturer to transport pistols or revolvers, the number, make and model number of the firearms to be transported and the place where the manufacturer regularly conducts business within the city of New York and such other information as the commissioner may deem necessary. The manufacturer must not transport such pistols and revolvers between the designated places of business for such reasonable period of time designated in writing by the police commissioner as such official may deem necessary for investigation and to give consent. The police commissioner may not unreasonably withhold his consent.
- 10. Engaging in the business of gunsmith or dealer in firearms by a person to whom a valid license therefor has been issued pursuant to section 400.00.
- 11. Possession of a firearm or large capacity ammunition feeding device by a police officer or sworn peace officer of another state while conducting official business within the state of New York.
- 12. Possession of a pistol or revolver by a person who is a member or coach of an accredited college or university target pistol team while transporting the pistol or revolver into or through New York state to participate in a collegiate, olympic or target pistol shooting competition under the auspices of or approved by the national rifle association, provided

such pistol or revolver is unloaded and carried in a locked carrying case and the ammunition therefor is carried in a separate locked container.

- 12-a. Possession and use of a pistol or revolver, at an indoor or outdoor shooting range, by a registered student of a higher education institution chartered by the state of New York, who is participating in a course in gun safety and proficiency offered by such institution, under the immediate supervision, guidance, and instruction of a person specified in paragraph seven of this subdivision.
- 13. Possession of pistols and revolvers by a person who is a nonresident of this state while attending or traveling to or from, an organized competitive pistol match or league competition under auspices of, or approved by, the National Rifle Association and in which he is a competitor, within forty-eight hours of such event or by a person who is a non-resident of the state while attending or traveling to or from an organized match sanctioned by the International Handgun Metallic Silhouette Association and in which he is a competitor, within forty-eight hours of such event, provided that he has not been previously convicted of a felony or a crime which, if committed in New York, would constitute a felony, and further provided that the pistols or revolvers are transported unloaded in a locked opaque container together with a copy of the match program, match schedule or match registration card. Such documentation shall constitute prima facie evidence of exemption, providing that such person also has in his possession a pistol license or firearms registration card issued in accordance with the laws of his place of residence. For purposes of this subdivision, a person licensed in a jurisdiction which does not authorize such license by a person who has been previously convicted of a felony shall be presumed to have no prior conviction. The superintendent of state police shall annually review the laws of jurisdictions within the United States and Canada with respect to the applicable requirements for licensing or registration of firearms and shall publish a list of those jurisdictions which prohibit possession of a firearm by a person previously convicted of a felony or crimes which if committed in New York state would constitute a felony.
- 13-a. Except in cities not wholly contained within a single county of the state, possession of pistols and revolvers by a person who is a nonresident of

this state while attending or traveling to or from, an organized convention or exhibition for the display of or education about firearms, which is conducted under auspices of, or approved by, the National Rifle Association and in which he is a registered participant, within forty-eight hours of such event, provided that he has not been previously convicted of a felony or a crime which, if committed in New York, would constitute a felony, and further provided that the pistols or revolvers are transported unloaded in a locked opaque container together with a copy of the convention or exhibition program, convention or exhibition schedule or convention or exhibition registration card. Such documentation shall constitute prima facie evidence of exemption, providing that such person also has in his possession a pistol license or firearms registration card issued in accordance with the laws of his place of residence. For purposes of this paragraph, a person licensed in a jurisdiction which does not authorize such license by a person who has been previously convicted of a felony shall be presumed to have no prior conviction. The superintendent of state police shall annually review the laws of jurisdictions within the United States and Canada with respect to the applicable requirements for licensing or registration of firearms and shall publish a list of those jurisdictions which prohibit possession of a firearm by a person previously convicted of a felony or crimes which if committed in New York state would constitute a felony.

- 14. Possession in accordance with the provisions of this paragraph of a self-defense spray device as defined herein for the protection of a person or property and use of such self-defense spray device under circumstances which would justify the use of physical force pursuant to article thirty-five of this chapter.
- (a) As used in this section "self-defense spray device" shall mean a pocket sized spray device which contains and releases a chemical or organic substance which is intended to produce temporary physical discomfort or disability through being vaporized or otherwise dispensed in the air or any like device containing tear gas, pepper or similar disabling agent.
- (b) The exemption under this paragraph shall not apply to a person who:
 - (i) is less than eighteen years of age; or
 - (ii) has been previously convicted in this state of a

felony or any assault; or

- (iii) has been convicted of a crime outside the state of New York which if committed in New York would constitute a felony or any assault crime.
- (c) The department of health, with the cooperation of the division of criminal justice services and the superintendent of state police, shall develop standards and promulgate regulations regarding the type of self-defense spray device which may lawfully be purchased, possessed and used pursuant to this paragraph. The regulations shall include a requirement that every self-defense spray device which may be lawfully purchased, possessed or used pursuant to this paragraph have a label which states: "WARNING: The use of this substance or device for any purpose other than self-defense is a criminal offense under the law. The contents are dangerous - use with care. This device shall not be sold by anyone other than a licensed or authorized dealer. Possession of this device by any person under the age of eighteen or by anyone who has been convicted of a felony or assault is illegal. Violators may be prosecuted under the law."
- 15. Possession and sale of a self-defense spray device as defined in paragraph fourteen of this subdivision by a dealer in firearms licensed pursuant to section 400.00 of this chapter, a pharmacist licensed pursuant to article one hundred thirty-seven of the education law or by such other vendor as may be authorized and approved by the superintendent of state police.
- (a) Every self-defense spray device shall be accompanied by an insert or inserts which include directions for use, first aid information, safety and storage information and which shall also contain a toll free telephone number for the purpose of allowing any purchaser to call and receive additional information regarding the availability of local courses in self-defense training and safety in the use of a self-defense spray device.
- (b) Before delivering a self-defense spray device to any person, the licensed or authorized dealer shall require proof of age and a sworn statement on a form approved by the superintendent of state police that such person has not been convicted of a felony or any crime involving an assault. Such forms shall be forwarded to the division of state police at such intervals as directed by the superintendent of state police. Absent any such direction the forms shall be maintained on the

premises of the vendor and shall be open at all reasonable hours for inspection by any peace officer or police officer, acting pursuant to his or her special duties. No more than two self-defense spray devices may be sold at any one time to a single purchaser.

- 16. The terms "rifle," "shotgun," "pistol," "revolver," and "firearm" as used in paragraphs three, four, five, seven, seven-a, seven-b, nine, nine-a, ten, twelve, thirteen and thirteen-a of this subdivision shall not include a disguised gun or an assault weapon.
- b. Section 265.01 shall not apply to possession of that type of billy commonly known as a "police baton" which is twenty-four to twenty-six inches in length and no more than one and one-quarter inches in thickness by members of an auxiliary police force of a city with a population in excess of one million persons or the county of Suffolk when duly authorized by regulation or order issued by the police commissioner of such city or such county respectively. Such regulations shall require training in the use of the police baton including but not limited to the defensive use of the baton and instruction in the legal use of deadly physical force pursuant to article thirty-five of this chapter. Notwithstanding the provisions of this section or any other provision of law, possession of such baton shall not be authorized when used intentionally to strike another person except in those situations when the use of deadly physical force is authorized by such article thirty-five.
- c. Sections 265.01, 265.10 and 265.15 shall not apply to possession of billies or blackjacks by persons:
- 1. while employed in fulfilling contracts with New York state, its agencies or political subdivisions for the purchase of billies or blackjacks; or
- 2. while employed in fulfilling contracts with sister states, their agencies or political subdivisions for the purchase of billies or blackjacks; or
- 3. while employed in fulfilling contracts with foreign countries, their agencies or political subdivisions for the purchase of billies or blackjacks as permitted under federal law.
- d. Subdivision one of section 265.01 and subdivision four of section 265.15 of this article shall not apply to possession or ownership of automatic knives by any cutlery and knife museum established pursuant to section two hundred

sixteen-c of the education law or by any director, officer, employee, or agent thereof when he or she is in possession of an automatic knife and acting in furtherance of the business of such museum.

§ 265.25 Certain wounds to be reported

Every case of a bullet wound, gunshot wound, powder burn or any other injury arising from or caused by the discharge of a gun or firearm, and every case of a wound which is likely to or may result in death and is actually or apparently inflicted by a knife, icepick or other sharp or pointed instrument, shall be reported at once to the police authorities of the city, town or village where the person reporting is located by: (a) the physician attending or treating the case; or (b) the manager, superintendent or other person in charge, whenever such case is treated in a hospital, sanitarium or other institution. Failure to make such report is a class A misdemeanor. This subdivision shall not apply to such wounds, burns or injuries received by a member of the armed forces of the United States or the state of New York while engaged in the actual performance of duty.

§ 265.26 Burn injury and wounds to be reported

Every case of a burn injury or wound, where the victim sustained second or third degree burns to five percent or more of the body and/or any burns to the upper respiratory tract or laryngeal edema due to the inhalation of super-heated air, and every case of a burn injury or wound which is likely to or may result in death, shall be reported at once to the office of fire prevention and control. The state fire administrator shall accept the report and notify the proper investigatory agency. A written report shall also be provided to the office of fire prevention and control within seventy-two hours. The report shall be made by (a) the physician attending or treating the case; or (b) the manager, superintendent or other person in charge, whenever such case is treated in a hospital, sanitarium, institution or other medical facility.

The intentional failure to make such report is a class A misdemeanor.

§ 265.30 Certain convictions to be reported

Every conviction under this article or section 400.00, of a person who is not a citizen of the United States, shall be certified to the proper officer of the United States government by the district attorney of the county in which such conviction was had.

§ 265.35 Prohibited use of weapons

- 1. Any person hunting with a dangerous weapon in any county wholly embraced within the territorial limits of a city is guilty of a class A misdemeanor.
- 2. Any person who wilfully discharges a loaded firearm or any other gun, the propelling force of which is gunpowder, at an aircraft while such aircraft is in motion in the air or in motion or stationary upon the ground, or at any railway or street railroad train as defined by the public service law, or at a locomotive, car, bus or vehicle standing or moving upon such railway, railroad or public highway, is guilty of a class D felony if thereby the safety of any person is endangered, and in every other case, of a class E felony.
- 3. Any person who, otherwise than in self defense or in the discharge of official duty, (a) wilfully discharges any species of firearms, air-gun or other weapon, or throws any other deadly missile, either in a public place, or in any place where there is any person to be endangered thereby, or, in Putnam county, within one-quarter mile of any occupied school building other than under supervised instruction by properly authorized instructors although no injury to any person ensues; (b) intentionally, without malice, points or aims any firearm or any other gun, the propelling force of which is gunpowder, at or toward any other person; (c) discharges, without injury to any other person, firearms or any other guns, the propelling force of which is gunpowder, while intentionally without malice, aimed at or toward any person; or (d) maims or injures any other person by the discharge of any firearm or any other gun, the propelling force of which is gunpowder, pointed or aimed intentionally, but without malice, at any such person, is guilty of a class A misdemeanor.

§ 265.40 Purchase of rifles and/or shotguns in contiguous states

Definitions. As used in this act:

- 1. "Contiguous state" shall mean any state having any portion of its border in common with a portion of the border of the state of New York;
- 2. All other terms herein shall be given the meaning prescribed in Public Law 90-618 known as the "Gun Control Act of 1968" (18 U.S.C. 921). IFN11

It shall be lawful for a person or persons residing in this state, to purchase or otherwise obtain a rifle and/or shotgun in a contiguous state, and to receive or transport such rifle and/or shotgun into this state; provided, however, such person is otherwise eligible to possess a rifle and/or shotgun under the laws of this state.

[FN1] 18 USCA § 921 et seq.

Article 270. Other Offenses Relating to Public Safety

- § 270.00 Unlawfully dealing with fireworks and dangerous fireworks
- § 270.10 Creating a hazard
- § 270.15 Unlawfully refusing to yield a party line
- § 270.20 Unlawful wearing of a body vest
- § 270.25 Unlawful fleeing a police officer in a motor vehicle in the third degree
- § 270.30 Unlawful fleeing a police officer in a motor vehicle in the second degree
- § 270.35 Unlawful fleeing a police officer in a motor vehicle in the first degree

Article 275. Offenses Relating to Unauthorized Recording

- § 275.00 Definitions
- § 275.05 Manufacture of unauthorized recordings in the second degree
- § 275.10 Manufacture of unauthorized recordings in the first degree
- § 275.15 Manufacture or sale of an unauthorized recording of a performance in the second degree.
- § 275.20 Manufacture or sale of an unauthorized recording of a performance in the first degree
- § 275.25 Advertisement or sale of unauthorized recordings in the second degree
- § 275.30 Advertisement or sale of unauthorized recordings in the first degree
- § 275.32 Unlawful operation of a recording device in a motion picture or live theater in the third degree
- § 275.33 Unlawful operation of a recording device in a motion picture or live theater in the second degree

§ 275.34 Unlawful operation of a recording device in a motion picture or live theater in the first degree

§ 275.35 Failure to disclose the origin of a recording in the second degree

§ 275.40 Failure to disclose the origin of a recording in the first degree

§ 275.45 Limitations of application

TITLE W. PROVISIONS RELATING TO FIREARMS, FIREWORKS, PORNOGRAPHY EQUIPMENT AND VEHICLES USED IN THE TRANSPORTATION OF GAMBLING RECORDS

Article 400. Licensing and Other Provisions Relating to Firearms

§ 400.00 Licenses to carry, possess, repair and dispose of firearms

§ 400.01 License to carry and possess firearms for retired sworn members of the division of state police

§ 400.05 Disposition of weapons and dangerous instruments, appliances and substances

§ 400.10 Report of theft or loss of a firearm, rifle or shotgun

Article 405. Licensing and Other Provisions Relating to Fireworks

§ 405.00 Permits for public displays of fireworks

§ 405.05 Seizure and destruction of fireworks

§ 405.10 Permits for indoor pyrotechnics

§ 405.12 Unpermitted use of indoor pyrotechnics in the second degree

§ 405.14 Unpermitted use of indoor pyrotechnics in the first degree

§ 405.16 Aggravated unpermitted use of indoor pyrotechnics in the second degree

§ 405.18 Aggravated unpermitted use of indoor pyrotechnics in the first degree

Article 410. Seizure and Forfeiture of Equipment Used in Promoting Pornography

§ 410.00 Seizure and forfeiture of equipment used in photographing, filming, producing, manufacturing, projecting or distributing pornographic still or motion pictures

Article 415. Seizure and Forfeiture of Vehicles, Vessels and Aircraft Used to Transport or Conceal Gambling Records

§ 415.00 Seizure and forfeiture of vehicles, vessels and aircraft used to transport or conceal gambling records

§ 420.05 Seizure and forfeiture of equipment used in the production of unauthorized recordings

Article 450. Disposal of Stolen Property

§ 450.10 Disposal of stolen property

TITLE X. ORGANIZED CRIME CONTROL ACT

Article 460. Enterprise Corruption

§ 460.00 Legislative findings

§ 460.10 Definitions

§ 460.20 Enterprise corruption

§ 460.25 Enterprise corruption; limitations

§ 460.30 Enterprise corruption; forfeiture

§ 460.40 Enterprise corruption; jurisdiction

§ 460.50 Enterprise corruption; prosecution

§ 460.60 Enterprise corruption; consent to prosecute

§ 460.70 Provisional remedies

§ 460.80 Court ordered disclosure

Article 470. Money Laundering

§ 470.00 Definitions

- § 470.05 Money laundering in the fourth degree
- § 470.10 Money laundering in the third degree
- § 470.15 Money laundering in the second degree
- § 470.20 Money laundering in the first degree
- § 470.21 Money laundering in support of terrorism in the fourth degree
- § 470.22 Money laundering in support of terrorism in the third degree
- § 470.23 Money laundering in support of terrorism in the second degree
- § 470.24 Money laundering in support of terrorism in the first degree
- § 470.25 Money laundering; fines

Article 480. Criminal Forfeiture — Felony Controlled Substance Offenses

- § 480.00 Definitions
- § 480.10 Procedure
- § 480.20 Disposal of property
- § 480.25 Election of remedies
- § 480.30 Provisional remedies

TITLE Y. HATE CRIMES ACT OF 2000

Article 485. Hate Crimes

- § 485.00 Legislative findings
- § 485.05 Hate crimes
- § 485.10 Sentencing

TITLE Y-1. TERRORISM

Article 490. Terrorism

- § 490.00 Legislative findings
- § 490.01 Liability protection
- § 490.05 Definitions
- § 490.10 Soliciting or providing support for an act of terrorism in the second degree

- § 490.15 Soliciting or providing support for an act of terrorism in the first degree
- § 490.20 Making a terroristic threat
- § 490.25 Crime of terrorism
- § 490.30 Hindering prosecution of terrorism in the second degree
- § 490.35 Hindering prosecution of terrorism in the first degree
- § 490.37 Criminal possession of a chemical weapon or biological weapon in the third degree
- § 490.40 Criminal possession of a chemical weapon or biological weapon in the second degree
- § 490.45 Criminal possession of a chemical weapon or biological weapon in the first degree
- § 490.47 Criminal use of a chemical weapon or biological weapon in the third degree
- § 490.50 Criminal use of a chemical weapon or biological weapon in the second degree
- § 490.55 Criminal use of a chemical weapon or biological weapon in the first degree
- § 490.70 Limitations

Article 500. Laws Repealed; Time of Taking Effect

§ 500.05 Laws repealed

Chapter eighty-eight of the laws of nineteen hundred nine, entitled "An act providing for the punishment of crime, constituting chapter forty of the consolidated laws," and all acts amendatory thereof and supplemental thereto, constituting the penal law as heretofore in force, are hereby repealed.

§ 500.10 Time of taking effect

This act shall take effect September first, nineteen hundred sixty-seven.